12-11076-shl Doc 690 Filed 12/04/12 Entered 12/04/12 18:01:19 Main Document

Pg 1 of 29
HEARING DATE AND TIME: December 7, 2012 at 3:00 p.m. (Eastern Time)
OBJECTION DEADLINE: December 6, 2012 at 3:00 p.m. (Eastern Time)

GIBSON, DUNN & CRUTCHER LLP

Michael A. Rosenthal (MR-7006) Craig H. Millet (admitted *pro hac vice*) Matthew J. Williams (MW-4081) Joshua Weisser (JW-0185) 200 Park Avenue New York, New York 10166-0193 Telephone: (212) 351-4000

Telephone: (212) 351-4000 Facsimile: (212) 351-4035

Attorneys for the Debtors and Debtors in Possession

UNITED STATES BANKRUPTCY COURT

SOUTHERN DISTRICT OF NEW YORK		
	X	
IN RE:	Chapter 11	
ARCAPITA BANK B.S.C.(c), et al.,	: Case No. 12-11076 (SHL)	
Debtors.	: Jointly Administered	
	: v	

NOTICE OF HEARING ON DEBTORS' MOTION FOR THE ENTRY OF INTERIM AND FINAL ORDERS PURSUANT TO 11 U.S.C. §§ 105, 362, 363(b)(1), 363(m), 364(c)(1), 364(c)(2), 364(c)(3), AND 364(e) AND BANKRUPTCY RULES 4001 AND 6004 (I) AUTHORIZING DEBTORS (A) TO ENTER INTO AND PERFORM UNDER DIP AGREEMENT, AND (B) TO OBTAIN CREDIT ON A SECURED SUPERPRIORITY BASIS, (II) SCHEDULING FINAL HEARING PURSUANT TO BANKRUPTCY RULES 4001(b) AND (c) AND (III) GRANTING RELATED RELIEF

PLEASE TAKE NOTICE that on December 4, 2012, Arcapita Bank B.S.C.(c) and certain of its subsidiaries and affiliates, as debtors and debtors in possession (collectively, the "*Debtors*") filed the annexed Motion (the "*Motion*").

PLEASE TAKE FURTHER NOTICE that an interim hearing to consider the annexed Motion will take place before the Honorable Sean H. Lane, United States Bankruptcy Judge, in Room 701 of the United States Bankruptcy Court for the Southern District of New

York (the "*Bankruptcy Court*"), One Bowling Green, New York, New York 10004, on December 7, 2012 at 3:00 p.m. (Eastern Time) (the "*Hearing*"), or as soon thereafter as counsel may be heard.

PLEASE TAKE FURTHER NOTICE that any responses or objections to the Motion (the "*Objections*") shall be filed electronically with the Court on the docket of *In re* Arcapita Bank B.S.C.(c), et al., Ch. 11 Case No. 12-11076 (SHL) (the "**Docket**"), pursuant to the Case Management Procedures approved by this Court¹ and the Court's General Order M-399 (available at http://nysb.uscourts.gov/orders/orders2.html), by registered users of the Court's case filing system and by all other parties in interest on a 3.5 inch disk, preferably in portable document format, Microsoft Word, or any other Windows-based word processing format (with a hard copy delivered directly to Chambers), in accordance with the customary practices of the Bankruptcy Court and General Order M-399, to the extent applicable, and served in accordance with General Order M-399 on (i) counsel for the Debtors, Gibson, Dunn & Crutcher LLP, 200 Park Avenue, New York, New York 10166 (Attn: Michael A. Rosenthal, Esq., Matthew J. Williams, Esq. and Joshua Weisser, Esq.); (ii) the Office of the United States Trustee for the Southern District of New York, 33 Whitehall Street, 21st Floor, New York, New York 10004 (Attn: Richard Morrissey, Esq.); (iii) the Official Committee of Unsecured Creditors, Milbank, Tweed, Hadley & McCloy LLP, 1 Chase Manhattan Plaza, New York, New York 10005 (Attn: Dennis F. Dunne, Esq. and Evan R. Fleck, Esq.; (iv) counsel for Standard Chartered Bank, Dechert LLP, 1095 Avenue of the Americas, New York, New York 10036 (Attn: Brian E. Greer,

See Order (A) Waiving the Requirement That Each Debtor File a List of Creditors and Equity Security Holders and Authorizing Maintenance of Consolidated List of Creditors in Lieu of a Matrix; (B) Authorizing Filing of a Consolidated List of Top 50 Unsecured Creditors; and (C) Approving Case Management Procedures [Docket No. 21].

Esq. and Nicole Herther-Spiro, Esq.); and (v) counsel for Fortress Credit Corp., Skadden, Arps, Slate, Meagher & Flom LLP, 4 Times Square, New York, New York 10036 (Attn: Kenneth S.

Ziman) so as to be received no later than December 6, 2012 at 3:00 p.m. (Eastern Time) (the

"Objection Deadline").

PLEASE TAKE FURTHER NOTICE that if no Objections are timely filed and served with respect to the Motion, the Debtors may, on or after the Objection Deadline, submit to the Bankruptcy Court an order substantially in the form of the proposed order annexed to the Motion, which order may be entered with no further notice or opportunity to be heard.

Dated:

New York, New York

December 4, 2012

/s/ Michael A. Rosenthal

Michael A. Rosenthal (MR-7006) Craig H. Millet (admitted *pro hac vice*)

Matthew J. Williams (MW-4081)

Joshua Weisser (JW-0185)

GIBSON, DUNN & CRUTCHER LLP

200 Park Avenue

New York, New York 10166-0193

Telephone: (212) 351-4000

Facsimile: (212) 351-4035

ATTORNEYS FOR THE DEBTORS AND DEBTORS IN POSSESSION

GIBSON, DUNN & CRUTCHER LLP

Michael A. Rosenthal (MR-7006) Craig H. Millet (admitted *pro hac vice*) Matthew J. Williams (MW-4081) Joshua Weisser (JW-0185) 200 Park Avenue New York, New York 10166-0193 Telephone: (212) 351-4000

Telephone: (212) 351-4000 Facsimile: (212) 351-4035

Attorneys for the Debtors and Debtors in Possession

UNITED STATES BANKRUPTCY COURT SOUTHERN DISTRICT OF NEW YORK

In re Chapter 11 Case

ARCAPITA BANK B.S.C.(c), et al.,

Chapter 11 Case

Case No. 12-11076 (SHL)

Debtors. : Jointly Administered

DEBTORS' MOTION FOR THE ENTRY OF INTERIM AND FINAL ORDERS PURSUANT TO 11 U.S.C. §§ 105, 362, 363(b)(1), 363(m), 364(c)(1), 364(c)(2), 364(c)(3), AND 364(e) AND BANKRUPTCY RULES 4001 AND 6004 (I) AUTHORIZING DEBTORS (A) TO ENTER INTO AND PERFORM UNDER DIP AGREEMENT, AND (B) TO OBTAIN CREDIT ON A SECURED SUPERPRIORITY BASIS, (II) SCHEDULING FINAL HEARING PURSUANT TO BANKRUPTCY RULES 4001(b) AND (c) AND (III) GRANTING RELATED RELIEF

Arcapita Bank B.S.C.(c) ("Arcapita"), Arcapita Investment Holdings Limited ("AIHL"), Arcapita LT Holdings Limited ("ALTHL"), WindTurbine Holdings Limited ("WindTurbine"), AEID II Holdings Limited ("AEID II") and RailInvest Holdings Limited ("RailInvest"), as debtors and debtors in possession (collectively, the "Debtors" and each, a "Debtor") in the above-captioned chapter 11 cases (collectively, the "Chapter 11 Cases") hereby submit this Motion (the "DIP Motion") and respectfully represent as follows:

PRELIMINARY STATEMENT¹

Arcapita, a Bahraini entity, is a leading arranger and manager of Shari'ah compliant alternative investments, investments that comply with Islamic banking rules which forbid the charging of interest. One form of Shari'ah compliant financing, a Murabaha, typically consists of a sale by the "lender" of a specific amount of commodities for a set price (which consists of the actual out of pocket costs of the "lender" plus an agreed upon profit) to the "borrower." The "borrower" agrees to pay for the commodities on deferred payment terms. The "borrower" then sells the commodities, for cash, to a third party. The end result is that the "borrower" receives an immediate cash infusion and incurs a future obligation to pay the "lender" the agreed upon price. Murabaha-based financings may be secured. Indeed, prior to the Petition Date, the Debtors entered into two separate secured Murabahas with Standard Chartered Bank that provided them with approximately \$100 million of financing.

By this Motion, the Debtors seek to enter into a non-priming secured Murabaha financing transaction with a principal amount of up to \$150 million. Notwithstanding the use of different terms and a modified financing structure, the proposed DIP Transaction is in essence a postpetition financing package that will enable the Debtors to successfully conclude these cases.

Numerous reasons justify approval of the proposed DIP Transaction, including:

• The DIP Transaction Is Critical to the Debtors' Reorganization Efforts:
At December 1, the Debtors only had immediate access to approximately
\$19.8 million in cash and cash equivalents. Absent additional liquidity,
there is a risk that the Debtors will be unable to pay future restructuring
costs or support Arcapita's interests in Arcapita Group portfolio
companies and investments, thereby impairing the Debtors' reorganization
efforts.

For ease of reference, this DIP Motion uses terms typical of a DIP loan, such as "lender" and "borrower". In actuality, as discussed above, the post-petition financing transaction will be in the form of a Shari'ah compliant Murabaha, or commodities transaction, not a loan.

- The DIP Transaction Represents the Best Offer Available: The Debtors and the Official Committee of Unsecured Creditors engaged in a lengthy, thorough and competitive process to procure the best available financing. The DIP Transaction is the product of this solicitation process. Alternative proposals either were on less favorable economic, business or legal terms, did not provide adequate liquidity, or were subject to due diligence or internal credit committee approval.
- <u>The DIP Transaction Accords with the SCB Settlement</u>: The SCB Facilities are secured, in part, by pledges of the equity of AEID II, RailInvest and WindTurbine. The DIP Transaction is structured to not impair the value of these pledges. As a result, the Debtors submit, this non-priming transaction complies with the SCB Order (as defined below).

For the foregoing reasons, the DIP Transaction should be approved.

JURISDICTION AND VENUE

- 1. The Court has jurisdiction to consider this DIP Motion pursuant to 28 U.S.C. §§ 157 and 1334. This is a core proceeding pursuant to 28 U.S.C. § 157(b)(2). Venue is proper pursuant to 28 U.S.C. §§ 1408 and 1409.
- 2. The statutory bases for the relief requested herein are sections 105, 362, 363(b)(1), 363(m), 364(c)(1), 364(c)(2), 364(c)(3), and 364(e) of title 11 of the United States Code (the "*Bankruptcy Code*"); Rules 2002, 4001, 6004 and 9014 of the Federal Rules of Bankruptcy Procedure (the "*Bankruptcy Rules*"); and Rule 4001-2 of the Local Bankruptcy Rules for the Southern District of New York (the "*Local Rules*").

BACKGROUND

3. On March 19, 2012 (the "*Petition Date*"), Arcapita and five of its affiliates commenced cases under chapter 11 of the Bankruptcy Code. On April 30, 2012, Falcon Gas Storage Co., Inc. ("*Falcon Gas*") commenced a case under chapter 11 of the Bankruptcy Code. The Debtors are operating their businesses and managing their properties as debtors in possession pursuant to sections 1107(a) and 1108 of the Bankruptcy Code.

- 4. On April 5, 2012, the United States Trustee for Region 2 (the "*United States Trustee*") appointed the Official Committee of Unsecured Creditors in the Chapter 11 Cases (the "*Committee*") [Dkt. No. 60] pursuant to sections 1102(a) and (b) of the Bankruptcy Code.
- 5. On September 25, 2012, the Debtors filed their *Motion for Entry of an Order*Authorizing the Debtors To Enter into a Financing Commitment Letter and Incur Related Fees,

 Expenses and Indemnities [Dkt. No. 513] (the "Original Commitment Letter Motion").
- 6. A hearing on the Original Commitment Letter Motion took place on October 9, 2012 (the "*Initial Hearing*"). Prior to ruling, the Court adjourned the Initial Hearing to afford the parties an opportunity to attempt to reach a consensual resolution of disputed issues and address the Court's stated concerns with respect to the Original Commitment Letter.
- 7. On November 1, 2012, the Debtors filed their Supplement to the Debtors' Motion for Entry of an Order Authorizing the Debtors To Enter into a Financing Commitment Letter and Incur Related Fees, Expenses and Indemnities in Connection Therewith [Docket No. 610] (the "Commitment Letter Motion Supplement" and together with the Original Commitment Letter Motion, the "Commitment Letter Motion"), by which they sought this Court's approval of the Debtors' entry into a commitment letter (as later revised, the "Fortress Commitment Letter") with Fortress Credit Corp. ("Fortress").
- 8. The Court approved the Commitment Letter Motion on November 7, 2012, authorizing the Debtors to enter into the Fortress Commitment Letter.

RELIEF REQUESTED

- 9. By this DIP Motion, and pursuant to sections 105, 362, 363(b)(1), 363(m), 364(c)(1), 364(c)(2), 364(c)(3), and 364(e) of the Bankruptcy Code, and Bankruptcy Rules 2002, 4001, 6004 and 9014 and Local Rule 4001-2, the Debtors seek entry of an interim order (the "*Interim Order*"), substantially in the form attached hereto as *Exhibit A* and a final order (the "*Final Order*" and, together with the Interim Order, the "*DIP Orders*") granting the following relief:
 - (a) authorizing the Debtors to:
 - (i) obtain secured, superpriority, post-petition Shari'ah compliant financing with a transaction value of up to \$150 million (the "DIP Transaction") pursuant to the terms and conditions of a Master Murabaha Agreement (as the same may be amended, supplemented, restated or otherwise modified from time to time, the "DIP Agreement") by and among AIHL and Fortress, in its capacity as Investment Agent (in such capacity, the "Investment Agent") for the participants from time to time party to the DIP Agreement (each, a "DIP Participant") and collectively, including the Investment Agent, the "DIP Participants"), substantially in the form of Exhibit B attached to this DIP Motion;
 - (ii) execute and deliver the DIP Agreement and other documents entered into, executed or delivered in connection with the DIP Transaction (collectively, and together with the DIP Agreement, the "DIP Transaction Documents");
 - (iii) use the proceeds from the DIP Transaction for general corporate purposes in accordance with the DIP Budget (as defined in the DIP Agreement and annexed hereto as *Exhibit C*) and subject to the financial covenants, terms, conditions and limitations set forth in the DIP Transaction Documents and the DIP Orders; and
 - grant to the Security Agent (as defined below), for the benefit of itself and the DIP Participants, security interests in and valid, enforceable, non-avoidable and automatically fully perfected liens on and in the now-existing or after-acquired DIP Collateral (as defined below) (with the priority set forth in the DIP Orders) to secure the obligations in favor of the DIP Participants arising under the DIP Transaction Documents (collectively, the "DIP Obligations");

- (b) authorizing the Debtors, on an interim basis, pursuant to the DIP Agreement, to incur DIP Obligations with an aggregate principal amount of \$25 million (the "*Interim Amount*");
- (c) authorizing the Debtors, subject to entry of the Final Order, pursuant to the DIP Agreement, to incur DIP Obligations with an aggregate principal amount of \$150 million (the "*Total Amount*");
- (d) granting to the DIP Participants superpriority administrative expense claims in the Chapter 11 Cases of the Debtor Obligors (as defined below) with respect to the DIP Obligations and to the extent set forth in the DIP Transaction Documents, subject to (i) the Carve Out (as defined below), (ii) the SCB Guarantee Claims (as defined below), (iii) the SCB Superpriority Claims (as defined in the SCB Order), and (iv) SCB's adequate protection as provided in the SCB Order, where applicable;
- (e) modifying the automatic stay imposed by section 362 of the Bankruptcy Code to the extent necessary to implement and effectuate the terms of the DIP Transaction Documents and the DIP Orders;
- (f) scheduling a final hearing on the DIP Motion (the "*Final DIP Hearing*") to consider entry of the Final Order and establishing notice procedures in respect of the Final DIP Hearing; and
- (g) granting related relief.

BANKRUPTCY RULE 4001 AND LOCAL RULE 4001-2 CONCISE STATEMENT

10. Pursuant to and in accordance with Bankruptcy Rule 4001(c)(1)(B)(i)-(xi) and Local Rule 4001-2(a)-(i), the material provisions of the DIP Agreement and/or the Interim Order, and the location of such provisions therein, are as follows:²

This summary is qualified in its entirety by the provisions of the DIP Agreement and/or the Interim Order, as applicable. Capitalized terms used but not otherwise defined herein shall have the meanings set forth in the DIP Agreement and/or the Interim Order, as applicable. To the extent there are any conflicts between this summary and the DIP Agreement and/or the Interim Order, as applicable, the terms of the DIP Agreement and/or the Interim Order, as applicable, shall govern.

MATERIAL TERMS OF THE DIP TRANSACTION DOCUMENTS	
DIP Agreement Parties Fed R. Bankr. P. 4001(c)(1)(B) DIP Agreement, § 1.1, Definitions, decretal paragraph	Purchaser/Borrower: AIHL Guarantors: (i) Arcapita; (ii) ALTHL; (iii) WindTurbine; (iv) AEID II; (v) RailInvest; (vi) Arcapita Investment Management Limited ("AIML"); (vii) Arcapita Inc.; (viii) Arcapita Structured Finance Ltd; (ix) Arcapita Investment Funding Limited ("AIFL"); (x) Arcapita Industrial Management I Ltd; (xi) Arcapita (US) Limited; (xii) Arcapita (Europe) Limited; (xiii) Arcapita (Singapore) Limited; (xiv) the LT CayCos³; (xv) all wholly owned WCF Entities⁴ (other than as agreed to by AIHL and the Investment Agent); except for specificed entities, as agreed by the parties, and entities formed for the purpose of implementing a chapter 11 plan (collectively, with AIHL, the "Obligors").⁵ Investment Agent: Fortress, pursuant to an investment agency agreement substantially in the form of that certain Investment Agency Agreement annexed hereto as Exhibit F. Security Agent: Fortress. Arranger: Fortress. DIP Participants: banks, financial institutions and other parties listed in Schedule 1 to the Investment Agency Agreement or any assignee or transferee which has become a DIP Participant in accordance therewith.
DIP Commitments Local Rule 4001-2(a)(1); Fed. R. Bankr. P. 4001(c)(1)(B) DIP Agreement, §§ 2.1, 9.4, 9.1.2	\$150 million (the "DIP Facility") comprised of an initial \$125 million multi-draw term facility (the "Initial Amount") which (subject to confirmatory due diligence) may be increased by \$25 million (the "Conditional Amount"); provided that, 120 days after closing, the \$125 million Initial Amount shall be reduced to \$100 million.
Economics Fed. R. Bankr. P. 4001(c)(1)(B) DIP Agreement, Art. 6, § 1.1, Definitions of Deferred Sale Price	 "Deferred Sale Price" (i.e. the amount payable by AIHL to the Investment Agent for the purchase of Commodities (as defined in the DIP Agreement)) is the aggregate of: (1) the Cost Price; plus (2) the Profit Amount; plus (3) the Mandatory Cost (as defined in the DIP Agreement) (if any); plus (4) the Increased Costs (as defined in the DIP Agreement) (if any); plus (5) VAT (as defined in the DIP Agreement) (if any). "Cost Price" means the amount (in Dollars) payable or paid by the Investment Agent for the purchase of Commodities by the Investment Agent. "Profit Amount" means: Cost Price * Rate * (N/360), where: "N" is the number of days to elapse from, and including, the proposed Transaction Date. "Rate" means the sum of (a) the greater of (i) LIBOR and (ii) 2% per annum plus (b) 10% per annum.

The "*LT CayCos*" are Cayman Island entities and subsidiaries of ALTHL, through which AIHL maintains its indirect equity interests in the Arcapita Group portfolio companies. For purposes hereof, and as defined herein, the LT CayCos do not include AEID II, RailInvest and WindTurbine.

⁴ The "*WCF Entities*" are Cayman Island entities and direct subsidiaries of AIHL which provide project level financing to the Debtors' portfolio companies.

⁵ AIHL, Arcapita, ALTHL, WindTurbine, AEID II and RailInvest are defined collectively as the "*Debtor Obligors*".

12-11076-shl Doc 690 Filed 12/04/12 Entered 12/04/12 18:01:19 Main Document Pg 11 of 29

Late Payment Fed. R. Bankr. P. 4001(c)(1)(B)	The late payment charges will accrue on a daily basis on the basis of a year of 360 days and shall be calculated in accordance with the following formula: (unpaid amount x (Rate + 6.0%)) /360
DIP Agreement, § 6.5	(unpara amount x (rate : 0.070))/500
Funding Conditions Local Rule 4001-2(a)(2); Local Rule 4001-2(h); Fed. R. Bankr. P. 4001(c)(1)(B) DIP Agreement, § 3.1, 3.2, Schedule 1	 The DIP Transaction is subject to the following conditions precedent: Delivery of a certified copy of the Obligors' constitutional documents, a certified copy of the board minutes of each Obligor, and directors certificates for each Guarantor; Delivery of a certified copy of shareholder resolutions for each Obligor; The delivery of legal opinions; Payment of fees, costs and expenses; Execution of the DIP Transaction Documents; Delivery of evidence of approval of the executive committee of each Obligor's Shari'ah advisory board; Delivery of the DIP Budget;⁶ Entry by the Court of the Interim Order by date certain; and Entry by the Court of the Final Order by date certain.
Fees and Expenses Local Rule 4001-2(a)(3); Fed. R. Bankr. P. 4001(c)(1)(B) DIP Agreement, §15	AIHL shall pay all costs and expenses (including legal fees) reasonably incurred by the Investment Agent and Arranger in connection with the negotiation, preparation, printing, execution and syndication of the DIP Agreement, and the fees in the amounts agreed in each Fee Letter (as defined in the DIP Agreement). AIHL shall also pay all costs and expenses reasonably incurred by the Participants and other Finance Parties (as defined in the DIP Agreement) to enforce their rights under the DIP Agreement.
Priority and Liens Local Rule 4001-2(a)(4); Fed. R. Bankr. P. 4001(c)(1)(B)(i) DIP Agreement, § 7.1, Interim Order, ¶ 12	The DIP Obligations of each Debtor under the DIP Agreement: i. pursuant to section 364(c)(1) of the Bankruptcy Code, shall constitute allowed Superpriority Claims, provided that so long as the SCB Facilities obligations are outstanding, the guarantees of and superpriority claims against AEID II, RailInvest and WindTurbine under the DIP Transaction shall be subordinated to the SCB Guarantees, and SCB shall have a prior superpriority administrative claim in all proceeds of the EuroLog IPO to the extent provided under the SCB Order; ii. pursuant to section 364(c)(2) of the Bankruptcy Code, shall be secured by a perfected first-priority lien on all assets of Arcapita, AIHL and ALTHL, in each case, that are not otherwise subject to SCB's liens; and iii. pursuant to section 364(c)(3) of the Bankruptcy Code, a perfected junior lien on all now owned or after acquired assets of the Debtors that are subject to (x) any valid, perfected and non-avoidable lien in existence on the Petition Date or (y) any valid lien in existence on the Petition Date that is perfected (but not granted) subsequent to the Petition Date pursuant to section 546(b) of the Bankruptcy Code (including, in each case and for so long as the obligations under the SCB Facilities remain unpaid, SCB's liens on the equity of AEID II, RailInvest and WindTurbine), subject and subordinate in each case with respect to subclauses (i) through (iii) above, to the Carve Out. Notwithstanding the foregoing, the liens described above shall not attach to (a) avoidance actions or the proceeds thereof, (b) AIHL's equity interests in PointPark Properties s.r.o., or (c) assets of AEID II, RailInvest and WindTurbine.

⁶ The Debtors submit that the DIP Budget accurately reflects the payment of all administrative expenses due or accruing during the term of the DIP Facility.

Adequate Protection Local Rule 4001-2(a)(4); Fed. R. Bankr. P. 4001(c)(1)(B)(ii) Interim Order, ¶ 11	The SCB Order will not be modified, altered, amended or superseded, and it remains in full force and effect, including with respect to any grant of adequate protection to SCB thereunder.
Carve Out Local Rule 4001-2(a)(5); Fed. R. Bankr. P. 4001(c)(1)(B) Interim Order, ¶ 10(b)	 i. any fees of the Clerk of the Bankruptcy Court and to the U.S. Trustee under 28 U.S.C. § 1930(a); ii. reasonable fees and expenses approved by the Court incurred by a trustee; iii. reasonable and documented expenses of Committee members in an amount not to exceed \$200,000; iv. all unpaid fees and expenses allowed by the Court of professionals or professional firms retained by the Debtors or the Committee and the reasonable fees and expenses of the joint provisional liquidators appointed in the Cayman Islands liquidation proceedings of AIHL (the "Joint Provisional Liquidators" and together with Debtor and Committee professionals, the "Professional Persons") that were accrued or incurred, as applicable through the date upon which AIHL and the Committee receive from the Investment Agent a written notice of the occurrence of an Event of Default (as defined in the DIP Agreement) and the Investment Agent's intention to involve the Carve Out (the "Carve Out Notice"); and v. all fees and expenses of Professional Persons incurred after the date upon which AIHL receives
	v. all fees and expenses of Professional Persons incurred after the date upon which AIHL receives the Carve Out Notice, in the aggregate amount not to exceed \$15,000,000.
Covenants Local Rule 4001-2(a)(8); Fed. R. Bankr. P. 4001(c)(1)(B) DIP Agreement, Art. 12	 AIHL shall: Supply the following to the Investment Agent: consolidated financial statements; information on each Investment Company (as defined in the DIP Agreement) in certain circumstances; details of any litigation, arbitration or administrative proceeding reasonably expected to have a Material Adverse Effect (as defined in the DIP Agreement); information regarding the financial condition and business operations of any member of the Group (as defined in the DIP Agreement) or any Investment Company (as defined in the DIP Agreement); prompt written notice of an initial public offering; Provide reasonable access to their books and records; Notify the Investment Agent of any default; Procure that the obligations of each non-Debtor Obligor will rank at least pari passu with all its
	other present and future unsecured and unsubordinated Financial Indebtedness (as defined in the DIP Agreement); Not incur or suffer or permit to exist any Financial Indebtedness, other than as existing on the closing date and identified on the Debtors schedules of assets and liabilities;
	 Not permit any liens to be created or subsist other than specified permitted liens; Not enter into a transaction to sell, lease, transfer or otherwise dispose of any asset; Not make any loan or capital contribution to, or purchase equity interests, debt securities of, or any assets constituting a business unit of, or make any other investment in, any person (subject to specified carve outs);
	 Effect and maintain insurance; Pay and discharge all taxes and governmental charges; Comply in all material respects with all laws, rules and regulations; Ensure that (a) no substantial change is made to the general nature of the business of AIHL or the Group; (b) each Guarantor (other than Arcapita) shall remain a wholly-owned subsidiary of Arcapita; (c) ALTHL will remain a wholly-owned subsidiary of AIHL; and (d) each LT CayCo will remain a wholly owned subsidiary of ALTHL;

Procure that no proxies are granted to it or any other Guarantor or any of their respective subsidiaries by any LT CayCo; Ensure that no member of the Group shall open a deposit account; Not incur, create, assume, suffer to exist or permit any other superpriority claim or lien on any Collateral in the Chapter 11 Cases which is part passu with or senior to the claims of the Investment Agent and the DIP Participants granted pursuant to the DIP Agreement, any DIP Transaction Document andfor the Interior Morder (or the Final Order, as applicable); No Obligor (A) shall, or shall permit any of its subsidiaries to, cancel any claim or debt owing to it, except for reasonable consideration negotiated on an arm's length basis and in the ordinary course of business; (B) shall, or shall permit any of its subsidiaries to enter into or become bound by any effective agreement, instrument, indenture or other obligation that could directly or indirectly restrict, prohibit or require the consent of any person with respect to the payment of dividends or distributions or the making or repayment of intercompany loans by a subsidiary or indirectly restrict, prohibit or require the consent of any person with respect to the payment of dividends or distributions or the making or repayment of intercompany loans by a subsidiary of bridge contracts or similar transactions; and No Debtor shall, without the express prior written consent of the Investment Agent or any payment or transfer with respect to any lien or Financial Indebtedness incurred or arising prior to the Petition Date (except pursuant to a chapter II plan). Purpose and Limitations Local Rule 4001-2(a)(9): Purpose and Limitations Local Rule 4001-2(a)(9): Purpose and Limitations Local Rule 4001-2(a)(10): Purpos		
Ensure that no member of the Group shall open a deposit account; Not incur, create, assume, suffer to exist or permit any other superpriority claim or lien on any Collateral in the Chapter 11 Classe which is pari passu with or senior to the claims of the Investment Agent and the DIP Participants granted pursuant to the DIP Agreement, any DIP Transaction Document and/or the Interim Order (or the Tianal Order, as applicable); No Obligor (A) shall, or shall permit any of its subsidiaries to, cancel any claim or debt owing to it, except for reasonable consideration negotiated on an arm's length basis and in the ordinary course of business; (B) shall, or shall permit any of its subsidiaries to, cancel any claim or debt owing to it, except for reasonable consideration negotiated on an arm's length basis and in the ordinary course of business; (B) shall, or shall permit any of its subsidiaries to, cancel any claim or debt owing to it, except for reasonable consideration and any of the consent of any person with respect to the payment of dividends or distributions or the making or repayment of intercompany loans by a subsidiary of any Obligor to any Obligor or between Obligors; (C) shall engage in any transaction involving commodity options, flutures contracts or similar transactions; and No Debtor shall, without the express prior written consent of the Investment Agent, make any payment or transfer with respect to any lien or Financial Indebtedness incurred by the Purpose and Limitations Local Rule 4001-2(a)(9): Fed. R. Banhr. P. 4001(c)(1/18) DIP Agreement, § 12.12 Termination Date: Maturity Local Rule 4001-2(a)(10): Fed. R. Banhr. P. 4001(c)(1/18) DIP Transaction Documents, or which results in the occurrence of a default. Local Rule 4001-2(a)(10): Fed. R. Banhr. P. 4001(c)(1/18) DIP June and (vii) the date of the conversion or dismissal of any of the Obligors' Chapter 11 Cases. Maturity Date: Six month maturity date extension, subject to approval by the Investment Agent. Each of the follo		
Collateral in the Chapter 11 Cases which is pari passus with or senior to the claims of the Investment Agent and the DIP Participants granted pursuant to the DIP Agreement, any DIP Transaction Document and/or the Interim Order (or the Final Order, as applicable); No Obligor (A) shall, or shall permit any of its subsidiaries to, cancel any claim or debt owing to it, except for reasonable consideration negotiated on an arm's length basis and in the ordinary course of business; (B) shall, or shall permit any of its subsidiaries to enter into or become bound by any effective agreement, instrument, indenture or other obligation that could directly or indirectly restrict, prohibit or require the consent of any person with respect to the payment of dividends or distributions or the making or repayment of intercompany Joans by a subsidiary of any Obligor to any Obligor or between Obligors; (C) shall engage in any transaction involving commodity options, futures contracts or similar transactions; and No Debtor shall, without the express prior written consent of the Investment Agent any payment or transfer with respect to any lien or Financial Indebtedness incurred or arising prior to the Petition Date (except pursuant to a chapter 11 plan). Local Rule 4001-2(a)(9): Fed. R. Bankr. P. 4001(c)(1/18) DIP Agreement, § 12.12 **Termination Date** Maturity Local Rule 4001-2(a)(10): Fed. R. Bankr. P. 4001(c)(1/18) **Termination Date** Maturity Local Rule 4001-2(a)(10): Fed. R. Bankr. P. 4001(c)(1/18) **Termination Date** Maturity Local Rule 4001-2(a)(10): Fed. R. Bankr. P. 4001(c)(1/18) **Termination Date** Maturity Local Rule 4001-2(a)(10): Fed. R. Bankr. P. 4001(c)(1/18) **Termination Date** Maturity Local Rule 4001-2(a)(10): Fed. R. Bankr. P. 4001(c)(1/18) **Termination Date** Maturity Local Rule 4001-2(a)(10): Fed. R. Bankr. P. 4001(c)(1/18) **Any date on which the sale of all or substantially all of the Obligors is sest is consummated, (v) the date of termination of the DIP Agreement, (vi) the effecti		• Ensure that no member of the Group shall open a deposit account;
except for reasonable consideration negotiated on an arm's length basis and in the ordinary course of business; (B) shall, or shall permit any of its subsidiaries to enter into or become bound by any effective agreement, instrument, indenture or other obligation that could directly or indirectly restrict, prohibit or require the consent of any person with respect to the payment of dividends or distributions or the making or repayment of intercompany loans by a subsidiary of any Obligor to any Obligor or between Obligors; (C) shall engage in any transaction involving commodity options, futures contracts or similar transactions; and • No Debtor shall, without the express prior written consent of the Investment Agent, make any payment or transfer with respect to any lien or Financial Indebtedness incurred or arising prior to the Petrition Date (except pursuant to a chapter I1 plant). **Formulations** **Local Rule 4001-2(a)(19): Fed. R. Bankr. P. 4001(c)(1)(B) **DIP Agreement, § 12.12* **DIP Transaction proceeds shall not be used (x) to make any prepetition payment, except as provided in the SCB Order or (y) for any act which has the effect of adversely modifying or compromising the rights and remedies of the Investment Agent or any DIP Participant under the DIP Transaction Documents, or which results in the occurrence of a default. **Local Rule 4001-2(a)(10); Fed. R. Bankr. P. 4001(c)(1)(B) **DIP Agreement, § 1.1, Definition of Termination Date and Manurity: § 2.3 **Local Rule 4001-2(a)(10); Fed. R. Bankr. P. 4001(c)(1)(B) **DIP Agreement, § 1.1, Definition of Termination Date and Manurity: § 2.3 **Event of Default** **Local Rule 4001-2(a)(10); Fed. R. Bankr. P. 4001(c)(1)(B) **DIP Agreement, § 1.1, Optimition of Termination Date: Six months after the Effective Date (as defined below), (ii) 30 days (or such later adule as the Investment Agent or any birt service of a consummated, (v) the date of termination of the DIP Agreement, (vi) the effective date of a chapter I1 plan and (vii) the date of the con		Collateral in the Chapter 11 Cases which is pari passu with or senior to the claims of the Investment Agent and the DIP Participants granted pursuant to the DIP Agreement, any DIP Transaction
Purpose and Limitations Local Rule 4001-2(a)(9): Fed. R. Bankr. P. 4001(c)(1)(B) DIP Agreement, § 12.12 Termination Date: Maturity Local Rule 4001-2(a)(10): Fed. R. Bankr. P. 4001(c)(1)(B) Transaction proceeds shall not be used (x) to make any prepetition payment, except as provided in the SCB Order or (y) for any act which has the effect of adversely modifying or compromising the rights and remedies of the Investment Agent or me DIP Transaction Documents; (ii) to satisfy any payment or bligations due and payable to the Investment Agent or the DIP Participants under the DIP Transaction Documents; and (iii) to pay any amounts due in accordance with the DIP Budget; DIP Transaction Documents; and (iii) to pay any amounts due in accordance with the DIP Budget; DIP Transaction Documents; and (iii) to pay any amounts due in accordance with the DIP Budget; DIP Transaction Documents; and (iii) to pay any amounts due in accordance with the DIP Budget; DIP Transaction Documents; (ii) to satisfy any payment or blight of the DIP Participant under the DIP Transaction Documents; and (iii) to pay any amounts due in accordance with the DIP Transaction Documents; (ii) to make any prepetition payment, except as provided in the SCB Order or (y) for any act which has the effect of adversely modifying or compromising the rights and remedies of the Investment Agent or any DIP Participant under the DIP Transaction Documents; (ii) and the provided in the SCB Order or (y) for any act which has the effect of adversely modifying or compromising the rights and remedies of the Investment Agent or any DIP Participant under the DIP Transaction Documents and the provided in the SCB Order or (y) for any act which has the effect of adversely modifying or compromising the rights and remedies of the Investment Agent or any DIP Participant under the DIP Participant under the DIP Transaction Documents and the DIP Agreement, with the DIP Baget and the DIP Agreement, with the DIP Baget and the DIP Agreement, with the DIP Baget and the DIP Ag		except for reasonable consideration negotiated on an arm's length basis and in the ordinary course of business; (B) shall, or shall permit any of its subsidiaries to enter into or become bound by any effective agreement, instrument, indenture or other obligation that could directly or indirectly restrict, prohibit or require the consent of any person with respect to the payment of dividends or distributions or the making or repayment of intercompany loans by a subsidiary of any Obligor to any Obligor or between Obligors; (C) shall engage in any transaction involving commodity options,
Investment Agent in the negotiation, preparation and administration of the DIP Transaction Documents; (ii) to satisfy any payment obligations due and payable to the Investment Agent or the DIP Participants under the DIP Transaction Documents; and (iii) to pay any amounts due in accordance with the DIP Budget; DIP Transaction proceeds shall not be used (x) to make any prepetition payment, except as provided in the SCB Order or (y) for any act which has the effect of adversely modifying or compromising the rights and remedies of the Investment Agent or any DIP Participant under the DIP Transaction Documents, or which results in the occurrence of a default. **Termination Date;** Maturity **Local Rule 4001-2(a)(10);** Fed. R. Bankr. P. 4001(c)(1)(B) DIP Agreement, § 1.1, Definition of Termination Date and Maturity; § 2.3 **Maturity Date: Six months after the Effective Date (as defined in the DIP Agreement) with the possibility of a six month maturity date extension, subject to approval by the Investment Agent. **Events of Default** Local Rule 4001-2(a)(10); Fed R. Bankr. P. 4001(c)(1)(B) **Events of Default** Local Rule 4001-2(a)(10); Fed R. Bankr. P. 4001(c)(1)(B) **AIHL fails to make payments within 5 business days of its due date; **AIHL, or any guarantor, fails to perform or comply with certain other DIP Agreement terms or covenants; **Any material representation made in any DIP Transaction Documents that proves to have been false; **Any material representation made in any DIP Transaction Documents that proves to have been false; **Any of the Obligors' Chapter 11 Cases are dismissed or converted to chapter 7; **A Court order granting a superpriority claim or other claim that is pari passu with or senior to the claims or liens of the Investment Agent and the DIP Participants;		payment or transfer with respect to any lien or Financial Indebtedness incurred or arising prior to
Local Rule 4001-2(a)(9): Fed. R. Bankr. P. 4001(c)(1)(B) DIP Agreement, § 12.12 Termination Date: Maturity Local Rule 4001-2(a)(10): Fed. R. Bankr. P. 4001(c)(1)(B) DIP Agreement, § 1.1. Definition of Termination Date and Maturity; § 2.3 Events of Default Local Rule 4001-2(a)(10): Fed. R. Bankr. P. 4001(c)(1)(B) Events of Default Local Rule 4001-2(a)(10): Fed. R. Bankr. P. 4001(c)(1)(B) DIP Agreement, § 1.1. Definition of Termination Date and Maturity; § 2.3 Events of Default Local Rule 4001-2(a)(10): Fed. R. Bankr. P. 4001(c)(1)(B) DIP Agreement, § 1.1. Definition of Termination Date and Maturity; § 2.3 Events of Default Local Rule 4001-2(a)(10): Fed. R. Bankr. P. 4001(c)(1)(B) Termination of Termination Date and Maturity; § 2.3 Events of Default Local Rule 4001-2(a)(10): Fed. R. Bankr. P. 4001(c)(1)(B) Termination of Termination Date and Maturity: § 2.3 Events of Default Local Rule 4001-2(a)(10): Fed. R. Bankr. P. 4001(c)(1)(B) DIP Agreement, § 1.4 AIHL fails to make payments within 5 business days of its due date; All L., or any guarantor, fails to perform or comply with certain other DIP Agreement terms or covenants; Any expropriation, attachment, sequestration, distress or execution affects any asset or assets of any member of the Group (other than a Debtor); Any of the Obligors' Chapter 11 Cases are dismissed or converted to chapter 7; A Court order granting a superpriority claim or other claim that is pari passu with or senior to the claims or liens of the Investment Agent on the DIP Participants;	Purpose and Limitations	
provided in the SCB Order or (y) for any act which has the effect of adversely modifying or compromising the rights and remedies of the Investment Agent or any DIP Participant under the DIP Transaction Documents, or which results in the occurrence of a default. **Termination Date:** Maturity** **Local Rule 4001-2(a)(10); Fed. R. Bankr. P. 4001(c)(1)(B)** **DIP Agreement. § 1.1, Definition of Termination Date and Maturity; § 2.3* **Events of Default** **Local Rule 4001-2(a)(10): Fed R. Bankr. P. 4001(c)(1)(B)** **DIP Agreement. § 1.1, Definition of Termination Date and Maturity; § 2.3* **Events of Default** **Local Rule 4001-2(a)(10): Fed R. Bankr. P. 4001(c)(1)(B)** **DIP Agreement. § 1.4* **Events of Default** **Local Rule 4001-2(a)(10): Fed R. Bankr. P. 4001(c)(1)(B)** **DIP Agreement. § 1.4* **Events of Default** **Local Rule 4001-2(a)(10): Fed R. Bankr. P. 4001(c)(1)(B)** **DIP Agreement. § 1.4* **AIHL fails to make payments within 5 business days of its due date; **AIHL, or any guarantor, fails to perform or comply with certain other DIP Agreement terms or covenants; **Any material representation made in any DIP Transaction Documents that proves to have been false; **Any expropriation, attachment, sequestration, distress or execution affects any asset or assets of any member of the Group (other than a Debtor); **Any of the Obligors' Chapter 11 Cases are dismissed or converted to chapter 7; **A Court order granting a superpriority claim or other claim that is pari passu with or senior to the claims or liens of the Investment Agent and the DIP Participants;	Fed. R. Bankr. P.	Documents; (ii) to satisfy any payment obligations due and payable to the Investment Agent or the DIP Participants under the DIP Transaction Documents; and (iii) to pay any amounts due in accordance with the DIP Budget;
Maturity Local Rule 4001-2(a)(10); Fed. R. Bankr. P. 4001(c)(1)(B) DIP Agreement, § 1.1, Definition of Termination Date and Maturity; § 2.3 Events of Default Local Rule 4001-2(a)(10); Fed R. Bankr. P. 4001(c)(1)(B) Events of Default Local Rule 4001-2(a)(10); Fed R. Bankr. P. 4001(c)(1)(B) DIP Agreement, § 1.1, Definition of Termination Date and Maturity; § 2.3 Events of Default Local Rule 4001-2(a)(10); Fed R. Bankr. P. 4001(c)(1)(B) DIP Agreement, § 1.4 Each of the following is an "Event of Default" AIHL, or any guarantor, fails to perform or comply with certain other DIP Agreement terms or covenants; Any material representation made in any DIP Transaction Documents that proves to have been false; The commencement of any insolvency proceeding (other than with respect to the Debtors); Any expropriation, attachment, sequestration, distress or execution affects any asset or assets of any member of the Group (other than a Debtor); A Court order granting a superpriority claim or other claim that is pari passu with or senior to the claims or liens of the Investment Agent and the DIP Participants;	DIP Agreement, § 12.12	provided in the SCB Order or (y) for any act which has the effect of adversely modifying or compromising the rights and remedies of the Investment Agent or any DIP Participant under the
 AIHL fails to make payments within 5 business days of its due date; AIHL, or any guarantor, fails to perform or comply with certain other DIP Agreement terms or covenants; Any material representation made in any DIP Transaction Documents that proves to have been false; The commencement of any insolvency proceeding (other than with respect to the Debtors); Any expropriation, attachment, sequestration, distress or execution affects any asset or assets of any member of the Group (other than a Debtor); Any of the Obligors' Chapter 11 Cases are dismissed or converted to chapter 7; A Court order granting a superpriority claim or other claim that is pari passu with or senior to the claims or liens of the Investment Agent and the DIP Participants; 	Maturity Local Rule 4001-2(a)(10); Fed. R. Bankr. P. 4001(c)(1)(B) DIP Agreement, § 1.1, Definition of Termination	date as the Investment Agent may agree in its sole discretion) after entry of the Interim Order if the Final Order has not been entered prior thereto, (iii) any date on which the DIP Facility is reduced to zero, (iv) any date on which the sale of all or substantially all of the Obligors' assets is consummated, (v) the date of termination of the DIP Agreement, (vi) the effective date of a chapter 11 plan and (vii) the date of the conversion or dismissal of any of the Obligors' Chapter 11 Cases. • Maturity Date: Six months after the Effective Date (as defined in the DIP Agreement) with the
 AIHL, or any guarantor, fails to perform or comply with certain other DIP Agreement terms or covenants; Any material representation made in any DIP Transaction Documents that proves to have been false; The commencement of any insolvency proceeding (other than with respect to the Debtors); Any expropriation, attachment, sequestration, distress or execution affects any asset or assets of any member of the Group (other than a Debtor); Any of the Obligors' Chapter 11 Cases are dismissed or converted to chapter 7; A Court order granting a superpriority claim or other claim that is pari passu with or senior to the claims or liens of the Investment Agent and the DIP Participants; 	Events of Default	Each of the following is an "Event of Default"
 Any material representation made in any DIP Transaction Documents that proves to have been false; The commencement of any insolvency proceeding (other than with respect to the Debtors); Any expropriation, attachment, sequestration, distress or execution affects any asset or assets of any member of the Group (other than a Debtor); Any of the Obligors' Chapter 11 Cases are dismissed or converted to chapter 7; A Court order granting a superpriority claim or other claim that is <i>pari passu</i> with or senior to the claims or liens of the Investment Agent and the DIP Participants; 	Local Rule 4001-2(a)(10);	
false; • The commencement of any insolvency proceeding (other than with respect to the Debtors); • Any expropriation, attachment, sequestration, distress or execution affects any asset or assets of any member of the Group (other than a Debtor); • Any of the Obligors' Chapter 11 Cases are dismissed or converted to chapter 7; • A Court order granting a superpriority claim or other claim that is <i>pari passu</i> with or senior to the claims or liens of the Investment Agent and the DIP Participants;		
 Any expropriation, attachment, sequestration, distress or execution affects any asset or assets of any member of the Group (other than a Debtor); Any of the Obligors' Chapter 11 Cases are dismissed or converted to chapter 7; A Court order granting a superpriority claim or other claim that is <i>pari passu</i> with or senior to the claims or liens of the Investment Agent and the DIP Participants; 	DIP Agreement, §14	
 member of the Group (other than a Debtor); Any of the Obligors' Chapter 11 Cases are dismissed or converted to chapter 7; A Court order granting a superpriority claim or other claim that is <i>pari passu</i> with or senior to the claims or liens of the Investment Agent and the DIP Participants; 		
• A Court order granting a superpriority claim or other claim that is <i>pari passu</i> with or senior to the claims or liens of the Investment Agent and the DIP Participants;		
claims or liens of the Investment Agent and the DIP Participants;		
Court order granting relief from the automatic stay to permit foreclosure on a lien;		claims or liens of the Investment Agent and the DIP Participants;
		Court order granting relief from the automatic stay to permit foreclosure on a lien;

12-11076-shl Doc 690 Filed 12/04/12 Entered 12/04/12 18:01:19 Main Document Pg 14 of 29

	Entry of the Interim Order and Final Order shall not have occurred by dates certain;
	The management of AIHL is displaced or any issued shares of AIHL are seized;
	Any DIP Transaction Document or material provision thereof ceases to be legal or valid;
	It becomes unlawful for any Obligor to comply with its obligations under the DIP Agreement;
	• The sale of all or substantially all of the Debtors' assets without the Investment Agent's consent;
	• The filing or confirmation of a chapter 11 plan or disclosure statement that (i) is not acceptable to the Investment Agent, or (ii) does not provide for repayment in full in cash of all of the DIP Obligations (unless the DIP Facility is converted to an exit facility);
	A Material Adverse Effect;
	• A substantial change in the nature of AIHL's business from that carried on as of the Effective Date;
	A Change of Control (as defined in the DIP Agreement); and
	Events of Default tied to Cayman Islands liquidation proceedings of AIHL.
Automatic Stay & Remedies Local Rule 4001-2(a)(10); Fed. R. Bankr. P. 4001(c)(1)(B)(iv) Interim Order, ¶ 14	The automatic stay provisions of section 362 of the Bankruptcy Code are vacated and modified to the extent necessary, without the need for any further order of the Court to permit the Investment Agent and/or Security Agent to exercise, upon not less than seven (7) days' written notice to the Debtors following the occurrence and continuation of any Event of Default under the DIP Transaction Documents, all rights and remedies under the DIP Transaction Documents.
Change of Control Local Rule 4001-2(a)(11) DIP Agreement, §§ 14.17	A Change of Control shall constitute an Event of Default.
Repayment Local Rule 4001-2(a)(13)	• 120 days after the Effective Date (as defined in the DIP Agreement), AIHL shall prepay the outstanding amount under the DIP Transaction in an aggregate amount equal to the difference between (i) \$25,000,000 and (ii) any previous commitment reductions.
DIP Agreement, §§ 9.1.2- 9.1.5	• In the event of (i) a cancellation or termination of the a DIP Facility, (ii) the acceleration of the DIP Obligations, or (iii) the reduction of the commitment to zero, AIHL shall repay the DIP Facility.
	If the DIP Obligations exceed \$150 million, AIHL shall immediately prepay the excess.
Joint Liability	All Obligors are jointly and severally liable for the DIP Obligations.
Local Rule 4001-2(a)(14); Fed. R. Bankr. P. 4001-2(e)	Falcon Gas is not liable for any of the DIP Obligations.
Indemnification Fed. R. Bankr. P. 4001(c)(1)(B)(ix) DIP Agreement, § 16.1	AIHL shall indemnify the Investment Agent, the Security Agent, the Arranger and the DIP Participants and their respective directors, officers, employees and agents (each, an " <i>Indemnified Party</i> ") on demand against any actual costs, loss, liability or expense which the Indemnified Party has sustained or incurred as a consequence of: (i) the failure of AIHL to make payment on the due date of any sum due to the Investment Agent under any DIP Transaction Document; (ii) the failure of AIHL to send an acceptance to the Investment Agent's offer to purchase or sell the Commodities; (iii) the occurrence of any Event of Default; (iv) any actions, claims, proceedings, liabilities, losses, damages, penalties, judgments, suits, costs and expenses that may be imposed on, incurred by, asserted of or claimed and howsoever arising out of the DIP Agreement, any of the other DIP Transaction Documents or any action taken or omitted by any of them; and (v) any prepayment not being made, provided that, indemnification is not available to the extent losses arise from gross negligence or willful misconduct.

Waivers and Consents Fed. R. Bankr. P. $4001(c)(1)(B)(x)$ Interim Order, ¶ 15	Subject to entry of a Final Order, the Debtors (for themselves and their estates) irrevocably waive and relinquish any rights they may have under section 506(c) of the Bankruptcy Code with respect to the Collateral (as defined in the Interim Order).
Exit Fortress Commitment Letter	As provided in the Fortress Commitment Letter, the DIP Facility may be converted to an Murabaha exit facility, subject to the Debtors' satisfaction of certain conditions precedent

DEBTORS' PREPETITION SECURED DEBT⁷

- available by Standard Chartered Bank ("SCB"): (i) a \$50 million facility dated May 30, 2011, of which approximately \$46.6 million was outstanding at the Petition Date and which matured on March 28, 2012 (the "SCB May 2011 Facility"); and (ii) a \$50 million facility dated December 22, 2011, of which approximately \$50.1 million was outstanding at the Petition Date and which matured on March 28, 2012 (the "SCB December 2011 Facility," and together with the SCB May 2011 Facility, the "SCB Facilities").
- 12. The SCB May 2011 Facility is guaranteed by each of AIHL, ALTHL and WindTurbine. These guarantees are secured by: (i) a first priority pledge of AIHL's shares in ALTHL; (ii) a first priority pledge of ALTHL's shares in WindTurbine; and (iii) a second priority pledge of ALTHL's shares in AEID II and RailInvest.
- 13. The SCB December 2011 Facility is guaranteed by each of AIHL, ALTHL, WindTurbine, AEID II, and RailInvest. These guarantees are secured by: (i) a second priority pledge of AIHL's shares in ALTHL; (ii) a first priority pledge of ALTHL's shares in AEID II and RailInvest; and (iii) a second priority pledge in ALTHL's shares in WindTurbine. Pursuant

The descriptions of the Debtors' prepetition facilities and the collateral securing such facilities provided herein do not constitute, and should not be construed as, an admission by the Debtors regarding the validity, permissibility, enforceability, perfection or amount of any obligation, claim, guarantee, lien, mortgage, pledge or other security interest, or any other fact with respect thereto.

to the SCB Facilities, SCB maintains a claim against AEID II, RailInvest and WindTurbine (collectively, the "SCB Guarantee Claims") and a lien on their equity.

14. On October 19, 2012, the Court entered its *Order pursuant to section 105(a) of* the Bankruptcy Code and Bankruptcy Rule 9019, Authorizing and Approving the Settlement with Standard Chartered Bank [Dkt. No. 587] (including the settlement term sheet annexed as "Exhibit 1" thereto, the "SCB Order").

PROPOSED DIP TRANSACTION

I. <u>Debtors' Need for Post-Petition Financing</u>

the Debtors have managed their businesses with cash on hand. Nonetheless, the Debtors, all of which are holding companies, have expended a significant amount of available cash to fund restructuring costs and to support the value of their portfolio companies and investments (collectively, the "Arcapita Group Investments"). Declaration of John Makuch in Support of Debtors' Motion for the Entry of Interim and Final Orders Pursuant to 11 U.S.C. §§ 105, 362, 363(b)(1), 363(m), 364(c)(1), 364(c)(2), 364(c)(3), and 364(e) and Bankruptcy Rules 4001 and 6004 (I) Authorizing Debtors (A) To Enter Into and Perform Under DIP Agreement, and (B) To Obtain Credit on a Secured Superpriority Basis, (II) Scheduling Final Hearing Pursuant to Bankruptcy Rules 4001(b) and (c) and (III) Granting Related Relief, annexed as Exhibit D hereto (the "Makuch Declaration") ¶ 4. The Debtors, through their direct and indirect subsidiaries, manage and maintain equity positions in numerous Arcapita Group Investments. These non-Debtor entities require cash from time to time to bridge them to monetization events.

In general, the Debtors' expenditures during the Chapter 11 Cases have been made with the support of both the Committee and the Joint Provisional Liquidators.

Makuch Declaration ¶¶ 16, 18, 19. As of November 17, 2012, the Debtors have funded \$63.2 million of deal funding expenses during the Chapter 11 Cases. Makuch Declaration ¶ 10.

- 16. As a result of these expenditures, and despite the Debtors' diligent cash management efforts, the Debtors' liquidity has decreased. Makuch Declaration ¶ 11. The net available cash to the Debtors at the Petition Date was approximately \$120.1 million. Makuch Declaration ¶ 11. As of November 17, 2012, liquidity dropped to less than \$24 million. Makuch Declaration ¶ 11. Thus, the Debtors require the DIP Transaction to maintain their businesses without disruption during the Chapter 11 Cases, to maximize the value of their assets and the estates and to bridge the cases to emergence. Makuch Declaration ¶ 19.
- 17. Furthermore, the Debtors have determined that they require the Interim Amount immediately in order to satisfy the Debtors' short-term financing needs. Makuch Declaration ¶ 17. The Debtors, therefore, submit that approval of their request for the Interim Amount, in the principal amount of \$25 million, pursuant to the DIP Transaction Documents and pending entry of the Final Order, is necessary to avoid immediate irreparable harm to the estates.

II. Structure of a Murabaha

21. One of the basic tenets of Islamic banking is the prohibition against interest.¹⁰ Nonetheless, Shari'ah compliant financings can be structured to yield returns that are similar to those in a conventional financing. Shari'ah compliant financings rely on structures, such as deferred payment obligations, to achieve economic effects similar to those of a conventional financing.

Net cash excludes approximately \$35 million held in placement accounts in Bahraini financial institutions. This cash may not be accessed without prolonged litigation against parties which may not have minimum contacts with the United States. To date, the Debtors' attempts to negotiate for release of the cash held in such placements accounts have been unsuccessful.

Emad H. Khalil, *An Overview of the* Sharia'a *Prohibition of* Riba, *in* INTEREST IN ISLAMIC ECONOMICS: UNDERSTANDING *RIBA* 53, 53 (Abdulkader Thomas ed., 2006).

- 22. A Murabaha is a "cost-plus financing" trading arrangement whereby a party that requires cash (the "borrower"), arranges for another party (the "lender"), often a financial institution, to purchase certain commodities and then sell such commodities to the "borrower" for a higher price effectively cost price plus a mark-up, to give a return on investment. The price payable by the "borrower" is deferred creating a debt obligation. Murabahas are a common structure in Islamic financing.¹¹
- and sale of the commodities: (a) the "borrower" requests the "lender" to purchase a certain amount of commodities to be on-sold by the "lender" to the "borrower"; (b) the "lender" purchases the commodities and sells them to the "borrower" who agrees to buy them for cost plus an agreed amount of profit (which often is tied to LIBOR or some other index) payable on a future date (the "*Deferred Payment Date*"); (c) the "borrower," on the same day, sells the commodities to a third party (typically a commodities broker) for cash; and (d) on the Deferred Payment Date, the "borrower" pays the "lender" the cost price plus the agreed profit, and the Murabaha financing is concluded. In a Murabaha financing, one purchase contract may not cover the entire expected duration of the facility. The Deferred Payment Date for each contract may be a short time after execution of the initial commodities sale, and after the Deferred Payment Date, the parties may enter into a new replacement purchase agreement, the proceeds of which will be used to pay off the amounts owing under the old purchase contract.
- 24. In addition to the purchase contract, the "borrower," guarantors, investment agent, security agent and participants typically enter into a separate investment agency agreement.

 Under such agreement, the participants the parties who advance the cash used to initially fund

Sh. Yusuf Talal DeLorenzo, *Introduction to Understanding* Riba, *in* INTEREST IN ISLAMIC ECONOMICS: UNDERSTANDING *RIBA* 1, 6 (Abdulkader Thomas ed., 2006).

the Murabaha facility – agree to appoint the investment agent as their agent in the provision of the Murabaha facility. The investment agency agreement governs the actions of the investment agent in such capacity, including after an event of default.

- 25. The DIP Facility follows the typical structure described above. Clause 2.2 of the DIP Agreement directs the Investment Agent to purchase the Commodities and sell them to the Debtors at a Deferred Sale Price on deferred payment terms. DIP Agreement § 2.2. Because, as discussed above, the duration of each purchase contract is short (one-month), the DIP Agreement establishes conditions precedent to the parties' entry into replacement agreements. DIP Agreement § 3.2. And, the procedures of offer, acceptance and ultimate repayment of each contract are set forth in Clause 5. The Investment Agency Agreement, in Clause 2, directs the appointment of the Investment Agent and the terms of the DIP Participant's indemnification of the agent for acts taken in that capacity. Investment Agency Agreement § 2.
- 26. Thus, the DIP Facility is not a loan given on interest; rather it is a sale of a commodity for a deferred price with a profit component. But the end result is similar insofar as the Debtors will obtain liquidity and will incur an obligation to pay the liquidity provider.

III. Marketing and Negotiations Leading to the DIP Transaction

26. In connection with the DIP Transaction, the Debtors engaged in an extensive and thorough solicitation and negotiation process. The *Declaration of Homer Parkhill in Support of Debtors' Motion for the Entry of Interim and Final Orders Pursuant to 11 U.S.C. §§ 105, 362, 363(b)(1), 363(m), 364(c)(1), 364(c)(2), 364(c)(3), and 364(e) and Bankruptcy Rules 4001 and 6004 (I) Authorizing Debtors (A) To Enter Into and Perform Under DIP Agreement, and (B) To Obtain Credit on a Secured Superpriority Basis, (II) Scheduling Final Hearing Pursuant to Bankruptcy Rules 4001(b) and (c) and (III) Granting Related Relief, annexed as <i>Exhibit E* hereto (the "*Parkhill Declaration*") provides an in depth summary of the solicitation process.

- 27. The Debtors initially selected the Original DIP Provider to move forward to final documentation. Parkhill Declaration ¶ 8. After the Initial Hearing was adjourned, the Debtors continued negotiations with multiple parties, including the Original DIP Provider, for a firm commitment for additional liquidity on the best available terms. Parkhill Declaration ¶ 10. The Debtors received, analyzed, compared and negotiated the terms of various competing debtor in possession financing proposals in what essentially amounted to an auction. Parkhill Declaration ¶ 10. The Debtors consulted with the Committee and the Joint Provisional Liquidators throughout the solicitation. Parkhill Declaration ¶ 10.
- On October 18, 2012, Fortress delivered to the Debtors its first proposal. Parkhill Declaration ¶ 12. From that date until the approval of the final version of the Fortress Commitment Letter on November 7, 2012, Fortress, the Debtors and the Committee engaged in extensive, good-faith arms' length negotiations relating to the terms of the DIP Transaction. Parkhill Declaration ¶ 12. At the same time, the Debtors continued negotiations with other parties. Parkhill Declaration ¶ 12. After extensive, multi-faceted negotiations, the Debtors ultimately agreed to a binding commitment from Fortress to provide up to \$150 million in Shari'ah compliant financing (the "Fortress Commitment"), approved by this Court on November 7, 2012. Parkhill Declaration ¶¶ 15, 16.
- 29. The Debtors were unable to obtain the same amount of post-petition financing from an alternative DIP provider or providers on more favorable terms than those set forth in the proposed DIP Transaction. The alternative proposals the Debtors received were either (i) for a significantly lesser full committed amount which, the Debtors and their advisors determined, would not be sufficient to preserve estate values and to maximize creditor recoveries, (ii) on less favorable and more restrictive legal terms, (iii) on worse economic or business terms, or

(iv) subject to further due diligence and/or internal credit committee approval, resulting in greater execution risk. Parkhill Declaration ¶ 17. The DIP Transaction constitutes the most favorable financing opportunity available.

IV. The DIP Transaction Accords with the SCB Order

30. The DIP Transaction will not modify, alter, amend or supersede the SCB Order, including in connection with any adequate protection provided to SCB thereunder. The SCB Order remains in full force and effect. Interim Order, ¶ 11.

BASIS FOR RELIEF REQUESTED

I. The Debtors' Entry into the DIP Transaction Documents is an Exercise of the Debtors' Sound Business Judgment

- 31. A bankruptcy court should grant the debtor wide deference to act in accordance with its sound business judgment in obtaining financing. *See In re Barbara K. Enters., Inc.*, 2008 WL 2439649, at *14 (Bankr. S.D.N.Y. June 16, 2008) (noting that courts defer to a debtor's sound business judgment "so long as a request for financing does not 'leverage the bankruptcy process' and unfairly cede control of the reorganization to one party in interest."); *In re Ames Dep't Stores, Inc.*, 115 B.R. 34, 40 (Bankr. S.D.N.Y. 1990) ("cases consistently reflect that the court's discretion under section 364 [of the Bankruptcy Code] is to be utilized on grounds that permit [a debtor's] reasonable business judgment to be exercised so long as the financing agreement does not contain terms that leverage the bankruptcy process and powers or its purpose is not so much to benefit the estate as it is to benefit a party-in-interest.").
- 32. The Debtors' execution of the DIP Transaction Documents and performance under such documents and the DIP Orders comprise an exercise of their sound business judgment, warranting approval of this DIP Motion. Approval of the DIP Agreement will provide the Debtors with immediate and ongoing access to the cash proceeds of the DIP Transaction to

pay their current and ongoing operating expenses, including post-petition wages and salaries, and to make scheduled deal funding payments required to preserve the value of the Debtors' assets.

33. Furthermore, as noted above, the terms and conditions of the DIP Transaction are fair and reasonable, and were negotiated extensively among the Debtors and the Investment Agent after considerable marketing was performed by the Debtors and their advisors. The DIP Agreement is on the best available terms. Accordingly, consummation of the DIP Transaction is in the best interest of the Debtors' estates, their creditors and all parties in interest in the Chapter 11 Cases and is consistent with the Debtors' exercise of their fiduciary duties.

II. The Court Should Authorize the Debtors to Obtain Post-Petition Financing on a Secured and Superpriority Basis.

- 34. Pursuant to this DIP Motion, the Debtors seek authority to incur financing secured, in part, by a first lien on certain unencumbered property and a second lien on the existing pledges in favor of SCB. The proposed DIP Transaction is non-priming.
- 35. Section 364(c) of the Bankruptcy Code provides, among other things, that if a debtor is unable to obtain unsecured credit allowable as an administrative expense under section 503(b)(1) of the Bankruptcy Code, the Court may authorize the debtor to obtain credit or incur debt (i) with priority over any and all administrative expenses as specified in section 503(b) or 507(b) of the Bankruptcy Code, (ii) secured by a lien on property of the estate that is not otherwise subject to a lien or (iii) secured by a junior lien on property of the estate that is subject to a lien. 11 U.S.C. § 364(c).
- 36. A debtor seeking to satisfy the requirements of section 364(c) need only demonstrate "by a good faith effort that credit was not available" to the debtor on an unsecured or administrative expense basis. *Bray v. Shenandoah Fed. Sav. & Loan Ass'n (In re Snowshoe Co.)*, 789 F.2d 1085, 1088 (4th Cir. 1986). "The statute imposes no duty to seek credit from

every possible lender before concluding that such credit is unavailable." *Id.*; *see also Pearl-Phil GMT (Far East) Ltd. v. Caldor Corp.*, 266 B.R. 575, 584 (S.D.N.Y. 2001) (superpriority administrative expenses authorized where debtor could not obtain credit as an administrative expense). When few lenders are likely to be able and willing to extend the necessary credit to a debtor, "it would be unrealistic and unnecessary to require [the debtor] to conduct such an exhaustive search for financing." *In re Sky Valley, Inc.*, 100 B.R. 107, 113 (Bankr. N.D. Ga. 1988), *aff'd sub nom., Anchor Savs. Bank FSB v. Sky Valley, Inc.*, 99 B.R. 117, 120 n.4 (N.D. Ga. 1989); *see also Ames*, 115 B.R. at 40.

37. As noted earlier, the Debtors engaged in months of marketing with respect to the DIP Transaction. No potential DIP provider was willing to advance money on better terms without receiving security interests on the Debtor Obligors' assets. Certain proposals contemplated a grant of priming liens (which no doubt would result in a contested DIP hearing). The grant of collateral and superpriority claims to Fortress thus satisfies section 364 of the Bankruptcy Code.

III. The Court Should Authorize the Debtors to Sell the Commodities

38. Section 363(b) of the Bankruptcy Code provides that a debtor "after notice and a hearing, may use, sell, or lease, other than in the ordinary course of business, property of the estate." 11 U.S.C. § 363(b)(1). A debtor must demonstrate a sound business justification for a sale or use of assets outside the ordinary course of business. *See, e.g., Licensing by Paolo, Inc. v. Sinatra (In re Gucci)*, 126 F.3d 380, 387 (2d Cir. 1997); *Comm. of Equity Sec. Holders v. Lionel Corp.* (*In re Lionel Corp.*), 722 F.2d 1063, 1070 (2d Cir. 1983). Further, "[w]here the debtor articulates a reasonable basis for its business decisions (as distinct from a decision made arbitrarily or capriciously), courts will generally not entertain objections to the debtor's conduct." *Comm. of Asbestos-Related Litigants and/or Creditors v. Johns-Manville Corp.* (*In re*

Johns-Manville Corp.), 60 B.R. 612, 616 (Bankr. S.D.N.Y. 1986). In addition, section 105(a) of the Bankruptcy Code confers upon the Court broad equitable powers to fashion relief in accordance with the policies underlying the Bankruptcy Code.

39. As noted above, the DIP Transaction is a Shari'ah complaint transaction, structured as a purchase and sale of commodities with deferred payment terms. For the reasons set forth above, the DIP Transaction, including its component parts, is supported by sound business justifications and should be approved by this Court. To the extent the purchase and sale of the Commodities by the Debtors is out of the ordinary course of business, reasonable business judgment in support of the DIP Transaction exists to satisfy section 363(b) of the Bankruptcy Code.

IV. The Scope of the Carve Out is Appropriate

40. The proposed DIP Transaction subjects the security interests and administrative expense claims of the DIP Participants to the Carve Out. Carve outs for professional fees have been found to be reasonable and necessary to ensure that a debtor's estate and any statutory committee can retain assistance from counsel. *See Ames*, 115 B.R. at 40. The DIP Transaction does not directly or indirectly deprive the Debtors' estates or other parties in interest of possible rights and powers by restricting the services for which professionals may be paid in these cases. *Id.* at 38 (observing that courts insist on carve outs for professionals representing parties in interest because "[a]bsent such protection, the collective rights and expectations of all parties-in-interest are sorely prejudiced."). Additionally, the Carve Out protects against administrative insolvency during the Chapter 11 Cases by ensuring that assets remain for the payment of United States Trustee and Professional Persons notwithstanding the grant of superpriority and administrative liens and claims under the DIP Transaction.

V. The DIP Participants Should Be Deemed Good Faith Lenders under Section 364(e)

41. Section 364(e) of the Bankruptcy Code protects a good faith lender's right to collect on loans extended to a debtor, and its right in any lien securing those loans, even if the authority of the debtor to obtain such loans or grant such liens is later reversed or modified on appeal. Section 364(e) provides that:

The reversal or modification on appeal of an authorization under this section to obtain credit or incur debt, or of a grant under this section of a priority or a lien, does not affect the validity of any debt so incurred, or any priority or lien so granted, to an entity that extended such credit in good faith, whether or not such entity knew of the pendency of the appeal, unless such authorization and the incurring of such debt, or the granting of such priority or lien, were stayed pending appeal.

11 U.S.C. § 364(e).

42. As explained in detail herein and in the Commitment Letter Motion Supplement, the DIP Transaction Documents are the result of the Debtors' reasonable and informed determination that the Investment Agent offered the most favorable terms on which to obtain needed post-petition financing, and of extended arms'-length, good faith negotiations between the Debtors and the Investment Agent. The terms and conditions of the DIP Transaction Documents are fair and reasonable, and the proceeds of the DIP Transaction will be used only for the purposes that are permissible under the Bankruptcy Code.

VI. The Automatic Stay Should Be Modified on a Limited Basis

43. The relief requested herein contemplates a modification of the automatic stay (to the extent applicable) to permit: (a) the Debtors to grant the security interests, liens and superpriority claims described above and to perform such acts as may be requested to assure the perfection and priority of such security interests and liens; (b) the Debtors to incur the DIP Obligations under the DIP Transaction Documents and the DIP Orders; and (c) the DIP Participants to exercise and enforce their rights and remedies as provided in the DIP Orders. The

DIP Transaction Documents provide, however, that the DIP Participants must provide the Debtors and various other parties, including the United States Trustee and counsel to the Committee, with seven business days' notice before exercising any enforcement rights or remedies.

44. Stay modifications of this kind are ordinary and standard features of post-petition debtor in possession financing facilities and, in the Debtors' business judgment, are appropriate under the present circumstances, and were required to secure the DIP Participants' agreement to enter into the DIP Transaction Documents. *See, e.g., In re Great Atl. & Pac. Tea Co.*, Case No. 10-24549 (Bankr. S.D.N.Y Jan. 11, 2011) [Dkt. No. 479]; *In re Tronox Inc.*, Case No. 09-10156 (Bankr. S.D.N.Y. Jan. 15, 2010) [Dkt. No. 1115]; *In re Gen. Growth Props. Inc.*, Case No. 09-11977 (Bankr. S.D.N.Y. May 14, 2009) [Dkt. No. 527]; *In re Chemtura Corp.*, Case No. 09-11233 (Bankr. S.D.N.Y. Apr. 29, 2009) [Dkt. No. 281].

VII. The Debtors Require Immediate Access to the Interim Amount

45. Bankruptcy Rule 4001(c) provides that a hearing on a motion for authority to obtain credit may be held "no earlier than 14 days after service of the motion." Fed. R. Bankr. P. 4001(c)(2). The Court, however, is authorized to conduct an expedited hearing prior to the expiration of such 14-day period and to authorize the obtaining of post-petition credit to the extent necessary to avoid immediate and irreparable harm to the Debtors' estates. Pursuant to Bankruptcy Rule 4001(c), the Debtors request that the Court (a) authorize the Debtors' entry into the DIP Agreement on an interim basis pursuant to the terms of the Interim Order, and (b) schedule a final hearing to consider approval of the Debtors' entry into the DIP Agreement on a final basis pursuant to the terms of the Final Order. Pursuant to Local Rule 4001-2(g), pending a final hearing, the Debtors request authority to use the proceeds of the DIP Transaction as set forth in the DIP Agreement. Bankruptcy Rule 6004(h) provides that an "order authorizing the

use, sale, or lease of property . . . is stayed until the expiration of 14 days after entry of the order, unless the court orders otherwise." Fed. R. Bankr. P. 6004(h). The Debtors request that any order authorizing the Debtors' entry into the DIP Agreement on an interim basis be effective immediately by providing that the 14-day stay under Bankruptcy Rules 4001 and 6004(h) is waived.

- 46. The Debtors will suffer immediate and irreparable harm if the interim relief requested herein, including authorization to access the Interim Amount, upon entry of an Interim Order, is not granted. Makuch Declaration ¶ 17. Without the immediate use of the proceeds of the DIP Transaction, the Debtors may be unable to satisfy certain future expected deal funding expenses, potentially eviscerating Arcapita's equity value in such entities and threatening the Debtors' ability to reorganize. Makuch Declaration ¶ 18. Without access to the DIP Transaction proceeds, the Debtors' businesses may be jeopardized to the significant detriment of the Debtors' estates, their creditors and all of their constituents. Makuch Declaration ¶ 18.
- 47. In light of the foregoing, the Debtors submit that they have satisfied the requirements of Bankruptcy Rule 4001(c) and 6004(h) (a) to support immediate entry into the DIP Agreement, pending the entry of the Final Order and (b) to waive Bankruptcy Rules 4001 and 6004(h). Accordingly, to forestall the immediate and irreparable harm that will inure to the Debtors' estate, creditors and parties in interest absent Court approval of this DIP Motion, the Debtors respectfully request that the Court grant the relief requested herein and authorize the immediate entry into the DIP Agreement pursuant to the terms and conditions set forth in the Interim Order.

REQUEST FOR FINAL HEARING

48. Pursuant to Bankruptcy Rule 4001(c)(2), the Debtors request that the Court set a date for the Final DIP Hearing that is as soon as practicable, but in no event later than 20 days

following the entry of the Interim Order, and fix the time and date prior to the Final DIP Hearing for parties to file objections to the DIP Motion.

NOTICE

50. No trustee or examiner has been appointed in the Chapter 11 Cases. The Debtors have provided notice of filing of the DIP Motion by electronic mail, facsimile and/or overnight mail to: (i) the Office of the United States Trustee for the Southern District of New York, 33 Whitehall Street, 21st Floor, New York, New York 10004 (Attn: Richard Morrissey, Esq.); (ii) Milbank, Tweed, Hadley & McCloy LLP, 1 Chase Manhattan Plaza, New York, New York 10005 (Attn: Dennis F. Dunne, Esq., Abhilash M. Raval, Esq., and Evan R. Fleck, Esq.), counsel for the Committee; (iii) Skadden, Arps, Slate, Meagher & Flom LLP, 4 Times Square, New York, New York 10036 (Attn: Kenneth S. Ziman), counsel for Fortress Credit Corp., as Investment Agent; (iv) Dechert LLP, 1095 Avenue of the Americas, New York, New York 10036 (Attn: Brian E. Greer, Esq. and Nicole Herther-Spiro, Esq.), counsel to SCB, and (v) all parties listed on the Master Service List established in these Chapter 11 Cases. A copy of the DIP Motion is also available on the website of the Debtors' notice and claims agent, GCG, at www.gcginc.com/cases/arcapita.

NO PRIOR REQUEST

51. No prior application for the relief requested herein has been made to this or any other court.

WHEREFORE, the Debtors respectfully request that the Court grant the relief requested herein and such other and further relief as the Court may deem just and proper.

Dated: New York, New York

December 4, 2012

Respectfully submitted,

/s/ Michael A. Rosenthal

Michael A. Rosenthal (MR-7006) Craig H. Millet (admitted *pro hac vice*) Matthew J. Williams (MW-4081) Joshua Weisser (JW-0185)

GIBSON, DUNN & CRUTCHER LLP

200 Park Avenue

New York, New York 10166-0193

Telephone: (212) 351-4000 Facsimile: (212) 351-4035

ATTORNEYS FOR THE DEBTORS AND DEBTORS IN POSSESSION

12-11076-shl Doc 690-1 Filed 12/04/12 Entered 12/04/12 18:01:19 Exhibit A-Interim Order Pg 1 of 30

Exhibit A

Interim Order

12-11076-shl Doc 690-1 Filed 12/04/12 Entered 12/04/12 18:01:19 Exhibit A-Interim Order Pg 2 of 30

UNITED STATES BANKRUPTCY COURSOUTHERN DISTRICT OF NEW YORK	
IN RE:	x : : Chapter 11
ARCAPITA BANK B.S.C.(c), et al.,	: Case No. 12-11076 (SHL)
Debtors.	: Jointly Administered

INTERIM ORDER PURSUANT TO 11 U.S.C. §§ 105, 362, 363(b)(1), 363(m), 364(c)(1), 364(c)(2), 364(c)(3), AND 364(e) AND BANKRUPTCY RULES 4001 AND 6004 (I) AUTHORIZING DEBTORS (A) TO ENTER INTO AND PERFORM UNDER MURABAHA AGREEMENT, AND (B) TO OBTAIN CREDIT ON A SECURED SUPERPRIORITY BASIS, (II) SCHEDULING FINAL HEARING PURSUANT TO BANKRUPTCY RULES 4001(b) AND (c) AND (III) GRANTING RELATED RELIEF

Upon the motion (the "**Motion**"), dated December 4, 2012, of Arcapita Bank B.S.C.(c) ("**Arcapita**") and its affiliated debtors, each as debtor and debtor-in-possession (collectively, the "**Debtors**"), in the above-captioned chapter 11 cases (the "**Cases**") pursuant to sections 105, 362, 363(b)(1), 363(m), 364(c)(1), 364(c)(2), 364(c)(3), and 364(e) of title 11 of the United States Code, 11 U.S.C. §§ 101, et seq. (the "**Bankruptcy Code**"), Rules 2002, 4001, 6004 and 9014 of the Federal Rules of Bankruptcy Procedure (the "**Bankruptcy Rules**"), and Rule 4001 2 of the Local Rules of Bankruptcy Practice and Procedure of the United States Bankruptcy Court for the Southern District of New York (the "**Local Rules**"), requesting, among other things:

(1) authorization for Arcapita Investment Holdings Limited (the "Purchaser") to obtain a senior secured superpriority debtor-in-possession multiple-draw term Murabaha facility, in an aggregate principal amount up to \$150,000,000 (the "DIP Facility") (the availability of which shall be subject to the terms and conditions set forth in the Finance Documents (as defined in the

12-11076-shl Doc 690-1 Filed 12/04/12 Entered 12/04/12 18:01:19 Exhibit A-Interim Order Pg 3 of 30

DIP Agreement (as defined below) and including any exhibits thereto)), to be provided by Fortress Credit Corp. ("Fortress"), acting as agent (in such capacity, the "Investment Agent") for institutions participating in the DIP Facility (together with Fortress, the "Participants"), to be arranged by Fortress (in such capacity, the "Arranger"), and for all of the other Debtors except for Falcon Gas Storage Company, Inc. ("Falcon") (collectively, the "Debtor Guarantors") to guaranty all of the Purchaser's obligations under such DIP Facility;

- (2) authorization for the Purchaser to enter into and perform under a senior secured Superpriority Debtor-in-Possession Master Murabaha Agreement substantially in the form filed as Exhibit B to the Motion (as the same may be hereafter amended, restated, amended and restated, supplemented or otherwise modified from time to time in accordance with terms thereof or hereof, the "DIP Agreement"), and the other Finance Documents and to perform such other and further acts as may be reasonably required or appropriate in connection with the Finance Documents;
- (3) authorization for the Purchaser to enter into and perform under agreements with the Investment Agent for the purchase of the Commodities (as defined in the DIP Agreement) on the terms set forth in the Finance Documents, and for the Purchaser to sell and convey such Commodities to a third-party purchaser;
- (4) the grant of superpriority administrative expense claims to the Investment Agent pursuant to section 364(c)(1) of the Bankruptcy Code over any and all administrative expenses of any kind or nature, subject only to the Carve

12-11076-shl Doc 690-1 Filed 12/04/12 Entered 12/04/12 18:01:19 Exhibit A-Interim Order Pg 4 of 30

Out (as defined below), the Prior SCB Claims (as defined below) and as set forth herein and in the Finance Documents:

- (5) the grant of valid, enforceable, non-avoidable and fully perfected first priority priming liens pursuant to section 364(c)(2) on and security interests in all of the property, assets, and other interests in property and assets of Arcapita, the Purchaser, and Arcapita LT Holdings Limited ("ALTHL") not otherwise subject to a lien, whether such property is presently owned or after-acquired, and all other "property of the estate" (within the meaning of the Bankruptcy Code) of the Purchaser, Arcapita and ALTHL, of any kind or nature whatsoever, real or personal, tangible, intangible or mixed, now existing or hereafter acquired or created, whether existing prior to or arising after the Petition Date (as defined in the DIP Agreement), excluding actions for preferences, fraudulent conveyances, and other avoidance power claims under sections 544, 545, 547, 548, 550 and 553 of the Bankruptcy Code (the "Avoidance Actions") and the proceeds thereof, subject only to the Carve Out on the terms and conditions set forth herein and in the Finance Documents;
- (6) the grant of valid, enforceable, non-avoidable and fully perfected junior liens pursuant to section 364(c)(3) on and security interests in all of the property, assets, and other interests in property and assets of the Debtors (except for Falcon) that are subject to valid, perfected, and non-avoidable liens in existence on the Petition Date (or that are perfected after the Petition Date pursuant to 546(b)), whether such property is presently owned or after-acquired, and all other "property of the estate" (within the meaning of the Bankruptcy Code)

12-11076-shl Doc 690-1 Filed 12/04/12 Entered 12/04/12 18:01:19 Exhibit A-Interim Order Pg 5 of 30

of such Debtors, of any kind or nature whatsoever, real or personal, tangible, intangible or mixed, now existing or hereafter acquired or created, whether existing prior to or arising after the Petition Date, excluding Avoidance Actions and the proceeds thereof, subject only to the Carve Out on the terms and conditions set forth herein and in the Finance Documents;

- (7) subject only to, and effective upon entry of, a final order granting such relief and such other relief as provided herein and in such final order (the "**Final Order**"), the waiver of the Debtors' and their estates' rights to surcharge the Collateral (as defined below) pursuant to section 506(c) of the Bankruptcy Code;
- (8) pursuant to Bankruptcy Rule 4001, that an interim hearing (the "Interim Hearing") on the Motion be held before this Court to consider entry of the proposed interim order annexed to the Motion (the "Interim Order") (a) authorizing the Debtors on an interim basis, to borrow under the DIP Agreement in an aggregate amount of \$25 million and (b) granting the other relief described in the Finance Documents and this Interim Order; and
- (9) that this Court schedule a final hearing (the "**Final Hearing**") to consider entry of the Final Order approving the Motion and approving the Debtors' notice with respect thereto;

and due and appropriate notice of the Motion, the relief requested therein and the Interim

Hearing having been served by the Debtors on the Investment Agent (for itself and the

Participants); Standard Chartered Bank ("SCB"), as agent under the prepetition secured

Murabaha facilities dated May 30, 2011, and December 22, 2011 (the "SCB Facilities"); the

12-11076-shl Doc 690-1 Filed 12/04/12 Entered 12/04/12 18:01:19 Exhibit A-Interim Order Pg 6 of 30

Office of the United States Trustee for the Southern District of New York, 33 Whitehall Street, 21st Floor, New York, New York 10004 (Attn: Richard Morrissey, Esq.) (the "U.S. Trustee"); Milbank, Tweed, Hadley & McCloy LLP, 1 Chase Manhattan Plaza, New York, New York 10005 (Attn: Dennis F. Dunne, Esq., Abhilash M. Raval, Esq., and Evan R. Fleck, Esq.), counsel for the official committee of unsecured creditors in the Cases (the "Committee"); the Joint Provisional Liquidators (as defined in the DIP Agreement), Sidley Austin LLP, Woolgate Exchange, 25 Basinghall Street London, EC2V 5HA (Attn: Patrick Corr and Benjamin Klinger); and all parties listed on the Master Service List established in the Cases (collectively, the "Notice Parties"), and upon the record of the Interim Hearing held by this Court on December 4, 2012; and upon the arguments of counsel at the Interim Hearing; and the Court having considered any objections to the relief sought herein; and after due deliberation and consideration and sufficient cause appearing therefor,

IT IS FOUND, DETERMINED, ORDERED AND ADJUDGED, that:

- 1. *Motion*. The Motion is hereby granted on an interim basis, to the extent set forth herein.
- 2. *Jurisdiction*. This Court has core jurisdiction over the Cases, the Motion, and the parties and property affected hereby pursuant to 28 U.S.C. §§ 157(b) and 1334. Venue is proper before this Court pursuant to 28 U.S.C. §§ 1408 and 1409.
- 3. Notice. Notice of the Motion, the relief requested therein and the Interim Hearing was served by the Debtors on the Notice Parties. The interim relief granted herein is necessary to avoid immediate and irreparable harm to the Debtors and their estates pending the Final Hearing. Under the circumstances, the request for the interim relief granted herein and the notice given by the Debtors of the Motion and the Interim Hearing constitutes due and sufficient notice

12-11076-shl Doc 690-1 Filed 12/04/12 Entered 12/04/12 18:01:19 Exhibit A-Interim Order Pg 7 of 30

thereof and complies with Bankruptcy Rule 4001(b) and (c), and Local Rule 4001-2, and no further notice of the relief sought at the Interim Hearing is necessary or required.

- 4. Effectiveness. This Interim Order shall constitute findings of fact and conclusions of law. Notwithstanding Bankruptcy Rules 4001(a)(3), 6004(h), 6006(d), 7062, and 9024, or any other Bankruptcy Rule or Rule 62(a) of the Federal Rules of Civil Procedure, this Interim Order shall be immediately effective and enforceable upon its entry, and there shall be no stay of execution or effectiveness of this Interim Order.
 - 5. Findings Regarding the Commodities Transactions.
 - (a) The purchases and sales of the Commodities are essential to the DIP Agreement and the DIP Facility, and thus provide a basis for the Debtors to access liquidity essential to operate their businesses and preserve and enhance their enterprise value for the benefit of their stakeholders, and are necessary for the Debtors' overall restructuring.
 - (b) The purchases and sales of the Commodities shall be entitled to the protections of section 363(m) of the Bankruptcy Code.
 - 6. Findings Regarding the DIP Facility.
 - (a) Good cause has been shown for approval of the DIP Facility and entry of this Interim Order.
 - (b) The Debtors have an immediate need to obtain the credit available under the DIP Facility in order to permit, among other things, the orderly continuation of the operation of their businesses, including to fund general corporate and working capital requirements; to fund administrative costs of the Cases; and to pay such other amounts in accordance with the Finance Documents. The Debtors' access to sufficient working

12-11076-shl Doc 690-1 Filed 12/04/12 Entered 12/04/12 18:01:19 Exhibit A-Interim Order Pg 8 of 30

capital and liquidity by obtaining new credit under the DIP Facility is vital to the preservation and maintenance of the going concern values of the Debtors and necessary to avoid immediate and irreparable harm to the Debtors' estates.

- terms from sources other than the Investment Agent and the Participants under the Finance Documents and are unable to obtain adequate unsecured credit allowable under section 503(b)(1) of the Bankruptcy Code as an administrative expense. The Debtors are also unable to obtain secured credit allowable under sections 364(c)(1), 364(c)(2) and 364(c)(3) of the Bankruptcy Code on better terms without the Debtors granting to Fortress, as Security Agent under the DIP Facility (in such capacity, the "Security Agent") for the benefit of the Finance Parties (as defined in the DIP Agreement), the DIP Liens (as defined below) and the Superpriority Claims (as defined below) upon the terms and conditions set forth in this Interim Order and in the Finance Documents.
- (d) The terms of the DIP Facility are fair and reasonable, reflect the Debtors' exercise of prudent business judgment consistent with their fiduciary duties and constitute "reasonably equivalent value" and "fair consideration" within the meaning of such terms under section 548 of the Bankruptcy Code and under applicable non-bankruptcy law.
- (e) The DIP Facility has been negotiated in good faith and at arm's length among the Debtors and the Investment Agent, among others, and the terms of the DIP Facility are fair and reasonable under the circumstances and reflect the Debtors' exercise of prudent business judgment consistent with their fiduciary duties. Further, all of the Debtors' obligations arising under, in respect of or in connection with the DIP

12-11076-shl Doc 690-1 Filed 12/04/12 Entered 12/04/12 18:01:19 Exhibit A-Interim Order Pg 9 of 30

Agreement or any of the other Finance Documents (collectively, the "**DIP Obligations**"), shall be deemed to have been extended by the Investment Agent and the Participants and their affiliates in good faith, as that term is used in section 364(e) of the Bankruptcy Code, and in express reliance upon the protections offered by section 364(e) of the Bankruptcy Code, and the Investment Agent and the Participants shall be entitled to the full protection of section 364(e) of the Bankruptcy Code in the event that this Interim Order or any provision hereof is vacated, reversed or modified, on appeal or otherwise.

- Bankruptcy Rule 4001(b)(2) and (c)(2). The authorization granted herein on an interim basis to enter into the Finance Documents and to purchase and sell Commodities up to an aggregate purchase price of \$25 million is necessary to avoid immediate and irreparable harm to the Debtors and their estates. Consummation of the DIP Facility in accordance with this Interim Order and the Finance Documents is therefore in the best interest of the Debtors' estates as its consummation will, among other things, allow the Debtors to facilitate their Chapter 11 goals and maximize the value of their estates.
 - 7. Authorization of the Commodities Transactions.
- (a) Pursuant to section 363(b)(1) of the Bankruptcy Code, the Purchaser is hereby authorized to (i) enter into and perform its obligations under the Purchase Contracts (as defined in the DIP Agreement), including the obligation to purchase Commodities from the Investment Agent at the Cost Price plus Profit Amount (as such terms are defined in the DIP Agreement) and such other amounts due and payable under the Finance Documents, and (ii) sell such Commodities as set forth in the DIP Agreement.

12-11076-shl Doc 690-1 Filed 12/04/12 Entered 12/04/12 18:01:19 Exhibit A-Interim Order Pg 10 of 30

The transactions described in this subparagraph (a) shall be referred to collectively as the "Commodities Transactions."

- (b) To the extent provided in the DIP Agreement, the Purchaser shall indemnify the Investment Agent for any actions, claim, proceedings, liabilities, costs, and expenses associated with, or arising in connection with, the Commodities Transactions or the other transactions contemplated under the Purchase Contracts, other than any actions, claims, proceedings, liabilities, costs, and expenses arising from the ownership of the Commodities by any of the Indemnified Parties (as defined in the DIP Agreement).
 - 8. *Authorization of the Finance Documents.*
- enter into the Finance Documents and the Finance Documents are hereby approved. The Purchaser is hereby authorized to enter into the DIP Agreement, and the Debtor Guarantors are hereby authorized to unconditionally guaranty (on a joint and several basis and except that the guarantees of AEID II Holdings Limited ("AEID II"), RailInvest Holdings Limited ("RailInvest"), and WindTurbine Holdings Limited ("WTHL") shall be expressly subordinated to the Prior SCB Claims) the Purchaser's obligations under the Finance Documents up to an aggregate principal or face amount of \$25,000,000 (plus profits, fees, costs and other expenses and amounts provided for in the Finance Documents) on an interim basis, in accordance with the terms of this Interim Order and the Finance Documents, the proceeds of which shall be used for all purposes permitted under the Finance Documents, including, without limitation, to provide working capital for the Purchaser and the Debtor Guarantors; to fund general corporate purposes; and to pay profits, fees and expenses, in each case in accordance with this Interim Order, the Finance

12-11076-shl Doc 690-1 Filed 12/04/12 Entered 12/04/12 18:01:19 Exhibit A-Interim Order Pg 11 of 30

Documents and the DIP Budget (as defined in the DIP Agreement) (subject to the variances set forth in Clause [13] of the DIP Agreement).

- (b) In furtherance of the foregoing and without further approval of this Court, each Debtor is authorized and directed to perform all acts, to make, execute and deliver all instruments and documents (including, without limitation, the execution or recordation of security agreements, mortgages and financing statements), and, without further application to the Court, to pay all fees under the Finance Documents, that may be reasonably required or necessary for the Debtors' performance of their obligations under the DIP Facility, including, without limitation:
 - (i) the execution, delivery and performance of the Finance Documents;
 - amendments, waivers, consents, or other modifications to and under the Finance Documents including, among other things, to at any time, add additional institutions as Participants or reallocate the commitments under the Finance Documents among Participants, in each case in such form as Arcapita, the Purchaser, the Investment Agent, and the Participants (to the extent required under the DIP Agreement) may reasonably agree, it being understood that no further approval of the Court shall be required for amendments, waivers, consents, or other modifications to and under the Finance Documents or the DIP Obligations that do not (A) shorten the maturity of the extensions of credit thereunder, (B) increase the commitments or the rate of profit or fees payable thereunder, (C) materially impair SCB's rights under the SCB Order, or (D) are otherwise not materially burdensome

12-11076-shl Doc 690-1 Filed 12/04/12 Entered 12/04/12 18:01:19 Exhibit A-Interim Order Pg 12 of 30

to the Debtors' estates; provided that, notwithstanding the foregoing, the Debtors shall provide 3 days' notice to and consult with counsel to the Committee, SCB, and the Joint Provisional Liquidators prior to entering into any amendment or other modification to the Finance Documents or the DIP Obligations;

- (iii) the non-refundable payment to the Investment Agent, the
 Security Agent, the Arranger and the Participants, as the case may be, of the fees
 referred to in the Finance Documents and the reasonable fees, costs and expenses of
 professionals retained by the Investment Agent and the Security Agent, as and to
 the extent provided for in the Finance Documents, without the necessity of filing
 retention applications or fee applications; and
- (iv) the performance of all other acts required under or in connection with the Finance Documents.
- Occuments shall constitute valid and binding obligations of the Debtors (except for Falcon), enforceable against each such Debtor party thereto in accordance with their terms and this Interim Order. No obligation, payment, transfer or grant of security under the Finance Documents or this Interim Order shall be stayed, restrained, voidable, or recoverable under the Bankruptcy Code or under any applicable law (including without limitation, under section 502(d) of the Bankruptcy Code), or subject to any defense, reduction, setoff, recoupment or counterclaim.
- 9. *Indemnity*. The indemnity provisions of the Finance Documents are hereby approved to the extent provided therein.

12-11076-shl Doc 690-1 Filed 12/04/12 Entered 12/04/12 18:01:19 Exhibit A-Interim Order Pg 13 of 30

- 10. Superpriority Claims.
- Pursuant to section 364(c)(1) of the Bankruptcy Code, all of the DIP (a) Obligations, including without limitation any obligations arising from the Commodities Transactions, shall constitute allowed administrative expenses against each of the Debtors (excluding Falcon) with priority over any and all administrative expenses, and all other claims against such Debtors, now existing or hereafter arising, of any kind whatsoever, including, without limitation, all administrative expenses of the kind specified in sections 503(b) and 507(b) of the Bankruptcy Code, and over any and all administrative expenses or other claims arising under section 105, 326, 328, 330, 331, 365, 503(b), 506(c) (upon entry of the Final Order, to the extent therein approved), 507(a), 507(b), 546, 726, 1113 or 1114 of the Bankruptcy Code, including, without limitation, any superpriority claims granted as adequate protection in favor of secured parties in the Cases, whether or not such expenses or claims may become secured by a judgment lien or other non-consensual lien, levy or attachment, which allowed claims shall be payable from and have recourse to all pre- and post-petition property of the Debtors (excluding Falcon) and all proceeds thereof (the "Superpriority Claims"), subject only to the payment of the Carve Out to the extent specifically provided for herein, and excluding Avoidance Actions and the proceeds thereof. Notwithstanding anything to the contrary contained in the DIP Agreement, Finance Documents, this Interim Order, or the Motion, so long as the obligations under the SCB Facilities remain outstanding, (i) the DIP Obligations and the Superpriority Claims shall be junior to the SCB Superpriority Claims (as defined in the Settlement Term Sheet

12-11076-shl Doc 690-1 Filed 12/04/12 Entered 12/04/12 18:01:19 Exhibit A-Interim Order Pg 14 of 30

attached as Exhibit 1 to the SCB Order¹) to the extent that the SCB Superpriority Claims relate to funds transferred by, or other disposition of, AEID II, RailInvest, or WTHL; (ii) the DIP Obligations and the Superpriority Claims against AEID II, RailInvest, and WTHL shall be subordinated to the existing guarantees in favor of SCB against AEID II, RailInvest; and WTHL; and (iii) SCB shall have a prior superpriority claim in all proceeds of the EuroLog IPO (as defined in the SCB Order) to the extent provided under the SCB Order (the claims in favor of SCB as described in (i) through (iii), and the Listco Pledge (as defined in the SCB Order) in connection with the EuroLog IPO, in each case to the extent allowed, referred to collectively as, the "**Prior SCB Claims**").

(b) For purposes hereof, the "Carve Out" shall mean (i) any unpaid fees required to be paid to the Clerk of the United States Bankruptcy Court for the Southern District of New York and to the U.S. Trustee under section 1930(a) of title 28 of the United States Code and interest thereon, (ii) the reasonable fees and expenses approved by the Bankruptcy Court incurred by a trustee under sections 726(b) or 1104 of the Bankruptcy Code in an aggregate amount not to exceed \$25,000, (iii) the reasonable and documented expenses of members of the Committee (excluding fees and expenses of professional persons employed by the Committee and/or such Committee member individually) in an amount not to exceed \$200,000; (iv) to the extent allowed at any time, all unpaid fees and expenses allowed by the Bankruptcy Court of professionals or professional firms retained pursuant to sections 327, 328, 330, 363, or 1103 of the Bankruptcy Code by the Debtors or the Committee (the "**Professional Persons**") and the

The "SCB Order" means the Order Pursuant to Section 105(a) of the Bankruptcy Code and Bankruptcy Rule 9019, Authorizing and Approving the Settlement with Standard Chartered Bank (Docket No. 587), entered October 19, 2012.

12-11076-shl Doc 690-1 Filed 12/04/12 Entered 12/04/12 18:01:19 Exhibit A-Interim Order Pg 15 of 30

reasonable fees and expenses of the Joint Provisional Liquidators, in each case, that were accrued or incurred, as applicable through the date upon which the Purchaser and the Committee receives from the Investment Agent a written notice of the occurrence of an Event of Default (as defined in the DIP Agreement) and the Investment Agent's intention to involve the Carve Out (the "Carve Out Notice"); and (v) to the extent allowed at any time, all fees and expenses of Professional Persons and the Joint Provisional Liquidators incurred after the date upon which the Purchaser receives the Carve Out Notice, in the aggregate amount not to exceed \$15,000,000 (the "Carve Out Cap"), provided that (a) the Carve Out Cap shall not be reduced nor increased by the amount of any compensation or reimbursement of expenses incurred by or paid to any Professional Person or the Joint Provisional Liquidators prior to the Purchaser's receipt of the Carve Out Notice or by any fees, expenses, indemnities, or other amounts paid to the Investment Agent, Participants, or their respective attorneys or agents under the DIP Facility or otherwise and (b) to the extent that the Carve Out Cap is reduced by an amount as a result of payment of fees and expenses during the continuation of an Event of Default and after delivery of the Carve Out Notice, and such Event of Default is subsequently cured or waived and the Carve Out Notice is rescinded in writing, such dollar limitation shall be increased by an amount equal to the amount by which it has been so reduced.

(c) Notwithstanding anything herein, without the prior written consent of the Investment Agent or the Security Agent, as applicable, the Carve Out shall not include, apply to, or be available for any fees or expenses incurred by any party in connection with (1) any challenge to the validity, perfection, priority, extent or enforceability of the DIP Obligations or other transactions under the DIP Facility, or the

12-11076-shl Doc 690-1 Filed 12/04/12 Entered 12/04/12 18:01:19 Exhibit A-Interim Order Pg 16 of 30

DIP Liens on any Collateral or security interests securing the DIP Obligations; (2) any investigation or assertion of any other claims, adversary proceedings, causes of action, or other litigation, including any action or obligation with respect to the Superpriority Claims or DIP Liens, against any Participant, the Investment Agent or any other holder of any DIP Obligations in such capacity; (3) to object to, contest, delay, prevent or interfere in any way with the exercise of rights or remedies by the Security Agent under the Finance Documents (except that the Debtors may dispute whether an Event of Default has occurred under paragraph 14(b) hereof and the Debtors shall be entitled to any notice provisions provided in this Interim Order); or (4) during the continuation of an Event of Default and after delivery of the Carve Out Notice, any separate or additional act or series of acts which would constitute an Event of Default, provided that deviations from the DIP Budget for purposes of making payments to Professional Persons that would constitute an Event of Default but that are otherwise permitted under the Carve Out shall not be subject to this clause (4). Nothing herein shall be construed to impair the ability of any party to object to the fees, expenses, reimbursement or compensation of any Professional Person. The Carve Out must be used in full and exhausted prior to the Debtors' use of the \$1,000,000 carve out for professional fees provided for in the SCB Order.

- 11. Adequate Protection. Nothing contained in this Interim Order modifies, alters, amends or supersedes the grant of adequate protection to SCB and the priority of SCB's claims against the Debtors pursuant to the SCB Order.
- 12. *DIP Liens*. As security for the DIP Obligations, effective and perfected upon the date of this Interim Order and without the necessity of the execution, recordation or filings of mortgages, security agreements, control agreements, pledge agreements, financing statements or

12-11076-shl Doc 690-1 Filed 12/04/12 Entered 12/04/12 18:01:19 Exhibit A-Interim Order Pg 17 of 30

other similar documents, or the possession or control by the Security Agent of, or over, any Collateral, the following security interests and liens are hereby granted to the Security Agent for the benefit of the Investment Agent and the Finance Parties (all tangible and intangible property, whether real or personal, identified in clauses (a) and (b) below being collectively referred to as the "Collateral"); all such liens and security interests granted to the Security Agent for the benefit of the Finance Parties pursuant to this Interim Order, the "DIP Liens", subject and subordinate only to the payment of the Carve Out and the Prior SCB Claims and, in the case of the collateral identified in clause (b) below, any claim secured by a senior lien therein; provided, however, that the Collateral shall not include (i) the Debtors' claims and causes of action under sections 544, 545, 547, 548 and 550 (but not to the extent of any recoveries under section 549 of the Bankruptcy Code) of the Bankruptcy Code (the "Avoidance Actions") or the proceeds thereof, (ii) Arcapita Investment Funding Limited's interests in PointPark Properties s.r.o., or (iii) the assets of AEID II, RailInvest, or WTHL (except to the extent such assets constitute collateral under the SCB Facilities, in which case the Security Agent's liens shall be junior to liens granted to SCB); provided further, however, that the Collateral shall not include any property to the extent that the Investment Agent reasonably determines, and notifies the Purchaser in writing, that the costs of obtaining liens or security interests with respect to such property are excessive in relation to the value of the security interest afforded thereby:

(a) First Lien Unencumbered Property. Pursuant to section 364(c)(2) of the Bankruptcy Code, a valid, binding, continuing, enforceable, fully-perfected first priority senior security interest in and lien upon all property of Arcapita, the Purchaser, and ALTHL, whether existing on the Petition Date or thereafter arising, coming into existence or acquired, whether tangible or intangible, whether real or personal, that (1) is not subject

12-11076-shl Doc 690-1 Filed 12/04/12 Entered 12/04/12 18:01:19 Exhibit A-Interim Order Pg 18 of 30

to valid, perfected and non-avoidable liens or security interests as of the Petition Date or (2) becomes unencumbered and is no longer subject to any lien or security interest, including, without limitation, the Purchaser's interests in the WCFs (as defined in the DIP Agreement) and the Purchaser's voting rights with respect thereto, ALTHL's interests in LT CayCos (as defined in the DIP Agreement) that are unencumbered or become unencumbered, the Purchaser's non-syndicated interests in the Syndication Companies (as defined in the DIP Agreement), cash, general intangibles, accounts, equipment, goods, inventory, fixtures, documents, instruments, chattel paper, letters of credit and letters of credit rights, investment property, commercial tort claims, money, deposit accounts, supporting obligations (each of foregoing terms as defined in the Uniform Commercial Code as in effect from time to time in the State of New York (the "UCC")), all books and records relating to the foregoing, and all other personal and real property, whether tangible or intangible, and all proceeds (as defined in the UCC) and products of each of the foregoing (the "First Lien Collateral").

(b) Liens Junior to Certain Other Liens. Pursuant to sections 364(c)(3) of the Bankruptcy Code, a valid, binding, continuing, enforceable, fully-perfected security interest in and junior lien upon all property of the Debtors (except for Falcon) that are subject to (i) any valid, perfected and non-avoidable lien in existence on the Petition Date or (ii) any valid lien in existence on the Petition Date that is perfected subsequent to the Petition Date pursuant to section 546(b) of the Bankruptcy Code or otherwise comes into existence or is acquired after the Petition Date, whether tangible or intangible, whether real or personal, together with all proceeds and products thereof, including, in each case and for

12-11076-shl Doc 690-1 Filed 12/04/12 Entered 12/04/12 18:01:19 Exhibit A-Interim Order Pg 19 of 30

so long as the obligations under the SCB Facilities remain unpaid, the liens on the collateral securing the obligations under the SCB Facilities.

- Collateral shall be free and clear of all senior liens, claims and encumbrances, other than the DIP Liens granted to the Security Agent for the benefit of the Finance Parties and except for those liens, claims, and encumbrances expressly permitted under the Finance Documents or this Interim Order. Any liens and claims granted as adequate protection to any secured party (other than those granted to SCB in connection with the SCB Facilities and the SCB Order) are junior and subordinate to the DIP Liens in the Collateral granted to the Security Agent, for the benefit of the Finance Parties pursuant to this Interim Order.
- which are now or shall hereafter come into the possession or control of Purchaser, Arcapita or ALTHL, or to which any such Debtor is now or shall become entitled under the Finance Documents, shall be promptly deposited into deposit accounts maintained by the Purchaser or Arcapita upon which the Security Agent shall have first priority liens pursuant to this Interim Order, and such collections and proceeds shall remain subject to the DIP Liens and shall be treated in accordance with this Interim Order and the Finance Documents. Subject to the provisions of this Interim Order, upon the occurrence and continuation of an Event of Default under the Finance Documents, all financial institutions in which any deposit accounts, lockboxes, blocked accounts, or other accounts of any of the Debtors holding the proceeds of any of the First Lien Collateral are located are hereby authorized and directed to comply with any request of the Security Agent to turn over to the Security Agent all funds therein without setoff, recoupment, or deduction of any kind.

12-11076-shl Doc 690-1 Filed 12/04/12 Entered 12/04/12 18:01:19 Exhibit A-Interim Order Pg 20 of 30

- 14. *Protection of Financing Parties' Rights*. The automatic stay provisions of section 362 of the Bankruptcy Code are hereby vacated and modified to the extent necessary, without the need for any further order of the Court:
 - To permit the Investment Agent and/or Security Agent to exercise, (a) upon the occurrence and continuation of any Event of Default under the Finance Documents or the Maturity Date (as defined in the DIP Agreement), all rights and remedies under the Finance Documents, and, to the extent provided for in the Finance Documents, to take any or all of the following actions without further order of or application to this Court: (i) cease to make any extensions of credit or advances to the Debtors and declare the Participants' commitments under the DIP Facility terminated; (ii) declare all DIP Obligations to be immediately due and payable without presentment, demand, protest or notice; (iii) set off and apply immediately any and all amounts in accounts maintained by Arcapita, the Purchaser, and ALTHL with the Investment Agent, Security Agent or any Participant against the DIP Obligations to the extent permitted under the Finance Documents or applicable law; (iv) exercise all rights and remedies against the Collateral to the extent provided for in any Finance Document; and (v) take any other actions or exercise any other rights or remedies permitted under this Interim Order, the Finance Documents, or applicable law to realize upon the Collateral and/or effect the repayment and satisfaction of the DIP Obligations, subject to SCB's rights under paragraph 14(b) and the Investment Agent and/or the Security Agent providing seven (7) days written notice (by facsimile, telecopy, electronic mail or otherwise) to the Debtors, counsel to the Debtors, the U.S. Trustee, the Joint Provisional Liquidators, counsel to the Committee, and

12-11076-shl Doc 690-1 Filed 12/04/12 Entered 12/04/12 18:01:19 Exhibit A-Interim Order Pg 21 of 30

counsel to SCB, prior to exercising any enforcement rights or remedies under (iii) through (v) above (but not any of the rights described in clauses (i) and (ii) above).

- In any hearing regarding any exercise of rights or remedies by the Investment Agent and/or the Security Agent, the only issue that may be raised by the Debtors or any party in interest shall be whether, in fact, an Event of Default under the Finance Documents has occurred and is continuing, and neither the Debtors nor any party in interest shall be entitled to seek relief, including, without limitation, under section 105 of the Bankruptcy Code, to the extent such relief would in any way impair or restrict the rights and remedies of the Investment Agent or the Security Agent set forth in this Interim Order or the Finance Documents. Notwithstanding the foregoing, so long as the obligations under the SCB Facilities remain outstanding, SCB shall be permitted to assert that any exercise of rights or remedies by the Investment Agent and/or the Security Agent against (i) collateral securing the obligations under the SCB Facilities and/or (ii) AEID II, RailInvest, or WTHL is not permitted under the Interim Order or the SCB Order. In no event shall the Investment Agent, the Security Agent or the Participants be subject to the equitable doctrine of "marshaling" or any similar doctrine with respect to the Collateral.
- (c) Until the payment in full of the DIP Obligations, any party other than the Security Agent that has or obtains a lien or security interest in the Collateral shall not exercise any rights or remedies with respect to the First Lien Collateral.
- 15. Limitation on Charging Expenses Against Collateral. Subject only to, and effective only upon entry of, the Final Order and only to the extent provided for therein, except to the extent of the Carve Out, no expenses of administration of the Cases or any future proceeding that may result therefrom, including a chapter 7 liquidation in bankruptcy and the

12-11076-shl Doc 690-1 Filed 12/04/12 Entered 12/04/12 18:01:19 Exhibit A-Interim Order Pg 22 of 30

cost of preservation or disposition of the Collateral, shall be charged against or recovered from the Collateral pursuant to section 506(c) of the Bankruptcy Code or any similar principle of law, without the prior written consent of the Investment Agent, and no such consent shall be implied from any other action, inaction, or acquiescence by the Investment Agent, the Security Agent or the Participants. Subject to entry of a Final Order which provides for a waiver of such claims, the Debtors (for themselves and their estates) hereby irrevocably waive and relinquish any rights they may have under section 506(c) of the Bankruptcy Code with respect to the Collateral.

16. Perfection of DIP Liens.

Finance Parties, are each hereby authorized, but not required, to file or record financing statements, trademark filings, copyright filings, mortgages, notices of lien or similar instruments in any jurisdiction, or take possession of or control over, or take any other action in order to validate and perfect the liens and security interests granted hereunder. Whether or not the Investment Agent or the Security Agent, on behalf of the Finance Parties shall, in its sole discretion, choose to file such financing statements, trademark filings, copyright filings, mortgages, notices of lien or similar instruments, or take possession of or control over, or otherwise confirm perfection of the liens and security interests granted hereunder, such liens and security interests shall be deemed valid, perfected, allowed, enforceable, non-avoidable and not subject to challenge, dispute or subordination, at the time and on the date of entry of this Interim Order. Without limitation of the foregoing, the Security Agent on behalf of the Finance Parties shall have a perfected lien upon and security interest of the same relative priority or priorities set forth

12-11076-shl Doc 690-1 Filed 12/04/12 Entered 12/04/12 18:01:19 Exhibit A-Interim Order Pg 23 of 30

in paragraphs 12(a) and 12(b) in all bank accounts in which any cash constituting the Collateral is deposited, without any need for entering into any control agreement.

- (b) A certified copy of this Interim Order may, in the discretion of the Investment Agent or the Security Agent, be filed with or recorded in filing or recording offices in addition to or in lieu of such financing statements, mortgages, notices of lien or similar instruments, and all filing offices are hereby authorized to accept such certified copy of this Interim Order for filing and recording. For the avoidance of doubt, the automatic stay of section 362(a) of the Bankruptcy Code shall be modified to the extent necessary to permit the Investment Agent, the Security Agent, and the Finance Parties to take all actions, as applicable, referenced in this paragraph 16.
 - 17. Preservation of Rights Granted Under this Interim Order.
- claim or lien having a priority superior to or *pari passu* with those granted by this Interim Order to the Security Agent (for the benefit of the Participants) and/or the Investment Agent shall be granted or allowed while any portion of the DIP Facility (or any refinancing thereof) or the commitments thereunder or the DIP Obligations remain outstanding, and the DIP Liens shall not be (i) subject or junior to any lien or security interest that is avoided and preserved for the benefit of the Debtors' estates under section 551 of the Bankruptcy Code or (ii) subordinated to or made *pari passu* with any other lien or security interest, whether under section 364(d) of the Bankruptcy Code or otherwise other than the Carve Out and the SCB Prior Claims as set forth herein, and as expressly provided in this Interim Order.

12-11076-shl Doc 690-1 Filed 12/04/12 Entered 12/04/12 18:01:19 Exhibit A-Interim Order Pg 24 of 30

- (b) Unless all DIP Obligations shall have been indefeasibly paid in full. the Debtors shall not seek (i) any modification or extension of this Interim Order without the prior written consent of the Investment Agent, and no such consent shall be implied by any other action, inaction or acquiescence by the Investment Agent, or (ii) an order converting or dismissing any of the Cases of the Debtors (excluding Falcon). If an order dismissing any of the Cases under section 305 or 1112 of the Bankruptcy Code or otherwise is at any time entered, such order shall provide (in accordance with sections 105 and 349 of the Bankruptcy Code) that (i) the Superpriority Claims and DIP Liens granted to the Investment Agent and the Security Agent (for the benefit of the Participants) pursuant to this Interim Order shall continue in full force and effect, shall maintain their priorities as provided in this Interim Order and shall, notwithstanding such dismissal, remain binding on all parties in interest until all DIP Obligations shall have been indefeasibly paid in full in cash and the commitments under the DIP Facility have been terminated in accordance with the Finance Documents and (ii) this Court shall retain jurisdiction, notwithstanding such dismissal, for the purposes of enforcing the Superpriority Claims and DIP Liens.
- (c) If any or all of the provisions of this Interim Order are hereafter reversed, modified, vacated or stayed, such reversal, stay, modification or vacation shall not affect (i) the validity of any DIP Obligations incurred prior to the actual receipt by the Investment Agent of written notice of the effective date of such reversal, stay, modification or vacation or (ii) the validity or enforceability of the DIP Liens and Superpriority Claims authorized or created hereby or pursuant to the Finance Documents with respect to any DIP Obligations. Notwithstanding any such reversal, stay, modification or vacation, the DIP

12-11076-shl Doc 690-1 Filed 12/04/12 Entered 12/04/12 18:01:19 Exhibit A-Interim Order Pg 25 of 30

Obligations incurred by the Debtors pursuant to the Finance Documents, prior to the actual receipt by the Investment Agent of written notice of the effective date of such reversal, stay, modification or vacation shall be governed in all respects by the original provisions of this Interim Order, and the Investment Agent, the Participants, and the other Finance Parties, shall be entitled to all the rights, remedies, privileges and benefits granted in section 364(e) of the Bankruptcy Code, this Interim Order and pursuant to the Finance Documents.

- (d) If any or all of the provisions of this Interim Order are hereafter reversed, modified, vacated or stayed, such reversal, stay, modification or vacation shall not affect the validity of any purchases or sales of Commodities pursuant to the Purchase Contracts or other Finance Documents, if such purchases and obligations were effected or incurred, as applicable, prior to the actual receipt by the Investment Agent of written notice of the effective date of such reversal, stay, modification or vacation. Notwithstanding any such reversal, stay, modification or vacation, purchases or sales of Commodities pursuant to the Purchase Contracts or other Finance Documents, if such purchases and obligations were effected or incurred, as applicable, prior to the actual receipt by the Investment Agent of written notice of the effective date of such reversal, stay, modification or vacation, shall be governed in all respects by the original provisions of this Interim Order, and the purchases and sales of Commodities shall be entitled to all the privileges, benefits, and protections granted in section 363(m) of the Bankruptcy Code, this Interim Order and pursuant to the applicable Finance Documents.
- (e) Except as expressly provided in this Interim Order or in the Finance Documents, no order providing for the sale of any First Lien Collateral under section 363 or any other provision of the Bankruptcy Code shall be entered by this Court unless in

12-11076-shl Doc 690-1 Filed 12/04/12 Entered 12/04/12 18:01:19 Exhibit A-Interim Order Pg 26 of 30

connection with and concurrently with the consummation of such sale, the proceeds of such sale are distributed in accordance with the terms of the Finance Documents.

- (f) Except as expressly provided in this Interim Order or in the Finance Documents, the DIP Liens, the Superpriority Claims and all other rights and remedies of the Investment Agent granted by the provisions of this Interim Order and the Finance Documents shall survive, and shall not be modified, impaired or discharged by (i) the entry of an order converting any of the Cases of the Debtors to a case under chapter 7 of the Bankruptcy Code, dismissing any of the Cases of the Debtors, terminating the joint administration of these Cases of the Debtors or by any other act or omission, (ii) the entry of an order approving the sale of any Collateral pursuant to section 363(b) of the Bankruptcy Code or (iii) the entry of an order confirming a plan of reorganization in any of the Cases and, pursuant to section 1141(d)(4) of the Bankruptcy Code, the Debtors have waived any discharge as to any remaining DIP Obligations. The terms and provisions of this Interim Order and the Finance Documents shall continue in these Cases, in any successor cases if these Cases cease to be jointly administered, or in any superseding chapter 7 cases under the Bankruptcy Code, and the DIP Liens, the Superpriority Claims and all other rights and remedies of the Investment Agent granted by the provisions of this Interim Order and the Finance Documents shall continue in full force and effect until the DIP Obligations are indefeasibly paid in full.
- 18. Limitation on Use of Financing Proceeds and Collateral. Notwithstanding anything herein or in any other order by this Court to the contrary, no portion of the proceeds under the DIP Facility, shall be used (i) in connection with the investigation, initiation or prosecution of any claims against the Investment Agent or any Participant or other holder of DIP

12-11076-shl Doc 690-1 Filed 12/04/12 Entered 12/04/12 18:01:19 Exhibit A-Interim Order Pg 27 of 30

Obligations under the Finance Documents, provided that, with respect to Participants and holders other than Fortress and its subsidiaries and affiliates, the foregoing shall apply solely to claims against such parties in their capacities as Participants and holders under the DIP Facility; (ii) to make any payment on account of any claims or indebtedness arising or incurred prior to the Petition Date except as permitted under the Finance Documents, and then only in accordance with the DIP Budget (subject to applicable variances under the DIP Agreement); (iii) for any act which has the effect of materially or adversely modifying or compromising the rights and remedies of the Investment Agent, the Security Agent, or any Participant as set forth herein and in the other Finance Documents, or which results in the occurrence of an Event of Default (except as permitted under the Carve Out); (iv) directly or indirectly, for any payments to any governmental official or employee, political party, official of a political party, candidate for political office, or anyone else acting in an official capacity, in order to obtain, retain or direct business or obtain any improper advantage, in violation of the United States Foreign Corrupt Practices Act of 1977, as amended; (v) be paid to (A) any Embargoed Person (as defined in the DIP Agreement), (B) any agency of the government of any Sanctioned Country (as defined in the DIP Agreement), (C) any organization controlled by a Sanctioned Country or (D) any person resident in a Sanctioned Country, to the extent subject to a sanctions program administered by U.S. Department of the Treasury's Office of Foreign Assets Control; or (vi) in any manner that violates Regulations T, U, or X of the Board of Governors of the Federal Reserve System of the United States or any other regulation thereof or to violate the Securities Exchange Act of 1934. Except as provided in the Finance Documents or in this Interim Order, no portion of the proceeds of any Collateral, including cash collateral, shall be used for any purpose other than as provided

12-11076-shl Doc 690-1 Filed 12/04/12 Entered 12/04/12 18:01:19 Exhibit A-Interim Order Pg 28 of 30

for in the DIP Budget (subject to the variances set forth in Clause 13 of the DIP Agreement and except as permitted under the Carve Out).

- 19. *Order Governs*. In the event of any inconsistency between the provisions of this Interim Order and the Finance Documents, the provisions of this Interim Order shall govern.
- 20. Binding Effect; Successors and Assigns. Except as expressly provided herein, the Finance Documents and the provisions of this Interim Order, including all findings herein, shall be binding upon all parties in interest in the Cases, including, without limitation, the Investment Agent, the Security Agent, the Participants, the Committee, and the Debtors and their respective successors and assigns (including any chapter 7 or chapter 11 trustee hereinafter appointed or elected for the estate of any of the Debtors) and shall inure to the benefit of the Financing Parties and the Debtors and their respective successors and assigns; provided, however, that the Investment Agent and the Participants shall have no obligation to extend any financing to any chapter 7 trustee or similar responsible person appointed for the estates of the Debtors. In determining to enter into any Commodities transaction, make any loan under the DIP Agreement or in exercising any rights or remedies as and when permitted pursuant to this Interim Order or the Finance Documents, the Investment Agent and the Participants shall not be deemed to be in control of the operations of the Debtors or to be acting as a "responsible person" or "owner or operator" with respect to the operation or management of the Debtors, so long as the Participants' actions do not constitute, within the meaning of 42 U.S.C. § 9601(20)(F), actual participation in the management or operational affairs of a vessel or facility owned or operated by a Debtor, or cause the status of responsible person or managing agent to exist under applicable law (as such terms, or any similar terms, are used in the United States Comprehensive Environmental Response, Compensation and Liability Act, 42 U.S.C. §§ 9601 et seq., as

12-11076-shl Doc 690-1 Filed 12/04/12 Entered 12/04/12 18:01:19 Exhibit A-Interim Order Pg 29 of 30

amended, or any similar federal or state statute). Nothing in this Interim Order or the Finance Documents shall permit the Debtors to violate 28 U.S.C. § 959(b). Further, nothing in this Interim Order providing for the release of non-Debtors or injunction of actions against non-Debtors shall apply to (a) any claims, debts, obligations, rights, suits, damages, actions, causes of action, remedies and liabilities of the United States and any agency thereof, (b) any claims, debts, obligations, rights, suits, damages, actions, causes of action, remedies, and liabilities of any State or any agency of any State, under state or federal environmental laws, or (c) any criminal liability under the laws of the United States.

Final Hearing. The Final Hearing is scheduled for [December 18, 2012] at 11:00 21. a.m. (prevailing Eastern time) before this Court. The Debtors shall promptly mail copies of this Interim Order (which shall constitute adequate notice of the Final Hearing, including without limitation, notice that the Debtors will seek approval at the Final Hearing of a waiver of rights under section 506(c) of the Bankruptcy Code) to the parties having been given notice of the Interim Hearing, and to any other party that has filed a request for notices with this Court and to counsel to the Committee. Any party in interest objecting to the relief sought at the Final Hearing shall serve and file written objections; which objections shall be served upon (a) counsel for the Debtors, Gibson, Dunn & Crutcher LLP, 200 Park Avenue, New York, New York 10166 (Attn: Michael A. Rosenthal, Esq., Matthew J. Williams, Esq., Matthew K. Kelsey, Esq., and Joshua P. Weisser, Esq.); (b) counsel for Fortress Credit Corp., as Investment Agent, Skadden, Arps, Slate, Meagher & Flom LLP, 4 Times Square, New York, New York 10036 (Attn: Kenneth S. Ziman); (c) the Office of the United States Trustee for the Southern District of New York, 33 Whitehall Street, 21st Floor, New York, New York 10004 (Attn: Richard Morrissey, Esq.); (d) the Official Committee of Unsecured Creditors, Milbank, Tweed, Hadley & McCloy

12-11076-shl Doc 690-1 Filed 12/04/12 Entered 12/04/12 18:01:19 Exhibit A-Interim Order Pg 30 of 30

LLP, 1 Chase Manhattan Plaza, New York, New York 10005 (Attn: Dennis F. I	Dunne, Esq.,		
Abhilash M. Raval, Esq., and Evan R. Fleck, Esq.); (e) counsel to SCB, Dechert LLP, 1095			
Avenue of the Americas, New York, New York 10036 (Attn: Brian E. Greer, Esq. and Nicole			
Herther-Spiro, Esq.); and (f) the Joint Provisional Liquidators, Sidley Austin LLP, Woolgate			
Exchange, 25 Basinghall Street London, EC2V 5HA (Attn: Patrick Corr and Benjamin Klinger),			
in each case to allow actual receipt by the foregoing no later than [_], 2012 at 4:00		
p.m. (prevailing Eastern time), with any replies filed by [], 2012 at 1:00 p.m.			
(prevailing Eastern time).			
Dated:, 2012 New York, New York			

THE HONORABLE SEAN H. LANE UNITED STATES BANKRUPTCY JUDGE

12-11076-shl Doc 690-2 Filed 12/04/12 Entered 12/04/12 18:01:19 Exhibit B - DIP Agreement Pg 1 of 97

Exhibit B

DIP Agreement

DATED DECEMBER , 2012

ARCAPITA INVESTMENT HOLDINGS LIMITED,

a Debtor and a Debtor-in-Possession under Chapter 11 of the Bankruptcy Code

- and -

FORTRESS CREDIT CORP.,

as Investment Agent, as agent for the Participants.

SUPERPRIORITY DEBTOR-IN-POSSESSION

MASTER MURABAHA AGREEMENT

CONTENTS

CLAU	JSE	PAGE
1.	DEFINITIONS AND INTERPRETATION	1
2.	MURABAHA FACILITY	
3.	CONDITIONS PRECEDENT	26
4.	PURCHASE CONTRACT LIMITATIONS	
5.	PROCEDURES	
6.	DEFERRED SALE PRICE AND PAYMENTS	
7.	PRIORITY AND LIENS; ETC	32
8.	TAX	34
9.	PREPAYMENTS; FACILITY LIMIT REDUCTION; INCREASE IN	
	LITY LIMIT	
10.	MITIGATION	
11.	REPRESENTATIONS AND WARRANTIES	
12.	UNDERTAKINGS	
13.	BUDGET COVENANT	
14.	DEFAULT AND REMEDIES	62
15.	FEES AND EXPENSES	69
16.	INDEMNITIES	70
17.	WAIVERS, REMEDIES CUMULATIVE	73
18.	MISCELLANEOUS	73
19.	NOTICES	76
20.	ASSIGNMENTS AND TRANSFERS	78
21.	GOVERNING LAW & DISPUTE RESOLUTION	79
SCHE	EDULE 1 CONDITION PRECEDENT DOCUMENTS	82
SCHE	EDULE 2 FORM OF TRANSACTION REQUEST	85
SCHE	EDULE 3 FORM OF OFFER LETTER AND ACCEPTANCE	87
SCHE	EDULE 4 DISCLOSURE SCHEDULE	88
SCHE	EDULE 5 LIST OF ENTITIES.	89
SCHE	EDULE 6 FORM OF ACCESSION LETTER	90
	EDULE 7 CONDITIONS PRECEDENT REQUIRED TO BE DELIVERED BY	
ADDI	TIONAL GUARANTOR	92
	CUTION PAGE OF SUPERPRIORITY DEBTOR-IN-POSSESSION MASTER ABAHA AGREEMENT	
111010		/ T

THIS SUPERPRIORITY DEBTOR-IN-POSSESSION MASTER MURABAHA AGREEMENT (as amended, modified or supplemented from time to time, this "Agreement") IS DATED DECEMBER , 2012.

BETWEEN:

- (1) **Arcapita Investment Holdings Limited,** an exempted company incorporated with limited liability under the laws of the Cayman Islands with company number 78594, a debtor and debtor-in-possession in the Cases under Chapter 11 the Bankruptcy Code ("**AIHL**"); and
- (2) Fortress Credit Corp. in its capacity as investment agent for the Participants (the "Investment Agent").

RECITALS

- A On the Petition Date, AIHL and certain of its Affiliates filed voluntary petitions with the Bankruptcy Court initiating their respective cases under Chapter 11 of the Bankruptcy Code and have continued in possession of their assets and in the management of their businesses pursuant to Sections 1107 and 1108 of the Bankruptcy Code.
- By an investment agency agreement (the "Investment Agency Agreement") dated on or about the date of this Agreement and made between the Investment Agent, AIHL, the Guarantors, the Arranger, the Security Agent and the Participants, the Participants have, among other things, appointed the Investment Agent as their agent to enter into the murabaha transactions contemplated by the Facility.
- C AIHL has requested that the Investment Agent, as agent for the Participants pursuant to the Investment Agency Agreement, provide the Facility, and the Investment Agent is willing to do so on the terms and subject to the conditions set forth herein and in the other Finance Documents.
- D To provide guarantees and security for the payment of the Obligations of the Obligors hereunder and under the other Finance Documents, the Obligors will provide and grant to the Investment Agent, for its benefit and the benefit of the Participants, certain Liens and superpriority administrative expense claims pursuant to Chapter 11 of the Bankruptcy Code, as more fully described herein.

IT IS AGREED:

1. DEFINITIONS AND INTERPRETATION

1.1 Definitions

In this Agreement (including its Preamble and the Recitals):

- "Accession Letter" means a document substantially in the form set out in Schedule 6.
- "Additional Guarantor" has the meaning given to it in clause 12.25(c).
- "Additional Guarantee" means any guarantee by an Additional Guarantor in favor of the Security Agent in accordance with clause 12.25(c).
- "AEID II" means AEID II Holdings Limited an exempted company incorporated with limited liability under the laws of the Cayman Islands with company number 203354, and a debtor and a debtor-in-possession in the Cases.
- "Affiliates" means, in relation to any Person, a Subsidiary of that Person or a Holding Company of that Person or any other Subsidiary of that Holding Company or any other Person that, directly or indirectly, is in control of, is controlled by, or is under common control with, such Person. For purposes of this definition, "control" of a Person means the power, directly or indirectly, either to (a) vote 10% or more of the securities having ordinary voting power for the election of directors (or persons performing similar functions) of such Person or (b) direct or cause the direction of the management and policies of such Person, whether by contract or otherwise.
- "Aggregate Cost Price" has the meaning given in clause 3.2(d).
- "AIHL" has the meaning given in the Recitals.
- "AIHL Sub" means Arcapita LT Holdings Limited, an exempted company incorporated with limited liability in the Cayman Islands with company number 239999, and a debtor and a debtor-in-possession in the Cases.
- "AIML" means Arcapita Investment Management Limited, an exempted company incorporated with limited liability under the laws of the Cayman Islands with company number CR-77989.
- "Anti-Terrorism Laws" has the meaning given in clause 11.22.
- "Arcapita Bank" means Arcapita Bank, B.S.C.(c) a closed joint stock company with registered address, Arcapita Building, Road 4612, Area 346, Bahrain Bay, Manama, Kingdom of Bahrain, and a debtor and a debtor-in-possession in the Cases.
- "Arranger" means Fortress Credit Corp.
- "Attributable Amount" means, with respect to the Net Cash Proceeds of any Prepayment Event, (a) if such Prepayment Event occurs at any Obligor or any wholly-owned Subsidiary of any Obligor, 100% of such Net Cash Proceeds and (b) if such Prepayment Event occurs at any other Person, the proportion of such Net Cash Proceeds equal to the proportion that the Obligors (or any of them) would receive in connection with any liquidating distribution of such Person and its direct and indirect parent entities (taking into account any preferred Equity Interests of such Person and its direct and indirect parent entities owned directly or indirectly by the Obligors or any of them and

any third party, respectively, and without giving effect to the payment of indebtedness or any other claims).

- "Authorization" means an authorization, consent, approval, resolution, license, exemption, filing, notarization or registration.
- "Availability Period" means the period from and including the Effective Date to and including the date that is one month prior to the Termination Date.
- "Bankruptcy Code" means title 11 of the United States Code entitled "Bankruptcy", codified as 11 U.S.C. Section 101 et seq., as now and hereafter in effect.
- "Bankruptcy Court" means the United States Bankruptcy Court for the Southern District of New York or any other court having jurisdiction over the Cases from time to time.
- "Board" means the Board of Governors of the Federal Reserve System of the United States (or any successor).
- "Bringdown Date" means, for the three month period commencing on the date that is three months after the Effective Date, and for each three month period thereafter until the Termination Date, the first Transaction Date that occurs in such period.
- "Business Day" means a day (other than a Friday, Saturday or a Sunday) on which banks are open for business in the Kingdom of Bahrain, London and New York.
- "Call Options" means any option granted by any Person to any LT CayCo or any Investment Company to purchase the Equity Interests of any other Investment Company.
- "Carve-Out" means, except as otherwise provided by the Orders, (i) any unpaid fees due to the United States Trustee pursuant to 28 U.S.C. section 1930 or otherwise and any fees due to the clerk of the Bankruptcy Court, (ii) all reasonable fees and expenses approved by the Bankruptcy Court incurred by a trustee under Section 726(b) or 1104 of the Bankruptcy Code in an aggregate amount not exceeding \$25,000, (iii) the reasonable and documented expenses of members of the Committee appointed in the Cases (excluding fees and expenses of professional persons employed by the Committee and/or such Committee members individually) in an aggregate amount not exceeding \$200,000, (iv) to the extent allowed at any time, all unpaid fees and expenses allowed by the Bankruptcy Court of professionals or professional firms retained pursuant to sections 327, 328, 330, 363 or 1103 of the Bankruptcy Code (the "Professional Persons") and the reasonable fees and expenses of the Joint Provisional Liquidators, in each case that were accrued or incurred, as applicable, through the date upon which AIHL and the Committee receives from the Investment Agent a written notification of the occurrence of an Event of Default and the intention to invoke the Carve-Out (a "Carve-Out Notice"), and (v) to the extent allowed at any time, all fees and expenses of Professional Persons and the Joint Provisional Liquidators incurred after the date upon which AIHL and counsel for the Committee receive from the Investment Agent a Carve-Out Notice, in an aggregate amount not to exceed \$15,000,000; provided that (1) the dollar limitations in clause (v)

on fees and expenses shall not be reduced or increased by the amount of any compensation or reimbursement of expenses incurred by or paid to any Professional Person or the Joint Provisional Liquidators prior to the date AIHL and counsel for the Committee receive from the Investment Agent a Carve-Out Notice or by any fees, expenses, indemnities, or other amounts paid to the Investment Agent, any Participant, or their respective attorneys or agent under this Agreement or otherwise, and (2) to the extent the dollar limitation in **clause** (v) on fees and expenses is reduced by an amount as a result of the payment of such fees and expenses during the continuation of and Event of Default and after delivery of a Carve-Out Notice, and such Event of Default is subsequently cured or waived and the Carve-Out Notice rescinded in writing, such dollar limitation shall be increased by an amount equal to the amount by which it has been so reduced.

- "Cases" means the cases commenced under chapter 11 of the Bankruptcy Code by Arcapita Bank and its affiliated debtors and debtors in possession by the filing of voluntary petitions with the Bankruptcy Court (other than the case commenced by Falcon).
- "Cayman Charge" means the Charges Over Shares entered into by Arcapita Bank in favor of the Security Agent pursuant to which the Security Agent has been granted Liens over the Equity Interests Arcapita Bank owns in its direct Cayman Islands Subsidiaries.
- "Cayman Debentures" means the Debentures entered into between an Obligor and the Security Agent pursuant to which the Security Agent has been granted Liens on any of the Collateral.
- "Cayman Proceedings" means the provisional liquidation of AIHL being performed in the Cayman Islands for which Simon Appell and Gordon MacRae were appointed joint provisional liquidators (the "Joint Provisional Liquidators") of AIHL on March 20, 2012.
- "Cayman Validation Order" means the validation order referred to in clause 6.3 of Schedule 1.
- "Change of Control" means any Person or group of Persons acting in concert gains Control of any Obligor. For the purpose of this definition, "acting in concert" means, in the reasonable opinion of the Investment Agent, a group of Persons who, pursuant to an agreement or understanding (whether formal or informal), actively cooperate, to obtain or consolidate Control of any Obligor.
- "Collateral" means the property of the Debtors, whether now owned or hereafter acquired, upon which a Lien is purported to be granted under the Orders in favor of any or all of the Finance Parties, and the property of the Obligors, whether now owned or hereafter acquired, upon which a Lien is purported to be granted under any Finance Document in favor of any or all of the Finance Parties.
- "Commitment Letter" means the commitment letter, dated November 1, 2012, between AIHL and the Investment Agent.

- "Committee" means the official committee of unsecured creditors appointed by the Office of the United States Trustee in the Cases.
- "Commodities" means, in relation to a Purchase Contract, the commodities specified in a Transaction Request, which may comprise Shari'ah compliant London Metal Exchange metals, platinum group metals or such other Shari'ah compliant commodities as may be agreed from time to time by AIHL and the Investment Agent and, in any event, will only include allocated commodities physically located outside of the United Kingdom.
- "Conditional Amount" has the meaning given in clause 9.4(a).
- "Conditional Increase" has the meaning given in clause 9.4(a).
- "Consolidated Financial Statements" means, starting in June 2013, the audited consolidated financial statements of AIHL and Arcapita Bank for each of its financial years ending June 30.
- "Contract Period" means, for any Purchase Contract, at any time, the one-month period commencing on the Transaction Date for such Purchase Contract and ending on the Deferred Payment Date applicable to such Purchase Contract.
- "Contribution" has the meaning given in the Investment Agency Agreement.
- "Control" means in respect of any Person:
- (a) the power (whether by way of ownership of shares, proxy, contract, agency or otherwise) to:
 - (i) cast, or control the casting of, more than one-half of the maximum number of votes that might be cast at a general meeting of that Person; or
 - (ii) appoint or remove all, or the majority, of the directors or other equivalent officers of that Person; or
 - (iii) give directions with respect to the operating and financial policies of that Person which the directors or other equivalent officers of that Person are obliged to comply with; or
- (b) the holding of more than one-half of the issued voting share capital of the Person.
- "Cost Price" means the amount (in Dollars) payable or paid by the Investment Agent to the Seller for the purchase of Commodities by the Investment Agent (on a spot basis on the value date upon which the payment is made, or is to be made) to be on-sold by the Investment Agent to AIHL under a Purchase Contract.
- "Currency of Obligations" has the meaning given in clause 16.3(a).
- "Currency of Payment" has the meaning given in clause 16.3(a).

► ["DD&Co Ltd Agreements" means:

- (a) the Letter of Understanding dated on or about the date of this Agreement and made between the Seller and the Investment Agent; and
- (b) the letter from the Seller to the Investment Agent dated on or about the date of this Agreement relating to such Letter of Understanding.]
- "**Debtors**" means Arcapita Bank, AIHL, AIHL Sub, WTHL, AEID II and RailInvest, all in their capacities as debtors and as debtors-in-possession in the Cases (and, for the avoidance of doubt, not including Falcon).
- "Default" means an Event of Default or any event or circumstance specified in clause 14 which, with the giving of any notice, the lapse of time, determination of materiality, or fulfillment of any other applicable condition (or any combination of the foregoing), would constitute an Event of Default.
- "Deferred Payment Date" means, in respect of a Deferred Sale Price, the date set out in the Offer Letter applicable to that Deferred Sale Price and which shall be the date falling on the last day of the 1 month period following the Transaction Date selected in the applicable Transaction Request, but if:
- (a) there is no numerically corresponding day in the succeeding calendar month, such date shall be the last Business Day of such succeeding calendar month;
- (b) such date is not a Business Day, then the next Business Day in the same calendar month if there is one, or the preceding Business Day if there is not; or
- (c) such date otherwise would fall after the Termination Date, the Termination Date.
- "Deferred Sale Price" means, in relation to a Purchase Contract, the amount (in Dollars) payable by AIHL to the Investment Agent for the purchase of Commodities, calculated in accordance with clause 5.2(e), excluding any Tax, if applicable.
- "DIP Budget" means the budget to be prepared by AIHL and agreed with the Investment Agent prior to the date hereof, as such initial budget may be modified or supplemented from time to time by additional budgets (covering any time period covered by a prior budget or covering additional time periods), with the written consent of the Investment Agent to be given in its sole discretion.

"Disclosure Schedule" means Schedule 4.

"Dollars" and "\$" means lawful money of the United States of America.

"Effective Date" means the date on which the conditions precedent specified in clauses 3.1 and 3.2 (other than clause 3.2(b)) are satisfied or waived in accordance with the terms hereof and the initial Purchase Contract is entered into pursuant hereto.

"Embargoed Person" means any Person that (a) is publicly identified on the most current list of "Specially Designated Nationals and Blocked Persons" published by OFAC or resides, is organized or chartered, or has a place of business in a country or territory subject to OFAC sanctions or embargo programs (a "Sanctioned Country") or (b) is publicly identified as prohibited from doing business with the United States under the International Emergency Economic Powers Act, the Trading With the Enemy Act, or any other requirement of any Governmental Authority.

"Equity Interest" shall mean, with respect to any Person, any and all shares, interests, participations or other equivalents, including membership interests (however designated, whether voting or non-voting), of equity of such Person, including, if such Person is a partnership, partnership interests (whether general or limited) and any other interest or participation that confers on a Person the right to receive a share of the profits and losses of, or distributions of assets of, such partnership, whether outstanding on the Effective Date or issued thereafter, but excluding debt securities convertible or exchangeable into such equity.

"Eurolog IPO" shall have the meaning set forth in the SCB Order.

"Event of Default" means any event specified as such in clause 14.

"Existing Security" means:

- (a) the first ranking charge over the shares of AIHL Sub dated May 30, 2011 granted by AIHL in favor of Standard Chartered Bank;
- (b) the second ranking equitable mortgage over the shares of AIHL Sub dated December 22, 2011 granted by AIHL in favor of Standard Chartered Bank;
- (c) the first ranking charge over the shares of WTHL dated May 30, 2011 granted by AIHL Sub in favor of Standard Chartered Bank;
- (d) the second ranking equitable mortgage over the shares of WTHL dated December 22, 2011 granted by AIHL Sub in favor of Standard Chartered Bank;
- (e) the first ranking equitable mortgage over the shares of AEID II dated December 22, 2011 granted by AIHL Sub in favor of Standard Chartered Bank;
- (f) the second ranking equitable mortgage ► [and charge] over the shares of AEID II dated December 28, 2011 granted by AIHL Sub in favor of Standard Chartered Bank;
- (g) the first ranking equitable mortgage over the shares of RailInvest dated December 22, 2011 granted by AIHL Sub in favor of Standard Chartered Bank; and

(h) the second ranking equitable mortgage and charge over the shares of RailInvest dated December 28, 2011 granted by AIHL Sub in favor of Standard Chartered Bank.

"Existing US\$ Facilities" means

- (a) the \$50 million master murabaha agreement dated May 30, 2011 as amended on October 2, 2011, November 2, 2011, November 29, 2011, December 28, 2011, January 30, 2012, February 13, 2012, February 28, 2012, March 14, 2012, and as amended pursuant to the SCB Order, between Arcapita Bank and Standard Chartered Bank, and, solely to the extent permitted in accordance with clause 12.32, as amended, supplemented or otherwise modified from time to time after the Effective Date; and
- (b) the \$50 million master murabaha agreement dated December 22, 2011 as amended on January 30, 2012, February 13, 2012, February 28, 2012, March 14, 2012 and as amended pursuant to the SCB Order, between Arcapita Bank and Standard Chartered Bank, and, solely to the extent permitted in accordance with clause 12.32, as amended, supplemented or otherwise modified from time to time after the Effective Date.

"Exit Plan Subsidiary" means any Subsidiary of Arcapita Bank designated in writing to the Investment Agent by AIHL as an "Exit Plan Subsidiary," that (a) is formed in accordance with, and solely for the purpose of, implementing a Reorganization Plan and (b) prior to the effective date of a Reorganization Plan, has no operations and no assets other than those received in connection with any minimum capitalization requirements of such Exit Plan Subsidiary's jurisdiction of organization.

"Extended Maturity Date" has the meaning given in clause 2.3(a).

"Extension Availability Date" means the date that the Investment Agent receives a report from KPMG LLP, which report shall be in form and substance satisfactory to the Investment Agent and shall cover the information required to be delivered pursuant to clause 12.2(a)(vi), and such additional information as the Investment Agent may reasonably request.

"Extension Effective Date" has the meaning given in clause 2.3(b).

"Facility" as defined in clause 2.1.

"Facility Commitment" has the meaning given in the Investment Agency Agreement.

"Facility Limit" means, at any time, the total amount of the Investment Agent's Cost Price commitment to purchase Commodities at such time under this Agreement, as such amount may be reduced, increased or terminated pursuant to clause 9 or clause 14.20; provided, however, that such amount may never be greater than that which is authorized under the Orders. Subject to the proviso in the preceding sentence, the Facility Limit on the Effective Date is \$150,000,000, as such amount may be reduced, increased or

12-11076-shl Doc 690-2 Filed 12/04/12 Entered 12/04/12 18:01:19 Exhibit B - DIP Agreement Pg 12 of 97

terminated pursuant to **clause 9** or **clause 14.20**; *provided* that, prior to the Increase Effective Date, the Facility Limit shall be reduced by the Conditional Amount.

"Facility Office" means the office notified by a Participant to the Investment Agent:

- (a) on or before the date it becomes a Participant; or
- (b) by not less than 5 Business Days' notice, as the office(s) through which it will perform all or any of its obligations under the Investment Agency Agreement.

"Falcon" means Falcon Gas Storage Company, Inc.

"Falcon Escrow Agreement" means that certain Escrow Agreement, dated as of April 1, 2010, by and among Alinda Natural Gas Storage I, L.P., Alinda Natural Gas Storage II, L.P., Falcon, and HSBC Bank USA.

"FCPA" has the meaning given in clause 11.22(c).

"Fee Letter" means the fee letter dated on or about the date of this Agreement between AIHL and the Investment Agent, Security Agent and Arranger relating to certain fees payable in relation to this Agreement.

"Final Order" means an order of the Bankruptcy Court entered in the Cases, in substantially the form of the Interim Order, with such modifications thereto as are satisfactory to the Investment Agent in its sole discretion, as the same may be amended, modified or supplemented from time to time with the consent of the Investment Agent (which consent shall be given or withheld in the sole discretion of the Investment Agent).

"Final Order Entry Date" means the date the Final Order is entered in the Cases.

"Finance Documents" means:

- (a) this Agreement;
- (c) the Investment Agency Agreement;
- (d) each Guarantee;
- (e) the Security Documents;
- (f) when entered into, each Purchase Contract;
- (g) the Netting Letter;
- (h) the Fee Letter; and
- (i) such other documents at any time designated as such by the parties.

12-11076-shl Doc 690-2 Filed 12/04/12 Entered 12/04/12 18:01:19 Exhibit B - DIP Agreement Pg 13 of 97

"Finance Parties" means the Investment Agent, the Security Agent, the Arranger and the Participants.

"Financial Indebtedness" means any indebtedness for or in respect of:

- (a) moneys borrowed;
- (b) any amount raised by acceptance under any acceptance credit facility;
- (c) any amount raised pursuant to any note purchase facility or the issue of sukuk, bonds, notes, debentures, loan stock or any similar instrument, including any debt securities convertible or exchangeable into Equity Interests;
- (d) the amount of any liability in respect of any lease or hire purchase contract which would, in accordance with IFRS, be treated as a finance or capital lease;
- (e) receivables sold or discounted (other than any receivables to the extent they are sold on a non-recourse basis);
- (f) any amount raised under any other transaction (including any purchase agreement and any murabaha agreement) having the commercial effect of a borrowing;
- (g) any obligations in respect of one or more Hedging Agreements (for purposes of determining Financial Indebtedness under this **clause** (g), the "principal amount" of the obligations of any Obligor or any of its Subsidiaries in respect of any Hedging Agreement at any time shall be the maximum aggregate amount (giving effect to any netting arrangements) that such Obligor or Subsidiary would be required to pay if such Hedging Agreement were terminated at such time);
- (h) the supply of any goods or services which is more than 30 days past the original due date for payment;
- (i) any counter-indemnity or reimbursement obligation in respect of a guarantee, indemnity, bond, standby or documentary letter of credit or any other instrument issued by a bank, surety or financial institution;
- (j) any obligation of others secured by (or for which the holder of such indebtedness has an existing right, contingent or otherwise, to be secured by) any Lien on assets owned or acquired by such Person, whether or not the obligations secured thereby have been assumed; and
- (k) (without double counting) the amount of any liability in respect of any guarantee or indemnity for any of the items referred to in **paragraphs** (a) to (i) above.

provided, however, that any obligations arising in connection with professional fees and expenses incurred in connection with the Cases shall not constitute or comprise Financial Indebtedness hereunder.

12-11076-shl Doc 690-2 Filed 12/04/12 Entered 12/04/12 18:01:19 Exhibit B - DIP Agreement Pg 14 of 97

- "Financial Statements" means, starting June 2013, the audited financial statements of Arcapita Bank and AIHL for each of their financial years ending June 30.
- "Governmental Authority" means any federal, state, provincial, municipal, national, foreign or other government, governmental department, commission, board, bureau, court, tribunal, agency or instrumentality or political subdivision thereof, any government sponsored entity or any authority, body, regulatory or self-regulatory organization or other entity or officer exercising executive, legislative, judicial (including any arbitrator), statutory, regulatory or administrative functions of or pertaining to any government or any court (including any supranational bodies such as the European Union), in each case whether associated with the United States or any state, commonwealth, province, district or territory thereof, or a foreign entity or government.
- "Governmental Authorization" means any permit, license, authorization, plan, directive, consent order or consent decree of or from any Governmental Authority.
- "Group" means Arcapita Bank and AIHL and their respective Subsidiaries from time to time, other than a Subsidiary which is an Investment Company.
- "Guarantee" means each joint and several guarantee by the Guarantors in favor of the Security Agent dated on or about the date of this Agreement, and each Additional Guarantee.
- "Guarantor" means each of Arcapita Bank, AEID II, AIHL Sub, RailInvest, WTHL, AIML, Arcapita Inc., Arcapita Structured Finance Ltd., Arcapita Investment Funding Limited, Arcapita Industrial Management I Ltd, Arcapita (US) Limited, Arcapita (Europe) Limited, Arcapita (Singapore) Limited, each other LT CayCo, each WCF and each Syndication Company, in each case that is wholly-owned by AIHL, and each Additional Guarantor that becomes a Guarantor pursuant to clause 12.25(c), other than (x) those Persons identified on Schedule 5F hereto (which schedule shall include the Specified Non-Guarantor Subsidiaries) and (y) each Exit Plan Subsidiary.
- "Hedging Agreement" means (a) any and all rate swap transactions, basis swaps, credit derivative transactions, forward rate transactions, commodity swaps, commodity options, forward commodity contracts, equity or equity index swaps or options, bond or bond price or bond index swaps or options or forward bond or forward bond price or forward bond index transactions, interest rate options, forward foreign exchange transactions, cap transactions, floor transactions, collar transactions, currency swap transactions, cross-currency rate swap transactions, currency options, spot contracts, or any other similar transactions or any combination of any of the foregoing (including any options to enter into any of the foregoing), whether or not any such transaction is governed by or subject to any master agreement, and (b) any and all transactions of any kind, and the related confirmations, which are subject to the terms and conditions of, or governed by, any form of ISDA Master Agreement, including any such obligations or liabilities under any ISDA Master Agreement.

12-11076-shl Doc 690-2 Filed 12/04/12 Entered 12/04/12 18:01:19 Exhibit B - DIP Agreement Pg 15 of 97

- "Holding Company" means, in relation to a company or corporation, any other company or corporation in respect of which it is a Subsidiary.
- "IFRS" means International Financial Reporting Standards from time to time, published by the International Accounting Standards Board.
- "Immaterial Subsidiary" means the Specified Non-Guarantor Subsidiaries and any Subsidiary of Arcapita Bank (other than an Obligor) designated in writing to the Investment Agent by AIHL as an "Immaterial Subsidiary" (which designation may be withdrawn by AIHL in a writing delivered to the Investment Agent), that, individually as of the relevant date of determination has total assets as of such date of less than \$2,000,000, as determined in accordance with IFRS; provided that at no time shall all Immaterial Subsidiaries have in the aggregate total assets in excess of \$5,000,000, as determined in accordance with IFRS. The Immaterial Subsidiaries on the Effective Date are listed on **Schedule 5G**.
- "Increase Availability Date" means, if the Rejection Date has not occurred, the later to occur of (a) December 31, 2012 and (b) the date that the Investment Agent provides written notice to AIHL that the Investment Agent has completed additional confirmatory due diligence the results of which are reasonably satisfactory to the Investment Agent.
- "Increase Effective Date" has the meaning given in clause 9.4(b).
- "Increased Costs" has the meaning given in clause 16.5(b).
- "Increased Costs Amount" means Increased Costs incurred by a Finance Party.
- "Indemnified Party" has the meaning given in clause 16.1(a).
- "Information Exception" means misstatements or omissions which would not be reasonably expected to decrease the aggregate value to Arcapita of its assets and Investment Companies based on the low-point valuations contained in the KPMG Reports and Waterfalls delivered to the Investment Agent prior to the Effective Date by more than \$70,000,000 (calculated as of the date in question and with such adjusted valuation for a given asset or Investment Company calculated pursuant to the methodology similar to the methodology utilized in the low-point valuations in the KPMG Reports and the Waterfalls).
- "Inter-Obligor Indebtedness" means any Financial Indebtedness owing by any Obligor to any other Obligor.
- "Interim Facility Limit" means \$25,000,000, as such amount may be reduced, increased or terminated pursuant to clause 9 or clause 14.13; *provided*, however, that such amount may never be greater than that which is authorized under the Interim Order.
- "Interim Order" means an order of the Bankruptcy Court entered in the Cases granting interim approval of the transactions involving the Debtors contemplated by this Agreement and the other Finance Documents and granting the Liens on the Collateral and

the Superpriority Claims in favor of the Investment Agent and the Participants, substantially in the form of **Exhibit A** hereto, with such changes thereto as may be in form and substance satisfactory to the Investment Agent in its sole discretion, as the same may be amended, modified or supplemented from time to time with the consent of the Investment Agent (such consent to be given or withheld in the sole discretion of the Investment Agent).

"Interim Order Entry Date" means the date on which the Interim Order is entered by the Bankruptcy Court.

"Investment Agency Agreement" means the agreement defined as such in Recital (A).

"Investment Agent" has the meaning given in the Recitals.

"Investment Companies" means the Transaction Holdcos and any entity in which the Transaction Holdcos have a direct or indirect Equity Interest.

"Investment Company Murabaha Facility" means each murabaha financing facility between Arcapita Bank or any of its Subsidiaries and an Investment Company from time to time.

"Investment Company Murabaha Facility Assignment" means each assignment of an Investment Company Murabaha Facility, between Arcapita Bank or any of its Subsidiaries (as assignor) and the Security Agent, in a form acceptable to the Security Agent.

"Investments" has the meaning given in clause 12.16.

"ISDA Master Agreement" shall mean the Master Agreement (Multicurrency-Cross Border) published by the International Swap and Derivatives Association, Inc., as in effect from time to time.

"Joint Provisional Liquidators" has the meaning given in the definition of Cayman Proceedings.

"KPMG Reports" means those reports prepared by KPMG LLP provided to the Investment Agent on October 24, 2012.

"LIBOR" means for any Contract Period with respect to a Purchase Contract, the rate per annum (rounded upwards, if necessary, to the next 1/16 of 1%) equal to the rate appearing on Reuters Page LIBOR01 (or on any successor or substitute page of such service, or any successor to or substitute for such service, providing rate quotations comparable to those currently provided on such page of such service, as determined by the Investment Agent from time to time for purposes of providing quotations of interest rates applicable to Dollar deposits in the London interbank market) at approximately 11:00 a.m. (London time), two (2) Business Days prior to the commencement of such Contract Period, as the rate for Dollar deposits with a maturity comparable to such Contract Period. In the event that such rate is not available at such time for any reason,

then "LIBOR" with respect to such Purchase Contract for such Contract Period shall be determined by reference to such other comparable publicly available service for displaying LIBOR for such for such Contract Period as selected by an agreement between AIHL and the Investment Agent.

"Lien" means (a) any lien (statutory or other), mortgage, deed of trust, pledge, hypothecation, preference, participation interest, assignment, deposit arrangement, security interest, charge or encumbrance of any kind (including any agreement to give any of the foregoing) and any option, trust or other preferential arrangement having the practical effect of any of the foregoing, (b) the interest of a vendor or a lessor under any conditional sale agreement, capital lease or title retention agreement (or any financing lease having substantially the same economic effect as any of the foregoing), and (c) in the case of Equity Interests, any purchase option, call or similar right of a third party with respect to such Equity Interests.

"LT CayCo" means each long term portfolio investment company that is a whollyowned Subsidiary of AIHL at any time, including each of the companies listed on Schedule 5A (which AIHL represents is a complete list of all LT CayCos in existence on the Effective Date).

"Lusail Obligations" has the meaning given in clause 12.13(b)(ii).

"Mandatory Cost" means the cost of compliance in respect of this Facility with:

- (a) for Participants with their Facility Office in the United Kingdom, the requirements of the Bank of England and/or the Financial Services Authority;
- (b) for Participants with their Facility Office in other countries of the European Union, the requirements of the European Central Bank; and
- (c) for Participants with their Facility Office in any other country, the equivalent requirements of that jurisdiction.

"Material Adverse Effect" means a material adverse effect on or a material adverse change in:

- (a) the condition (financial or otherwise), business, operations, assets or liabilities of AIHL and its Subsidiaries, taken as a whole, other than as customarily result from the continuation of a bankruptcy case; or
- (b) the ability of an Obligor to perform its obligations under the Finance Documents; or
- (c) the validity or enforceability of, or the effectiveness or ranking of any Lien granted or purported to be granted to the Security Agent pursuant to, any Order or any of the Finance Documents or the material rights or remedies of any Finance Party under any of the Finance Documents.

12-11076-shl Doc 690-2 Filed 12/04/12 Entered 12/04/12 18:01:19 Exhibit B - DIP Agreement Pg 18 of 97

"Maturity Date" means ▶ [June __, 2013], as such date may be extended pursuant to clause 2.3; provided that if such date is not a Business Day, the Maturity Date shall be the immediately preceding Business Day.

"Maturity Date Extension" has the meaning given in clause 2.3(a).

"Net Cash Proceeds" means, with respect to any Prepayment Event, the proceeds of any refund of Taxes, conveyance, lease, sublease, assignment, transfer, disposition, dividend, distribution, issuance, sale, initial public offering, insurance proceeds or condemnation award (or other covered transaction), as the case may be, received in the form of cash and cash equivalents (including any such proceeds received by way of deferred payment of principal pursuant to a note or installment receivable or purchase price adjustment receivable or otherwise, but only as and when received), net of discounts, commissions, costs, awards, attorneys' fees, accountants' fees, investment banking fees and other customary fees and expenses actually incurred in connection therewith (other than fees paid to an Affiliate) and, in connection with any disposition of assets, amounts required to be applied to the repayment of (a) the Existing \$US Facilities solely to the extent required by the SCB Order and (b) Financial Indebtedness secured by a Lien expressly permitted hereunder (and not junior to the Liens granted on such assets under the Financing Documents or pursuant to the Orders) on any assets so disposed of (other than any Lien securing the Obligations) and net of taxes paid or reasonably estimated to be payable as a result thereof (after taking into account any available tax credits or deductions and any tax sharing arrangements).

"Netting Letter" means the letter agreement between the Investment Agent, AIHL, DD&Co Ltd and Condor Trade Limited dated on or about the date of this Agreement.

"Nominee Declarations" means a nominee declaration held by any Person in the Equity Interests of any Investment Company held for the beneficial use and ownership of any LT CayCo or any other Investment Company.

"Obligations" means all liabilities and obligations of every kind and description (whether or not evidenced by any instrument and whether or not for the payment of money), direct or indirect (including those acquired by assumption), absolute or contingent, due or to become due, now existing or hereafter arising, in each case of any Obligor from time to time owed to the Investment Agent, any other Finance Party, or any of them under any Finance Document, whether for the Deferred Sale Price, Cost Price, Profit Amount (including Profit Amount which, but for the filing of a petition in bankruptcy with respect to such Obligor, would have accrued on any Obligation, whether or not a claim is allowed against such Obligor for such Profit Amount in the related bankruptcy proceeding), costs, fees, charges, expenses, attorneys' fees, accountants' fees, consultants' fees, indemnification or otherwise.

"Obligors" means AIHL and the Guarantors.

"OFAC" means the Office of Foreign Assets Control of the US Department of the Treasury.

"Offer Letter" means an offer in writing from the Investment Agent to AIHL (offering to enter into a Purchase Contract) substantially in the form set out in **Schedule 3**.

"Orders" means the collective reference to the Interim Order and the Final Order.

"Original Financial Statements" means:

- (a) in respect of AIHL, the audited consolidated financial statements of Arcapita Bank for the year ending December 31, 2011; and
- (b) in respect of Arcapita Bank, its audited consolidated financial statements for the year ending December 31, 2011.

"Participants" means those banks, financial institutions and other Persons listed in Schedule 1 to the Investment Agency Agreement or any assignee or transferee which has become a Participant in accordance with clause 11 of the Investment Agency Agreement.

"Participation" has the meaning given in the Investment Agency Agreement.

"Patriot Act" means the Uniting and Strengthening America by Providing Appropriate Tools Required to Intercept and Obstruct Terrorism Act (USA PATRIOT ACT) of 2001 (Title III of Pub. L. 107-56), as the same may be amended from time to time, and corresponding provisions of future laws.

"Permitted Liens" means:

- (a) any Lien for taxes not yet due or delinquent or being contested in good faith by appropriate proceedings for which adequate reserves have been established in accordance with generally acceptable accounting principles;
- (b) any statutory Lien arising in the ordinary course of business by operation of law with respect to a liability that is not yet due or delinquent;
- (c) any Lien created by operation of law, such as materialmen's liens, mechanics' liens and other similar Liens, arising in the ordinary course of business with respect to a liability that is not yet due or delinquent or that are being contested in good faith by appropriate proceedings;
- (d) Liens (A) upon or in any equipment acquired or held by the Obligors or any of their subsidiaries to secure the purchase price of such equipment or indebtedness incurred solely for the purpose of financing the acquisition or lease of such equipment, or (B) existing on such equipment at the time of its acquisition, provided that the Lien is confined solely to the property so acquired and improvements thereon, and the proceeds of such equipment;
- (e) easements, rights-of-way, municipal and zoning and building ordinances and similar charges, encumbrances, title defects or other irregularities, restrictions of Governmental Authorities on the use of property or conduct of business, and

Liens in favor of Governmental Authorities and public utilities, that do not materially interfere with the ordinary course of business of the Obligors and their subsidiaries, taken as a whole;

- (f) Liens arising out of conditional sale, title retention, consignment or similar arrangements for the sale of goods entered into by the Obligors or any of their subsidiaries in the ordinary course of business of the Obligors or any of their subsidiaries;
- (g) Liens incurred in connection with the extension, renewal or refinancing of the indebtedness secured by Liens of the type described in **clauses** (a) through (f) above, provided that any extension, renewal or replacement Lien shall be limited to the property encumbered by the existing Liens and the principal amount of the indebtedness being extended, renewed or refinanced does not increase;
- (h) leases or subleases and licenses and sublicenses granted to others in the ordinary course of the Obligors' business, not interfering in any material respect with the business of the Obligors and their subsidiaries taken as a whole;
- (i) Liens in favor of the Finance Parties securing the Obligations under the Facility Documents; and
- (j) Liens in favor of customs and revenue authorities arising as a matter of law to secure payments of custom duties in connection with the importation of goods.

"Permitted Representation Exceptions" means (a) with respect to representations and warranties contained in clause 11.9(a), changes resulting solely from transactions permitted by clauses 12.15, 12.16, 12.22 or 12.27), and (b) with respect to representations and warranties contained in clause 11.19, changes resulting solely from actions taken by (i) holders of Equity Interests of Syndication Companies or PVs (that are not, in each case, Affiliates (excluding Affiliates under clause (a) of the second sentence of the definition thereof) or employees of any Obligor) pursuant to shareholder agreements, proxies, administration agreements, management agreements or similar agreements in effect on the Effective Date or (ii) authorized officers of any Syndication Company or PV, solely to the extent that such actions are taken at the direction of the Person described in clause (i) above that directly holds Equity Interests in such Syndication Company or PV.

"Person" means and includes any natural person, corporation, limited partnership, general partnership, limited liability company, limited liability partnership, joint stock company, joint venture, association, company, trust, bank, trust company, land trust, business trust or other organization, whether or not a legal entity, and any Governmental Authority.

"Petition Date" means March 19, 2012.

"Prepayment Event" means:

- (a) any conveyance, sale, lease, sublease, assignment, transfer or other disposition (including by way of merger or consolidation and including any sale and leaseback transaction) of any property (including the Equity Interests by the holder thereof) by any Obligor, or any other member of the Group to any Person other than (i) any Obligor and (ii) in the case of any member of the Group that is not an Obligor, to any other member of the Group that is not an Obligor, excluding any such conveyance, sale, lease, sublease, assignment, transfer or other disposition permitted in accordance with clause 12.15(b);
- (b) any dividend, distributions or other amounts paid by any Syndication Company, Investment Company, any non-wholly owned Subsidiary or any other Person to any Obligor or any other member of the Group, other than received from Falcon constituting distributions under the Falcon Escrow Agreement and customary management fees paid in accordance with past practice;
- (c) any issuance or sale by any Obligor or any other member of the Group of its Equity Interests to any Person other than (i) to any Obligor, (ii) in the case of any member of the Group that is not an Obligor, to any other member of the Group that is not an Obligor and (iii) Equity Interests issued or sold in connection with a Reorganization Plan;
- (d) any incurrence, issuance or sale by any Obligor or any other member of the Group of any Financial Indebtedness after the Effective Date other than (i) as permitted by **clause 12.13** and (ii) securities issued in connection with a Reorganization Plan;
- (e) any refund of Taxes received by any Obligor or any other member of the Group;
- (f) unless the Investment Agent otherwise agrees in its sole discretion, any initial public offering (other than the Eurolog IPO) of the Equity Interests of any Obligor or any member of the Group; and
- (g) any event that gives rise to the receipt by any Obligor or any other member of the Group of any insurance proceeds or condemnation awards in respect of any assets or properties.
- "Prepetition Payment" means a payment (by way of adequate protection or otherwise) of principal, interest, profit, fees or otherwise on account of any prepetition indebtedness, trade payables (including, without limitation, in respect of reclamation claims) or other prepetition claims against any Debtor.

"Profit Amount" means:

Cost Price * Profit Rate * (N/360)

where:

12-11076-shl Doc 690-2 Filed 12/04/12 Entered 12/04/12 18:01:19 Exhibit B - DIP Agreement Pg 22 of 97

- N is the number of days to elapse from, and including, the proposed Transaction Date to, but excluding, the Deferred Payment Date.
- "**Profit Rate**" means the sum of (a) the greater of (i) LIBOR determined, with respect to any Purchase Contract, on the Transaction Date by the Investment Agent and (ii) 2% per annum plus (b) 10% per annum.
- "Proposed Additional Cost Price" has the meaning given in clause 4.1(b).
- "Purchase Contract" means the agreement for the sale by the Investment Agent of Commodities and the purchase of those Commodities by AIHL pursuant to clause 5 and shall include each Subsequent Purchase Contract.
- "PVs" means the investment vehicles organized in the Cayman Islands and owned by a group of 50 third party investors (including insiders of the Debtors), each of whom invest in each U.S.-based AIHL portfolio investment, including each of the companies listed on **Schedule 5D** (which AIHL represents is a complete list of all PVs in existence on the Effective Date).
- "RailInvest" means RailInvest Holdings Limited an exempted company incorporated with limited liability under the laws of the Cayman Islands with company number 211119, and a debtor and a debtor-in-possession in the Cases.
- "Recipient" has the meaning given in clause 8.4(b).
- "**Regulation T**" means Regulation T of the Board as from time to time in effect and all official rulings and interpretations thereunder or thereof.
- "**Regulation** U" means Regulation U of the Board as from time to time in effect and all official rulings and interpretations thereunder or thereof.
- "Regulation X" means Regulation X of the Board as from time to time in effect and all official rulings and interpretations thereunder or thereof.
- "Rejection Date" means the date, if any, prior to December 31, 2012, that AIHL provides written notice to the Investment Agent that AIHL is no longer interested in the Investment Agent providing the Conditional Amount on behalf of the Participants (at which time the Investment Agent shall be relieved of any obligation to continue its confirmatory due diligence with respect to, or to seek to provide, the Conditional Amount).
- "Relevant Party" has the meaning given in clause 8.4(b).
- "Remedies Notice Period" has the meaning given in clause 14.20(c).
- "Reorganization Plan" means a plan of reorganization or plan of liquidation in the Cases of the Debtors.

12-11076-shl Doc 690-2 Filed 12/04/12 Entered 12/04/12 18:01:19 Exhibit B - DIP Agreement Pg 23 of 97

"Reportable Investment Assets" means each investment asset of AIHL, the value of which has been reduced by 30% or more from the KPMG LLP mid-point current valuations for such investment asset as set forth in the Waterfalls delivered to the Investment Agent prior to the Effective Date, but only to the extent that such investment asset was valued in excess of \$10,000,000 in such Waterfall.

"Requested Extension Date" has the meaning given in clause 2.3(a).

"Responsible Officer" means the Chief Executive Officer and each Executive Director of AIHL.

"Sanctioned Country" has the meaning given in the definition of "Embargoed Person".

"SCB" means Standard Chartered Bank.

"SCB Order" means the Order pursuant to section 105(a) of the Bankruptcy Code and Bankruptcy Rule 9019, Authorizing and Approving the Settlement with Standard Chartered Bank [Docket No. 587] (including the settlement term sheet annexed thereto), as the same is in effect on the date hereof.

"Security Agent" means Fortress Credit Corp., as security agent for the Participants.

"Security Documents" means:

- (a) the U.S. Security Agreement;
- (b) the Cayman Charge;
- (c) the Cayman Debentures;
- (d) each Investment Company Murabaha Facility Assignment; and
- (e) any other document creating, evidencing or acknowledging Liens in favor of the Security Agent (as agent for the Participants) in respect of the obligations of AIHL or any other Obligor under any of the Finance Documents.

"Security Period" means the period starting on the date of this Agreement and ending on the date all the liabilities of each Obligor under or in connection with each Finance Document are irrevocably discharged in full and no Finance Party has any commitment or liability, whether present or future, actual or contingent, in relation to the Facility.

"Seller" means ► [DD&Co Limited].

"SDN List" means the Office of Foreign Assets Control list of specifically designated nationals and blocked Persons as published by the US Department of the Treasury as updated from time to time or such other list as might replace it.

"SOFA Schedules" means SOFA Schedules D, F and G filed on June 8, 2012, as provided to the Investment Agent on October 24, 2012.

12-11076-shl Doc 690-2 Filed 12/04/12 Entered 12/04/12 18:01:19 Exhibit B - DIP Agreement Pg 24 of 97

"Specified Non-Guarantor Subsidiaries" means Arcapita Limited (England), Arcapita Pte. Limited (Singapore), and Arcapita Hong Kong Limited.

"Specified Reduction" has the meaning given in clause 9.1.2(c).

"**Spot Rate of Exchange**" means in respect of a Finance Party, that Finance Party's spot rate of exchange for the purchase of the relevant currency in the London foreign exchange market at or about 11:00 a.m. (London time) on a particular day.

"Structure Chart" means the structure charts provided to Investment Agent on October 24, 2012.

"Subsequent Purchase Contract" means any Purchase Contract (i) that has a Transaction Date that is the same date as the Deferred Payment Date of another Purchase Contract (such other Purchase Contract, the "Existing Purchase Contract") and (ii) with a Cost Price equal to or less than the Cost Price component of the Deferred Sale Price of the Existing Purchase Contract (provided that there may be only one Subsequent Purchase Contract with respect to any Existing Purchase Contract).

"Subsidiary" means an entity of which a Person (a) has direct or indirect control or (b) owns directly or indirectly more than 50 percent of the voting capital or similar right of ownership; "control" for this purpose means the power to direct the management and the policies of the entity whether through the ownership of voting capital, by contract or otherwise, including, for the avoidance of doubt, the Nominee Declaration; *provided* that no Syndication Company or Investment Company shall be deemed to be a Subsidiary solely by virtue of **clause (a)** above.

"Superpriority Claim" means a claim against any Debtor in any of the Cases which is an administrative expense claim having priority over any and all administrative expenses, diminution claims and all other claims against the Debtors, now existing or hereafter arising, of any kind whatsoever, including, without limitation, all administrative expenses of the kind specified in sections 503(b) and 507(b) of the Bankruptcy Code, and over any and all administrative expenses or other claims arising under sections 105, 326, 328, 331, 365, 503(b), 506(c), 507(a), 507(b), 546, 726, 1113 or 1114 of the Bankruptcy Code.

"Supplier" has the meaning given in clause 8.4(b).

"Syndication Companies" means each of the companies organized in the Cayman Islands that are co-owned by AIHL, on the one hand, and third party investors, on the other hand, for the purpose of funding AIHL's portfolio investments through the sale of Syndication Companies' Equity Interests to third party investors, including each of the companies listed on Schedule 5E, which schedule shall designate whether such company is wholly-owned or majority-owned by any Obligor on the Effective Date, and which AIHL represents is a complete list of all of the Syndication Companies and its direct and indirect ownership interests therein in existence on the Effective Date.

"Tax" means any direct or indirect present or future tax, zakat, impost, charge, duty, levy or any similar assessment whatsoever, including any stamp tax, documentary tax, value

12-11076-shl Doc 690-2 Filed 12/04/12 Entered 12/04/12 18:01:19 Exhibit B - DIP Agreement Pg 25 of 97

added tax, sales tax and duty and withholding tax (including any penalty or expense payable in connection with any failure to pay or delay in paying the same) and Taxation shall be construed accordingly.

"Tax Credit" means a credit against, relief or remission for, or repayment of any Tax.

"**Tax Deduction**" means a deduction or withholding for or on account of Tax from a payment under a Finance Document.

"**Tax Payment**" means a payment made by AIHL to a Finance Party in any way relating to a Tax Deduction or under any indemnity given by AIHL in respect of Tax under any Finance Document.

"Termination Date" means the earliest of (i) the Maturity Date, (ii) 30 days (or such later date as the Investment Agent may agree in its sole discretion) after entry of the Interim Order if the Final Order has not been entered prior thereto, (iii) any date on which the Facility Limit or, prior to the Final Order Entry Date, the Interim Facility Limit, are reduced to zero or otherwise terminated pursuant to the terms hereof, (iv) any date on which the sale of all or substantially all of the Obligors' assets is consummated, (v) the date of termination of this agreement pursuant to the terms hereof, (vi) the effective date of a Reorganization Plan and (vii) the date of the conversion of any of the Cases to a liquidation proceeding under Chapter 7 of the Bankruptcy Code or the dismissal of any of the Cases by the Bankruptcy Court.

"Third Party" means any party other than a Finance Party or member of the Group but which shall include Investment Companies.

"Transaction Date" means the date on which a Purchase Contract is, or is proposed to be, made, being the same date as the Offer Letter.

"Transaction Holdcos" means the parent holding companies of AIHL portfolio companies, the equity of which is wholly owned by the combination of the applicable Syndication Company, the applicable LT CayCo, and Arcapita Incentive Plan Limited, a Cayman Islands company and, in some cases, certain third-party investors, including each of the companies listed on **Schedule 5B**, which schedule shall designate whether such company is wholly-owned by any Obligor on the Effective Date, and which AIHL represents is a complete list of all of the Transaction Holdcos in existence on the Effective Date.

"Transaction Request" means a request from AIHL to the Investment Agent to make a Purchase Contract, substantially in the form set out in **Schedule 2**.

"U.S. Security Agreement" means the Pledge and Security Agreement, dated as of December ___, 2012, between Arcapita Inc., a Delaware corporation, each other Obligor from time to time party thereto, and the Security Agent, as amended, restated, supplemented or otherwise modified from time to time.

"VAT" has the meaning given to it in clause 8.4(a).

12-11076-shl Doc 690-2 Filed 12/04/12 Entered 12/04/12 18:01:19 Exhibit B - DIP Agreement Pg 26 of 97

"Voluntary Prepayment Amount" has the meaning given to it in clause 9.1.4(a).

"Waterfalls" means the analysis performed by AIHL and/or its Affiliates detailing the distribution of proceeds upon disposition of each of the AIHL investments provided to the Investment Agent on October 27, 2012.

"WCFs" means the companies organized in the Cayman Islands that are subsidiaries of AIHL (subject to syndication of the Equity Interests in such companies to third party investors in certain cases), formed to enter into working capital facilities with Transaction Holdcos or subsidiaries of Transaction Holdcos, including each of the companies listed on **Schedule 5C**, which schedule shall designate whether such company is wholly-owned by any Obligor on the Effective Date, and which AIHL represents is a complete list of all of the WCFs and its direct or indirect ownership interest therein in existence on the Effective Date.

"WTHL" means WindTurbine Holdings Limited a company with limited liability incorporated in the Cayman Islands under commercial registration number 211910, and a debtor and a debtor-in-possession in the Cases.

1.2 Construction

Unless the context otherwise requires, in this Agreement:

- (a) words denoting the singular shall include the plural and vice versa, words denoting a gender shall include every gender;
- (b) references herein to any Person shall be construed to include such Person's permitted successors and assigns;
- (c) references to a "party" means a party to this Agreement, including that party's successors in title and assigns or transferees permitted in accordance with this Agreement;
- (d) the use herein of the word "include" or "including," when following any general statement, term or matter, shall not be construed to limit such statement, term or matter to the specific items or matters set forth immediately following such word or to similar items or matters, whether or not no limiting language (such as "without limitation" or "but not limited to" or words of similar import) is used with reference thereto, but rather shall be deemed to refer to all other items or matters that fall within the broadest possible scope of such general statement, term or matter;
- (e) references to Recitals, clauses and Schedules are to those contained in this Agreement and all Schedules are an integral part of this Agreement;
- (f) any reference in this Agreement to any law, decree, regulation, rule, statute or statutory provision shall include interpretations of, and be construed as a reference to, such law, decree, regulation, rule, statute or statutory provision as same may

- have been or may from time to time be amended, modified, extended, consolidated, supplemented, re-enacted or replaced;
- (g) references to this Agreement or to any other agreement and document shall be construed as a reference to such agreement or document as amended, amended and restated, modified, supplemented, novated or replaced from time to time, to the extent such amendment, amendment and restatement, modification, supplement, novation or replacement is permitted hereunder or under any other Finance Document;
- (h) any references in this Agreement to "Articles," "Sections" and/or "Titles" which make reference to any particular piece of legislation or statute, including without limitation, the Bankruptcy Code, shall for greater certainty mean the equivalent section in the applicable piece of legislation to the extent that the context implies reference to such other similar or equivalent legislation as is in effect from time to time in any other applicable jurisdiction, as applicable; furthermore, where any such reference is meant to apply to such other similar or equivalent legislation where such other similar or equivalent legislation has parallel or like concepts, then such references shall import such parallel or like concepts from such other similar or equivalent legislation, as applicable;
- (i) the "winding-up" or "dissolution" of a company includes any equivalent or analogous proceeding under the law of the jurisdiction in which that company is incorporated or carries on business;
- (j) "know your customer requirements" are the checks that a Participant or the Investment Agent requests in order to meet its obligations under applicable money laundering regulations to identify a Person who is (or is to become) its customer;
- (k) references to "assets" includes present and future properties, revenues and rights of every description;
- (l) unless the contrary intention appears, a reference to a "month" or "months" is a reference to a period starting on one day in a calendar month (determined in accordance with the Gregorian calendar) and ending on the numerically corresponding day in the next calendar month or the calendar month in which it is to end, except that:
 - (i) if the numerically corresponding day is not a Business Day, the period will end on the next Business Day in that month (if there is one) or the preceding Business Day (if there is not);
 - (ii) if there is no numerically corresponding day in that month, that period will end on the last Business Day in that month; and

- (m) notwithstanding **paragraph** (l)(i) above, a period which commences on the last Business Day of a month will end on the last Business Day in the next month or the calendar month in which it is to end, as appropriate; and
- (n) a Default (including an Event of Default) is "continuing" if it has not been remedied or waived.

1.3 Rights of Third Parties

Except for the Participants or unless expressly provided to the contrary in this Agreement, a Person who is not a party to this Agreement may not enforce any of its terms.

1.4 Headings

The headings in this Agreement are for convenience only and shall not affect the construction of this Agreement.

2. MURABAHA FACILITY

2.1 Facility

Subject to the terms and conditions of the Finance Documents, in order to finance the purchase of Commodities from the Seller in accordance with the terms of this Agreement, the Investment Agent agrees to make available to AIHL during the Availability Period and on behalf of the Participants a senior secured Dollar-denominated multiple-draw term murabaha facility (the "Facility") in an aggregate amount outstanding at any time not to exceed the Facility Limit at such time and, prior to the Final Order Entry Date, the Interim Facility Limit at such time. No Finance Party is bound to monitor or verify the application of any amount purchased, raised or guaranteed by any Obligor pursuant to this Agreement. Other than with respect to an Existing Purchase Contract that is replaced by a Subsequent Purchase Contract in accordance with the terms hereof, the Facility Limit shall be reduced by the amount of any Cost Price from time to time paid, prepaid, reduced or cancelled under this Agreement, and the amount by which the Facility Limit is so reduced shall no longer be available and shall not be reinstated or reused by AIHL at any time.

2.2 Purchase Contracts

Subject to the terms and conditions of the Finance Documents (including, but not limited to, the limitations set forth in **clauses 2.1** and **4.1** hereof and the satisfaction of each applicable condition precedent set forth in **clause 3**), the Investment Agent will, on behalf of the Participants, purchase Commodities from the Seller at Cost Price and sell those Commodities to AIHL at the Deferred Sale Price on deferred payment terms pursuant to a Purchase Contract. The obligations of AIHL to purchase and pay for the Commodities

pursuant to the preceding sentence shall at all times constitute an allowed Superpriority Claim in the Cases, subject to the Carve-Out.¹

2.3 Maturity Date Extension

- (a) Following the Extension Availability Date, and in any event no later than the date that is thirty days prior to the Termination Date, AIHL may, by written notice to the Investment Agent, request that the Investment Agent extend the Maturity Date for an additional period not to exceed six months from the initial Maturity Date (such extension (if granted), the "Maturity Date Extension," the last date of such additional period, the "Extended Maturity Date"); provided that there shall be no more than one Maturity Date Extension. Such notice shall state (i) that AIHL is requesting that the Maturity Date be extended, (ii) the Extended Maturity Date, which shall be a Business Day, and (iii) the date on which such extension is requested to become effective (the "Requested Extension Date").
- (b) The effectiveness of the Maturity Date Extension shall be subject to the satisfaction of the following conditions precedent (the date (if any) such conditions precedent are satisfied, the "Extension Effective Date"), (i) on or prior to the Requested Extension Date, the Investment Agent shall have notified AIHL that it approves the Maturity Date Extension, which approval shall be given or withheld in the Investment Agent's sole discretion, and (ii) on the Requested Extension Date, the conditions precedent set forth in clause 3.2 shall have been satisfied as of such date, both immediately before and immediately after giving effect to the Maturity Date Extension (with all references in such clause to the purchase of Commodities being deemed references to such Maturity Date Extension) and AIHL shall have delivered to the Investment Agent a certificate in form and substance satisfactory to the Investment Agent, signed by an authorized officer of AIHL, certifying that such conditions precedent have been satisfied.
- (c) On the Extension Effective Date, the Maturity Date shall be extended to the Extended Maturity Date.

3. CONDITIONS PRECEDENT

3.1 Initial Conditions Precedent

AIHL may not deliver the initial Transaction Request or request the Investment Agent to enter into a Purchase Contract unless the Investment Agent has confirmed to AIHL and the Participants that all of the conditions precedent listed in **Schedule 1** have been satisfied, in form and substance, satisfactory to the Investment Agent, at which time, subject to the satisfaction of the conditions precedent contained in **clause 3.2** the Investment Agent will be obligated to purchase the Commodities on behalf of the Participants pursuant to **clause 2.2**.

¹ Commodity purchase and sale mechanics are subject to continued review and revisions.

3.2 Conditions Precedent to each Purchase

The Investment Agent's obligations to purchase Commodities pursuant to **clause 2.2** at any time, the effectiveness of the Maturity Date Extension pursuant to **clause 2.3**, and the effectiveness of the Conditional Increase pursuant to **clause 9.4**, shall also be subject to the satisfaction, or waiver in accordance with the terms hereof, of the following conditions precedent; *provided* that the satisfaction of the conditions precedent contained in **clauses (e)**, **(g)** and **(i)** shall not be required in connection with any Subsequent Purchase Contract (except that the condition precedent in **clause (e)** shall be required in connection with any Purchase Contract effected on a Bringdown Date):

- (a) Entry by the Bankruptcy Court of the Interim Order, on or prior to December ___, 2012, and no order modifying, reversing, staying or vacating the Interim Order shall have been entered.
- (b) For Commodity purchases made on or after January ____, 2013, entry by the Bankruptcy Court of the Final Order, and no order modifying, reversing, staying or vacating the Final Order shall have been entered, or any motion for reconsideration or appeal of the Final Order shall have been timely filed.
- (c) For Commodity purchases made on and after the Final Order Entry Date (or such later date as the Investment Agent may agree in its sole discretion), AIHL shall have delivered to the Investment Agent a certified copy of the special resolution of each LT CayCo amending the liens and restrictions on the transfer of shares in each LT CayCo in form and substance satisfactory to the Investment Agent.
- (d) After giving effect to such Commodities purchase, (i) the Proposed Additional Cost Price, when aggregated with (ii) the aggregate amount of the thenoutstanding Cost Price amounts under all other Purchase Contracts in existence at such time (which shall exclude any Purchase Contract in existence at such time to the extent the Deferred Sale Price of such Purchase Contract is to be paid in full contemporaneously with AIHL and the Investment Agent entering into the new Purchase Contract) (the "Aggregate Cost Price"), shall not exceed (x) prior to the Final Order Entry Date, the Interim Facility Limit at such time and (y) thereafter, the Facility Limit at such time.
- (e) After giving effect to such Commodities purchase, all representations and warranties contained in this Agreement and the other Finance Documents (subject to, after the Effective Date, any Permitted Representation Exceptions), shall be true and correct in all material respects, except for representations qualified by materiality or Material Adverse Effect, in which case such representation and warranty shall be true and correct in all respects, as of the date of such purchase (or such earlier date as may be expressly referenced in any such representation and warranty).
- (f) Before and after giving effect to such Commodities purchase, no Default shall have occurred and be continuing.

- (g) No Material Adverse Effect shall have occurred since November 1, 2012.
- (h) the Debtors shall have utilized any proceeds of the Eurolog IPO in accordance with the SCB Order prior to such Transaction Date.
- (i) No Purchase Contract (other than any Subsequent Purchase Contract) shall be entered into following the date that is 75 days after the Effective Date.

4. PURCHASE CONTRACT LIMITATIONS

4.1 Limitations

AIHL may only issue a Transaction Request if the following limitations, which apply to each Transaction Request and Purchase Contract, are complied with:

- (a) the Transaction Date for each Purchase Contract must be a Business Day falling within the Availability Period;
- (b) the Cost Price payable in respect of such Purchase Contract (the "**Proposed** Additional Cost Price") must be a minimum amount of \$25,000,000 and an integral multiple of ▶[\$_____] in excess thereof;
- (c) the applicable conditions precedent set forth in **clauses 3.2** shall be satisfied on the date of the Transaction Request;
- (d) a Purchase Contract will not be entered into if it would be illegal or unlawful for AIHL or the Investment Agent to do so, or if it would be illegal or unlawful for a Participant to participate in the funding of the Purchase Contract;
- (e) there shall be no more than two Purchase Contracts outstanding at any time during the Availability Period (including during any extension thereof); and
- (f) No Transaction Request (other than a Transaction Request for any Subsequent Purchase Contract) shall be delivered for a Transaction Date that is 75 days after the Effective Date.

4.2 Expiration

The Facility shall expire and be terminated, and the Facility Limit and, prior to the Final Order Entry Date, the Interim Facility Limit, as the case may be, shall be reduced to zero, on the Termination Date.

5. PROCEDURES

5.1 Transaction Request

(a) When AIHL wants to utilize the Facility, it shall give to the Investment Agent a duly completed Transaction Request by no later than 12:00 noon ►[(London

time)] three Business Days before the proposed Transaction Date (or such other time and date as may be agreed between AIHL and the Investment Agent).

(b) Once given, a Transaction Request will be irrevocable.

5.2 Offer

Following the Investment Agent's receipt of a duly completed Transaction Request, after the Investment Agent has purchased the requested Commodities from the Seller on the Transaction Date, the Investment Agent shall by no later than 11.00 a.m. \triangleright [(London time)] on the Transaction Date (or such other time as may be agreed between AIHL and the Investment Agent) offer to sell to AIHL the same Commodities and send the terms of a Purchase Contract by facsimile or other electronic transmission via e-mailed pdf or other similar format in an Offer Letter specifying:

- (a) the Transaction Date;
- (b) the Deferred Payment Date;
- (c) the quantity and type of Commodities to be sold;
- (d) the Cost Price of those Commodities; and
- (e) the Deferred Sale Price, which shall be the aggregate of:
 - (i) the Cost Price; plus
 - (ii) the Profit Amount; plus
 - (iii) the Mandatory Cost (if any); plus
 - (iv) the Increased Costs (if any); plus
 - (v) VAT (if any).

5.3 Acceptance

AIHL shall communicate its acceptance of the Offer Letter by facsimile or other electronic transmission via e-mailed pdf or other similar format (by no later than 12:00 noon > [(London time)]) on the date of the Offer Letter (or such other time as may be agreed between AIHL and the Investment Agent), with the original acceptance to be delivered to the Investment Agent by courier. Non-receipt (for whatever reason) of such original acceptance shall not in any way affect any Purchase Contract.

Upon AIHL communicating acceptance of the Offer Letter:

(a) a Purchase Contract shall be created between the Investment Agent and AIHL incorporating all of the terms and conditions of this Agreement, the relevant Offer Letter and the communication from AIHL accepting the Offer Letter;

- (b) ownership of, and title to, the relevant Commodities shall immediately pass to and be vested in AIHL, together with all rights and obligations relating thereto; and
- (c) risk in all Commodities purchased by AIHL from the Investment Agent pursuant to the relevant Purchase Contract will pass to AIHL once title to such Commodities passes to AIHL.

5.4 Investment Agent Warranties

- (a) The Investment Agent represents and warrants to AIHL that any Commodities sold by the Investment Agent to AIHL in connection with a Purchase Contract will not be subject to any Lien created by it. All commodities sold by Investment Agent to AIHL will be sold with the benefit of the warranties related to the condition or title to the Commodities (if any) granted by the Seller to the Investment Agent under the ▶[DD&Co Ltd] Agreements.
- (b) Save as provided in the first sentence of **clause 5.4(a)**, the Investment Agent shall not be deemed to give any warranty or representation (express or implied) whatsoever in respect of any Purchase Contract, whether arising by law, by statute or otherwise and, without prejudice to the generality of the foregoing, any such warranty or representation is hereby expressly excluded to the full extent permitted by applicable law. AIHL shall be considered to have accepted the Commodities unconditionally and without reservation and shall have no remedy against the Investment Agent in respect of quality, condition, quantity, description or otherwise in respect of any Commodities other than for breach of the Investment Agent's representation contained in the first sentence of **clause 5.4(a)**.

5.5 Audit

The Investment Agent shall, upon request of AIHL, promptly supply to AIHL copies of any documentation, provided by the Seller to the Investment Agent, evidencing the Commodities that are the subject of a Purchase Contract.

5.6 [Reserved]

5.7 No Cancellation

Once a Purchase Contract is created, by the Investment Agent executing the Offer Letter and AIHL's acceptance of the Offer Letter, it shall be irrevocable.

5.8 On-Sale

Provided that a Purchase Contract has been created in accordance with **clause 5.3**, AIHL will sell to ►[Condor Trade Limited], on the same day, the Commodities it has purchased under that Purchase Contract.

6. DEFERRED SALE PRICE AND PAYMENTS

6.1 Deferred Sale Price

- - (i) if included as part of the Deferred Sale Price, pay the Increased Costs Amount component of the Deferred Sale Price within 5 Business Days of the Transaction Date for the account of the Participants that have provided a certificate pursuant to **clause 16.5(d)** on or prior to such fifth Business Day; and
 - (ii) pay the Deferred Sale Price less any Increased Costs Amount paid under **paragraph** (i) of this clause 6.1(a) on its Deferred Payment Date for the account of the Participants,

in each case in immediately available funds to the account the Investment Agent notifies in writing to AIHL for this purpose.

(b) AIHL shall be absolutely and irrevocably required to pay the Deferred Sale Price and all other Obligations and amounts from time to time payable by it under any Finance Document in accordance with the terms hereof and thereof.

6.2 Currency of Account; General Provisions Regarding Payments

- (a) All payments by AIHL relating to Cost Price, Deferred Sale Price, costs, losses, expenses, Taxes and all other Obligations and amounts from time to time payable by it under any Finance Document shall be ▶ [made in the currency in which the relative Cost Price, Deferred Sale Price, costs, losses, expenses, Taxes, other Obligations or amounts were incurred. Any other amount must be] paid in Dollars. All such payments shall be made in immediately available funds and delivered to the Investment Agent not later than ▶ [12:00 noon (London time)] on the date due at the office or account designated by the Investment Agent for the account of the Participants. For the purpose of computing Profit Amount and fees, funds received by the Investment Agent after that time on such due date shall be deemed to have been paid by AIHL on the next succeeding Business Day.
- (b) Unless otherwise specified herein, all payments in respect of the Cost Price of any Purchase Contract shall be accompanied by payment of accrued and unpaid Profit Amount, Mandatory Cost (if any), Increased Costs (if any) and VAT (if any), in each case in respect of the Cost Price for such Purchase Contract being repaid or prepaid.
- (c) Whenever any payment to be made hereunder shall be stated to be due on a day that is not a Business Day, such payment shall be made (i) on the next preceding Business Day, or (ii) if such day would fall after the Termination Date, the Termination Date.

6.3 Accounts as Evidence

The Investment Agent shall maintain in accordance with its usual practice an account which shall, as between the parties, in the absence of manifest error, be conclusive evidence of the amounts from time to time owing to and paid to the Investment Agent and Participants under the Finance Documents.

6.4 Set-Off

▶ [Unless expressly provided otherwise in the Netting Letter], all payments (including the Deferred Sale Price) by AIHL under any Finance Document shall be made in full without any set-off or counterclaim.

6.5 Late Payment

- (a) If any sum (including, without limitation, any late payment charge) which is due and payable by AIHL under or in connection with this Agreement is not paid in full on the due date in accordance with this Agreement (an "unpaid amount"), AIHL undertakes to pay late payment charges (calculated in accordance with clause 6.5(b)) to the Investment Agent on demand for each day that an unpaid amount remains outstanding. The Investment Agent shall pay the amount of any late payment charges received by it:
 - (i) to each Participant to compensate it for any actual costs (not to include any opportunity costs or funding costs) certified to the Investment Agent by that Participant, provided that any such amount shall not exceed such Participant's pro rata share of the late payment amount; and
 - (ii) the balance, on behalf of AIHL, to such charitable foundations as may be selected by AIHL and approved by the Investment Agent.
- (b) The late payment charge in respect of an unpaid amount will accrue on a daily basis on the basis of a year of 360 days and shall be calculated in accordance with the following formula:

(UPA x (Profit Rate + 6.0%)) /360

where:

UPA is the unpaid amount.

7. PRIORITY AND LIENS; ETC.

7.1 Priority and Liens

AIHL hereby covenants, represents and warrants that, upon entry of the Interim Order (and the Final Order, as applicable), the Obligations of each of the Debtors hereunder and under the other Finance Documents:

- (i) pursuant to Section 364(c)(1) of the Bankruptcy Code, shall at all times constitute allowed Superpriority Claims, *provided* that so long as the Existing US\$ Facilities obligations are outstanding, the guarantees of and superpriority claims against WTHL, AEID II, and RailInvest shall be subordinated to the existing guarantees in favor of SCB, and SCB shall have a prior superpriority administrative claim in all proceeds of the EuroLog IPO to the extent provided under the SCB Order,
- (ii) pursuant to Section 364(c)(2) of the Bankruptcy Code, shall be secured by a perfected first-priority Lien on all now owned or after acquired assets of Arcapita Bank, AIHL and AIHL Sub, in each case, that are not otherwise subject to the Existing Security (such first-priority Lien to include (a) AIHL's interests in the WCFs and AIHL's voting right with respect thereto, (b) AIHL Sub's interests in the unencumbered LT Caycos (i.e., all LT Caycos other than WTHL, AEID II, and RailInvest, but only for so long as the Existing US\$ Facilities guaranteed by such entities remains unpaid)) and AIHL's non-syndicated interests in the Syndication Companies; and
- (iii) pursuant to section 364(c)(3) of the Bankruptcy Code, a perfected junior lien on all now owned or after acquired assets of the Debtors that are subject to (x) any valid, perfected and non-avoidable lien in existence on the Petition Date or (y) any valid lien in existence on the Petition Date that is perfected (but not granted) subsequent to the Petition Date pursuant to section 546(b) of the Bankruptcy Code or otherwise comes into existence or is acquired after the Petition Date (including, in each case and for so long as the obligations under the Existing US\$ Facilities remain unpaid, the Existing Security),

subject and subordinate in each case with respect to **subclauses (i)** through **(iii)** above, to the Carve-Out.

Notwithstanding the foregoing, the Liens described above shall not attach to (a) actions for preferences, fraudulent conveyances, and other avoidance power claims under sections 544, 545, 547, 548, 550 and 553 of the Bankruptcy Code or the proceeds thereof or (b) AIHL's Equity Interests in PointPark Properties s.r.o.

The Obligations will also be secured by the Collateral owned by the non-Debtor Obligors as provided in the Security Documents.

7.2 Perfection

Each Debtor acknowledges that, pursuant to the Orders, the Liens granted in favor of the Investment Agent and the Participants in all of the Collateral of such Debtor shall be perfected without the recordation of any financing statements, notices of Lien or other instruments of mortgage, charge or assignment. Each Debtor further agrees that (a) the Investment Agent shall have the rights and remedies set forth in **clause 14.20**, the Security Documents and the Orders in respect of the Collateral of the Debtors and (b) if requested by the Investment Agent, the Debtors shall enter into separate security agreements, pledge agreements, charges and mortgages with respect to such Collateral on terms reasonably satisfactory to the Investment Agent.

7.3 Payment of Obligations

Upon the maturity (whether by acceleration or otherwise) of any of the Obligations, the Participants shall be entitled to immediate payment of such Obligations without further application to or order of the Bankruptcy Court.

7.4 No Discharge; Survival of Claims

Each Debtor agrees that to the extent its Obligations hereunder are not satisfied in full, (a) its Obligations arising hereunder shall not be discharged by the entry of a Confirmation Order (and each Debtor, pursuant to Section 1141(d)(4) of the Bankruptcy Code, hereby waives any such discharge) and (b) the Superpriority Claim granted to the Investment Agent and the Participants pursuant to the Orders and the Liens granted to the Investment Agent pursuant to the Orders, in each case and described in **clause 7.1** shall not be affected in any manner by the entry of a Confirmation Order.

7.5 Conflicts

To the extent of any conflict between the provisions of the Finance Documents and provisions contained in either Order, the provisions of the applicable Order shall govern.

8. TAX

8.1 No Deductions and Grossing-Up

- (a) All payments under the Finance Documents from AIHL or any other Obligor to the Investment Agent or any Participant shall be made without deduction and free from any present or future Tax unless AIHL or such other Obligor is compelled by law to make any such deduction or withholding. In such event, AIHL or such other Obligor shall pay to the appropriate authorities the amount required to be deducted or withheld and shall pay to the Investment Agent or such Participant, as the case may be, such further amounts, as may be necessary, in order that the net amounts received and retained by the Finance Parties, after such deduction or withholding, shall equal the amount which would have been received and retained by the Finance Parties in the absence of such deduction or withholding.
- (b) Within 30 days of any Obligor making either a Tax Deduction or any payment required in connection with that Tax Deduction, AIHL shall deliver to the Investment Agent, for the Finance Party entitled to the payment, evidence reasonably satisfactory to that Finance Party that the Tax Deduction has been made or (as applicable) any appropriate payment paid to the relevant taxing authority.

8.2 Tax Credit

If AIHL or any other Obligor makes a Tax Payment and the relevant Finance Party determines that:

- (a) a Tax Credit is attributable to that Tax Payment; and
- (b) that Finance Party has obtained, utilized and retained that Tax Credit,

the Finance Party must, if no Event of Default shall have occurred and be continuing, promptly upon making such determination, pay an amount to AIHL or such other Obligor which that Finance Party determines will leave it (after that payment and taking into account any out-of-pocket expenses incurred by the Finance Party in connection with obtaining such Tax Credit or making such payment) in the same after-Tax position as it would have been in had the Tax Payment not been required to be made by AIHL; provided that AIHL, upon the request of the Finance Party, agrees to repay the amount paid over to AIHL (plus any penalties, interest or other charges imposed by the relevant taxing authority) to the Finance Party in the event the Finance Party is required to repay such Tax Credit to the relevant taxing authority.

8.3 Indemnity

- (a) If a Finance Party is required to make any payment on account of Tax or otherwise on or in relation to any sum received or receivable under or pursuant to any Finance Document by the Finance Party (including, without limitation, any sum received or receivable under this **clause 8**) or any loss, liability or cost in respect of any such payment is asserted, imposed, levied or assessed directly or indirectly against the Finance Party, AIHL shall, upon demand of the Finance Party, promptly indemnify the Finance Party against such payment or liability, together with any actual penalties and expenses payable or incurred in connection therewith.
- (b) **Paragraph** (a) above shall not apply:
 - (i) with respect to any Tax assessed:
 - (1) under the law of the jurisdiction in which the Finance Party is incorporated or, if different, the jurisdiction (or jurisdictions) in which the Finance Party is treated as resident for tax purposes; or
 - (2) under the law of the jurisdiction in which the Finance Party's Facility Office is located in respect of amounts received or receivable in that jurisdiction,

if that Tax is imposed on or calculated by reference to the net income received or receivable (but not any sum deemed to be received or receivable) by the Finance Party; or

(ii) to the extent a loss, liability or cost is compensated for by an increased payment under **clause 8.1**.

8.4 Value Added Tax

- (a) All amounts set out, or expressed to be payable under a Finance Document by any party to a Finance Party which (in whole or in part) constitute the consideration for value added tax as provided for in the Value Added Tax Act 1994 and any other tax of a similar nature ("VAT") shall be deemed to be exclusive of any VAT which is chargeable on such supply, and accordingly, subject to clause 8.4(c), if VAT is chargeable on any supply made by any Finance Party to any party under a Finance Document, that party shall pay to the Finance Party (in addition to and at the same time as paying the consideration) an amount equal to the amount of the VAT (and such Finance Party shall promptly provide an appropriate VAT invoice to such party).
- (b) If VAT is chargeable on any supply made by any Finance Party (the "Supplier") to any other Finance Party (the "Recipient") under a Finance Document, and any party (the "Relevant Party") is required by the terms of any Finance Document to pay an amount equal to the consideration for such supply to the Supplier (rather than being required to reimburse the Recipient in respect of that consideration), such Party shall also pay to the Supplier (in addition to and at the same time as paying such amount) an amount equal to the amount of such VAT. The Recipient will promptly pay to the Relevant Party an amount equal to any credit or repayment from the relevant tax authority which it reasonably determines relates to the VAT chargeable on that supply; provided that the Relevant Party, upon the request of the Recipient, agrees to repay the amount paid over to the Relevant Party (plus any penalties, interest or other charges imposed by the relevant taxing authority) to the Recipient in the event the Recipient is required to repay such credit to the relevant taxing authority.
- (c) Where a Finance Document requires any party to reimburse a Finance Party for any costs or expenses, that party shall also at the same time pay and indemnify the Finance Party against all VAT incurred by the Finance Party in respect of the costs or expenses to the extent that the Finance Party reasonably determines that neither it nor any other member of the group of which it is a member for VAT purposes is entitled to credit or repayment from the relevant tax authority in respect of the VAT.

8.5 Determinations and Tax Affairs

- (a) In this **clause 8** a reference to "determines" or "determined" means a determination made in the discretion (acting reasonably) of the relevant Finance Party making the determination.
- (b) Subject to **clause 10**, no provision of this Agreement will:
 - (i) interfere with the right of any Finance Party to arrange its affairs (Tax or otherwise) in whatever manner it thinks fit;

- (ii) oblige any Finance Party to investigate or claim any credit, relief, remission or repayment available to it or the extent, order and manner of any claim; or
- (iii) oblige any Finance Party to disclose any information relating to its affairs (Tax or otherwise) or any computations in respect of Tax.

9. PREPAYMENTS; FACILITY LIMIT REDUCTION; INCREASE IN FACILITY LIMIT

9.1 Mandatory Prepayments and Facility Limit Reduction

- 9.1.1 <u>Illegality</u>. If at any time following the execution of the Finance Documents it becomes unlawful in any applicable jurisdiction for a Participant to perform any of its obligations as contemplated by the Finance Documents or any of the obligations of a Participant under a Finance Document cease to be legal, valid and binding, that Participant shall, pursuant to the Investment Agency Agreement, promptly notify the Investment Agent upon becoming aware of that event, the Investment Agent shall in turn notify AIHL accordingly, and:
 - (i) the Facility Commitment of each such Participant under the Investment Agency Agreement will be immediately cancelled;
 - (ii) AIHL shall pay to the Investment Agent on behalf of each such Participant the aggregate amount of such Participant's Contributions and its entitlement to any other outstanding amounts in respect of the Deferred Sale Price on the earlier of (x) the next occurring Deferred Payment Date and (y) the date specified by such Participant in the notice delivered to the Investment Agent as the date by which such Participant is no longer permitted to maintain such Contributions under applicable law (being no earlier than the last day of any applicable grace period permitted by law); and
 - (iii) on the payment date determined in accordance with **clause 9.1.1(ii)**, after giving effect to such payment, the Facility Limit and, prior to the Final Order Entry Date, the Interim Facility Limit, in each case shall be reduced by an amount equal to the Contributions so cancelled.

9.1.2 Special Reductions.

(a) In the event of (i) a cancellation or termination of the Facility, (ii) the acceleration of the Obligations, or (iii) the reduction of the Facility Limit (or, prior to the Final Order Entry Date, the Interim Facility Limit) to zero, AIHL shall, on the date of such cancellation, termination, acceleration or reduction, immediately repay or prepay to the Investment Agent the aggregate amount of the then outstanding Deferred Sale Price in relation to all Purchase Contracts.

- (b) In the event that (i) the Aggregate Cost Price exceeds (ii) the Facility Limit (or, prior to the Final Order Entry Date, the Interim Facility Limit), AIHL shall immediately pay any outstanding amounts in respect of the Cost Price component of the aggregate then outstanding Deferred Sale Price in an amount equal to such excess (which payment shall be applied pro rata to the Cost Price of each Purchase Contract then outstanding).
- (c) On the date that is 120 days after the Effective Date, AIHL shall immediately pay any outstanding amounts in respect of the Cost Price component of the aggregate then outstanding Deferred Sale Price in an aggregate amount equal to the difference (if a positive number) between (i) \$25,000,000 and (ii) the aggregate amount by which the Facility Limit has been reduced during such 120-day period pursuant to clauses 9.1.3, 9.1.4, 9.1.5 and 9.2 (such amount, the "Specified Reduction") (which payment shall be applied pro rata to the Cost Price of each Purchase Contract then outstanding). On such date, after giving effect to such payment, the Facility Limit shall be reduced by an amount equal to the Specified Reduction.
- 9.1.3 Eurolog IPO. The proceeds of the Eurolog IPO shall be applied to pay any outstanding amounts in respect of the Cost Price component of the aggregate then outstanding Deferred Sale Price in relation to all Purchase Contracts in accordance with, and in an aggregate amount equal to the amount required to be applied to the Obligations under, the SCB Order (which payment shall be applied pro rata to the Cost Price of each Purchase Contract then outstanding). On the payment date required under the SCB Order, after giving effect to such payment, the Facility Limit and, prior to the Final Order Entry Date, the Interim Facility Limit, in each case shall be reduced by an amount equal to the amount of such payment.
- Prepayment Events. After receipt by AIHL or any member of the Group of the 9.1.4 Net Cash Proceeds of any Prepayment Event, AIHL shall apply an amount equal to 100% of the Attributable Amount of such Net Cash Proceeds to pay any outstanding amounts in respect of the then outstanding Deferred Sale Price on the next occurring Deferred Payment Date. On such Deferred Payment Date, after giving effect to such payment, the Facility Limit (and, prior to the Final Order Entry Date, the Interim Facility Limit) shall, in each case, be reduced by an amount equal to the Attributable Amount of such Net Cash Proceeds payable on such date hereunder. Notwithstanding the foregoing, so long as no Default shall then exist or would arise therefrom, AIHL shall not be required to make any payments under this clause 9.1.4 with respect to a Prepayment Event under clause (g) of the definition thereof to the extent that AIHL shall have delivered to the Investment Agent a certificate signed by an authorized officer of AIHL on or prior to the date which is three Business Days after the receipt of insurance proceeds or a condemnation award giving rise to such Prepayment Event certifying that such proceeds shall be used to repair or replace the assets so damaged, lost, destroyed or taken no later than 365 days following the date of the receipt of such insurance proceeds or condemnation award (which certificate shall

set forth the estimates of the proceeds to be so expended); *provided* that if all or any portion of such proceeds not required to be applied to the payment of the Deferred Sale Price under this sentence shall not be utilized to repair or replace the assets so damaged, lost, destroyed or taken within such 365-day period, the Attributable Amount of such unused portion of such Net Cash Proceeds shall be applied on the last day of such period as a mandatory payment as provided in the first sentence of this **clause 9.1.4**.

9.1.5 WCFs. After receipt by any WCF of any payment or prepayment of any Financial Indebtedness owing to such WCF, AIHL shall apply an amount equal to 100% of the Attributable Amount of such payment or prepayment amount to pay any outstanding amounts in respect of the then outstanding Deferred Sale Price on the next occurring Deferred Payment Date. On such Deferred Payment Date, after giving effect to such payment, the Facility Limit (and, prior to the Final Order Entry Date, the Interim Facility Limit) shall, in each case, be reduced by an amount equal to the Attributable Amount of such payment or prepayment amount payable on such date hereunder.

In connection with any payment under **clauses 9.1.4** or **9.1.5**, a senior officer of AIHL shall deliver a certificate to the Investment Agent setting forth the Net Cash Proceeds and the Attributable Amount of the Net Cash Proceeds attributable to such payment and the calculation thereof.

In the case of this **clause 9.1** (other than **clause 9.1.2(a)**), after giving effect to any payment made prior to such date, the remaining portion of the Deferred Sale Price will remain due and payable on the next Deferred Payment Date in accordance with the terms of this Agreement and the applicable Purchase Contracts. Any payment made pursuant to this **clause 9.1** shall be paid as specified in **clause 6.2** and applied in accordance with the terms of the Investment Agency Agreement, and (other than prepayments made pursuant to **clause 9.1.1**) the Cost Price component of such payment shall reduce the Facility Commitments of the Participants ratably.

9.2 Voluntary Prepayment and Facility Limit Reduction

- (a) AIHL may, if it gives the Investment Agent not less than 3 Business Days' prior written notice by no later than ▶[12:00 noon (London time)] on the date required, prepay all or part (being a minimum amount of \$10,000,000 and an integral multiple of \$1,000,000) of the then outstanding Deferred Sale Price (each such payment being a "Voluntary Prepayment Amount"). Upon the giving of any such notice, the Voluntary Prepayment Amount specified in such notice shall become irrevocably due and payable on the prepayment date specified therein. Any such voluntary prepayment shall be paid as specified in clause 6.2.
- (b) On the prepayment date specified in such notice, after giving effect to such prepayment, the Facility Limit and, prior to the Final Order Entry Date, the Interim Facility Limit, in each case shall be reduced by an amount equal to the Voluntary Prepayment Amount.

(c) After giving effect to each prepayment made pursuant to this **clause 9.2**, the remaining portion of the Deferred Sale Price will remain due and payable on the next Deferred Payment Date in accordance with the terms of this Agreement and the applicable Purchase Contracts. Any prepayment made pursuant to this **clause 9.2** shall reduce the Facility Commitments of the Participants ratably.

9.3 Prepayment Notice Requirements and Restrictions

- (a) Any notice of cancellation or prepayment given by any party under this **clause 9** shall be irrevocable and, unless a contrary indication appears in this Agreement, shall specify the date or dates upon which the relevant cancellation or prepayment is to be made and the amount of that cancellation or prepayment.
- (b) AIHL shall not prepay the Deferred Sale Price or reduce or cancel all or part of the Facility Limit or, prior to the Final Order Entry Date, the Interim Facility Limit, except at the times and in the manner expressly provided for in this Agreement.

9.4 Increase in Facility Limit

- (a) Following the Increase Availability Date, and in any event no later than the date that is 30 days prior to the Maturity Date, AIHL may, by written notice to the Investment Agent, request a one-time increase to the Facility Limit of \$25,000,000 (such amount, the "Conditional Amount," and such increase, the "Conditional Increase"). Such notice shall state (i) that the Increase Availability Date has occurred, (ii) that AIHL is requesting that the Facility Limit be increased by \$25,000,000 and (iii) the date on which such increase is to become effective (the "Requested Increase Date").
- (b) The effectiveness of the Conditional Increase shall be subject to the satisfaction of the following conditions precedent (the date (if any) such condition precedent is satisfied, the "Increase Effective Date"), on the Requested Increase Date, the conditions precedent set forth in clause 3.2 shall have been satisfied as of such date, both immediately before and immediately after giving effect to the Conditional Increase (with all references in such clause to the purchase of Commodities being deemed references to such Conditional Increase) and AIHL shall have delivered to the Investment Agent a certificate in form and substance satisfactory to the Investment Agent, signed by an authorized officer of AIHL, certifying that such conditions precedent have been satisfied.
- (c) On the Increase Effective Date, the Facility Limit shall be increased by the Conditional Amount.

10. MITIGATION

10.1 Duty to Mitigate

Each Finance Party shall, in consultation with AIHL, take all reasonable steps to mitigate any circumstances which arise and which would result in any amount becoming payable under or pursuant to any of clauses 8.1, 8.3, 9.1.1 (to the extent clause 9.1.1 relates to illegality) and clause 16.5(a) including (but not limited to) transferring its rights and obligations under the Finance Documents to another affiliate or Facility Office. This clause 10 does not in any way limit the obligations of AIHL under the Finance Documents.

10.2 Indemnity

AIHL shall indemnify each Finance Party for all costs and expenses reasonably incurred by that Finance Party as a result of steps taken by it under **clause 10.1**.

10.3 No Prejudice

A Finance Party is not obliged to take any steps under **clause 10.1** if, in the opinion of that Finance Party (acting reasonably), to do so might be prejudicial to it.

11. REPRESENTATIONS AND WARRANTIES

AIHL makes the representations and warranties set out in this **clause 11** to the Investment Agent.

11.1 Status and Authority

Each Obligor:

- (a) is duly organized and validly existing and, except for AIHL (as a result of the Cayman Proceedings), is in good standing under the laws of the jurisdiction of its organization and each other jurisdiction where is it required to be qualified and in good standing and, in the case of the Debtors, subject to Bankruptcy Court approval (which has been obtained), and in the case of AIHL, subject to approval by the Joint Provisional Liquidators in the Cayman Proceedings (which has been obtained), has the power and authority to own its property and assets and to transact the business in which it is engaged;
- (b) subject to (in the case of the Debtors) the entry of the Orders and (in the case of AIHL) approval by the Joint Provisional Liquidators in the Cayman Proceedings, has the power and authority to execute, deliver and carry out the terms and provisions of the Finance Documents to which it is a party;
- (c) subject to (in the case of the Debtors) the entry of the Orders and (in the case of AIHL) approval by the Joint Provisional Liquidators in the Cayman Proceedings, has taken all necessary corporate or other organizational action to authorize the

- execution, delivery and performance of the Finance Documents to which it is a party; and
- (d) has duly executed and delivered each Finance Document to which it is a party, and, subject to (in the case of the Debtors) the entry of the Orders and (in the case of AIHL) approval by the Joint Provisional Liquidators in the Cayman Proceedings, and (with respect to non-Debtor Obligors) to any general principles of law limiting its obligations which are specifically referred to in any legal opinion delivered to the Investment Agent, each such Finance Document constitutes the legal, valid and binding obligation of each Obligor party thereto, enforceable against each such Obligor in accordance with its terms.

11.2 Approvals

Other than the Orders, the Cayman Validation Order and those approvals identified in the Finance Documents, no Governmental Authorization or contractual or other consent or approval is required:

- (a) for the execution, delivery or performance by any Obligor of the Finance Documents to which it is a party or any of the transactions contemplated under the Facility (including the granting and enforcement of the security interests contemplated herein and therein), or, subject to any qualifications set out in any legal opinion in relation to the law of the Cayman Islands to be delivered in connection with the Facility, the enforcement of the Investment Agent's or any Participant's rights thereunder; or
- (b) to make the Finance Documents admissible in evidence in its jurisdiction of organization, other than those which have been obtained or effected and are in full force and effect.

11.3 Non-Conflict

Upon entry of the Orders and the Cayman Validation Order (in each case, with respect to the Debtors), the execution, delivery and performance by each Obligor of the Finance Documents to which it is a party and the performance by each such Obligor of its obligations thereunder shall not:

- (a) violate, breach or cause a default under any agreement to which it is a party or to which it or its assets are bound (other than, in the case of Obligors that are Debtors, agreements entered into prior to the commencement of the Cases if the enforcement of such agreement by the counterparty thereto is stayed);
- (b) violate any law, rule, regulation, order, consent or judgment binding on such Obligor; or
- (c) conflict with its Organizational Documents.

11.4 Liens

No Liens exist on the assets of any Obligor except for Existing Security and Permitted Liens.

11.5 Applicable Laws

Each Obligor has complied in all material respects with applicable laws and regulations including (without limitation) applicable environmental laws and regulations.

11.6 Material Adverse Effect

No event having a Material Adverse Effect has occurred and is continuing.

11.7 Litigation

Other than the Cases and the Cayman Proceedings, no litigation, investigation or proceeding of or before any arbitrator or Governmental Authority is pending or, to the knowledge of any Obligor, threatened (i) relating to the Facility or Finance Documents or (ii) against such Obligor which could reasonably be expected to have a Material Adverse Effect.

11.8 Insurance

Insurance maintained by the Obligors is sufficient and in such amounts as are commercially reasonable and customarily carried by similar companies engaged in similar business. All premiums in respect of such insurance, to the extent due, have been paid. The insurance maintained by the Obligors is in full force and effect in accordance with its terms and complies with the requirements set forth in **clause 12.18**.

11.9 Structure and Organization

- (a) The Structure Charts set forth the equity percentage ownership owned directly or indirectly by each Obligor in its Subsidiaries and their respective Investment Companies (provided that the Structure Charts do not reflect certain non-material changes in shareholdings in such Subsidiaries and Investment Companies occurring since September 30, 2012).
- (b) Except for Permitted Liens, Existing Security and, except with respect to subclause (ii) below, as set forth in the Structure Charts and the Disclosure Schedule, all equity securities of AIHL and its wholly-owned Subsidiaries, and all equity securities directly owned by AIHL and its wholly-owned Subsidiaries in their respective Subsidiaries or Investment Companies, are free and clear of (i) Liens and (ii) shareholder, management or other agreements affecting the voting of such shares or the exercise of other rights with respect thereto, and there are no put/calls, subscriptions, options, warrants, rights or other agreements or commitments with respect to such equity securities, or securities exchangeable or

- convertible into, such equity securities and there are no other classes of capital (including preferred shares) outstanding or authorized for such companies.
- (c) Subject to the Information Exception (with respect to the information delivered to the Investment Agent prior to the date of the Commitment Letter), the KPMG Reports, SOFA Schedules and the Waterfalls (as of the respective dates they were provided to the Investment Agent) together identify, except for Financial Indebtedness owed to the WCFs and intercompany Financial Indebtedness owed by Investment Companies to their direct or indirect equity holders and except as set forth in the Disclosure Schedule, all indebtedness for borrowed money and (x) all guarantees of AIHL (and all intermediate holding companies between AIHL and the Transaction Holdcos), and (y) (excluding guarantees) of AIHL's other Subsidiaries and their respective Investment Companies, in each case as of the Effective Date.

11.10 Collateral

Subject (in the case of the Debtors) to the Orders and the Cayman Validation Order, each of the Finance Documents will be effective to create in favor of the Investment Agent, for the benefit of the Participants, a legal, valid and enforceable security interest in the Collateral described therein and upon the filing of any UCC financing statements and the taking of any other actions or making of filings required for perfection under the laws of the relevant jurisdictions and specified in such Finance Documents, such security interest will constitute perfected and continuing Liens on such Collateral, securing the applicable obligations described in such Finance Documents, subject to any qualifications set out in any legal opinion in relation to the law of the Cayman Islands to be delivered in connection with the Facility, enforceable against the applicable Obligor and all third parties, and having priority over all other Liens on such Collateral to the extent provided (in the case of the Debtors) in the Orders, except Existing Security and Permitted Liens. Other than (in the case of the Debtors) the Orders and the Cayman Validation Order, and subject to any qualifications set out in any legal opinion in relation to the law of the Cayman Islands to be delivered in connection with the Facility, no consent, notice to or filing with any Governmental Authority, corporate or organizational board, director or management or with any other Person or entity is required or advisable in connection with the foreclosure or sale of any Collateral by the Investment Agent.

11.11 Investment Company

No Obligor is an "investment company" as defined in, or subject to regulation under, the Investment Company Act of 1940. No Obligor nor any of its subsidiaries is engaged principally, as one or more of its important activities, in the business of extending credit for the purpose of purchasing any "margin stock" as defined in Regulation U. Neither the transactions contemplated in this Agreement nor the use of the proceeds of any Purchase Contracts will violate the provisions of Regulation T, U or X of the Board.

11.12 Orders

- (a) The Debtors are in compliance in all respects with the Orders and any order entered in connection with the Cayman Proceedings.
- (b) The Orders and each order authorizing the Facility entered in connection with the Cayman Proceedings are in full force and effect and be in full force and effect and have not been stayed, reversed, vacated, rescinded, modified or amended in any respect.
- (c) No trustee or examiner has been appointed with respect to the Obligors or their respective properties.

11.13 Tax Returns

Except as permitted not to be filed or paid under the Bankruptcy Code or the Cayman Proceedings, each Obligor has filed all income tax returns and all other material tax returns, domestic and foreign, required to be filed by any of them and have paid all income and other material taxes payable by them that have become due, other than (i) those not yet delinquent, (ii) those contested in good faith as to which adequate reserves have been provided in accordance with generally accepted accounting principles or (iii) those which could not reasonably be expected to have a Material Adverse Effect.

11.14 Enforceability

Subject to any qualifications set out in any legal opinions to be delivered in connection with the Closing of the Facility, the choice of English law, New York law and Cayman Islands law (as applicable) as the governing law of the Finance Documents will be recognized and enforced in each relevant Obligor's jurisdiction of incorporation.

11.15 Filings

It is not necessary that the Finance Documents be filed, recorded or enrolled with any court or other authority in the jurisdiction of organization of any Obligor or that any stamp, registration or similar tax be paid on or in relation to the Finance Documents or the transactions contemplated by the Finance Documents.

11.16 No Defaults

No Default is continuing or might reasonably be expected to result from the consummation of any of the transactions contemplated by the Finance Documents.

11.17 Waiver of Immunity

No Obligor is entitled to claim for itself or any of its assets immunity from suit, execution, attachment or other legal process in any proceedings taken in its jurisdiction of organization in relation to this Agreement.

11.18 Shari'ah Compliance

No Obligor has relied on any representation by or any written declaration, fatwa, opinion or other documents prepared by, on behalf of, or at the request of, the Investment Agent or any other Finance Party as to the Shari'ah compliance of the transactions contemplated by this Agreement or any other Finance Document and the Obligors have independently made their own assessment as to whether such transactions are compliant with the Shari'ah and no Obligor will claim any dispute on the grounds of Shari'ah compliance of the Finance Documents.

11.19 Proxies

- (a) Each investor in a Syndication Company or PV has, except in respect of the election of directors, appointed AIML as its proxy and attorney-in-fact for the purpose of voting and giving written consents in respect of such investor's shares in such Syndication Company or PV, and no such appointments have been revoked;
- (b) all of the directors of each Syndication Company and PV are AIHL employees, and pursuant to the organizational documents of each Syndication Company and PV, the removal of the required directors requires approval of 66.67% of the shareholders of such Syndication Company or PV or an affirmative vote of the board of directors or such Syndication Company or PV; and
- (c) AIML is a party to administration agreements or management agreements with each Syndication Company and PV, pursuant to which AIML has the sole power and authority to manage and administer the affairs of each Syndication Company and PV (subject to the overriding authority and overall supervision or the Board of Directors of such Syndication Company or PV).

11.20 Information

All written information prepared by any Obligor, or by the advisors of any Obligor on such Obligor's behalf (the "Information") other than financial projections, forecasts, budgets and other forward-looking statements and other information of a general economic or industry nature, which has been or is hereafter made available to Investment Agent or any Participant by any Obligor or on such Obligor's behalf by any of such Obligor's authorized representatives with respect to such Obligor and its Subsidiaries in connection with the Facility or provided to any of them in connection with their consideration of providing the Facility, as and when furnished, taken as a whole, is and will be correct in all material respects and does not and will not contain any untrue statement of a fact or omit to state a fact necessary to make the statements contained therein not misleading in light of the circumstances under which such statements are made; *provided* that this representation, as it relates to Information delivered to the Investment Agent prior to the date of the Commitment Letter, shall be subject to the Information Exception.

11.21 DIP Budget

The Debtors represent and affirm that each DIP Budget was prepared in good faith based on assumptions believed to be reasonable at the time such DIP Budget was delivered to the Investment Agent.

11.22 Anti-Terrorism Laws; Foreign Corrupt Practices Act

- (a) Each Obligor, each of its Subsidiaries, and each of the respective employees, officers, directors, brokers or agents of such Obligor or such Subsidiary, is in compliance with, and does not engage in or conspire to engage in any transaction that (x) evades or avoids, (y) has the purpose of evading or avoiding, or (z) attempts to violate, in each case, the (a) Trading with the Enemy Act (50 U.S.C. § 1 et seq., as amended), and each of the foreign assets control regulations of the United States Department of the Treasury (31 C.F.R., Subtitle B, Chapter V, as amended) and any other enabling legislation or executive order relating thereto, (b) the Currency and Foreign Transactions Reporting Act (also known as the "Bank Secrecy Act", 31 U.S.C. §§ 5311-5330 and 12 U.S.C. §§ 1818(s), 1820(b) and 1951-1959, as amended), (c) the Patriot Act and (d) Executive Order 13224 of September 23, 2001 Blocking Property and Prohibiting Transactions With Persons Who Commit, Threaten to Commit, or Support Terrorism (66 Fed. Reg. 49079 (2001)) (clauses (a) through (d) collectively, the "Anti-Terrorism Laws").
- (b) No Obligor, none of its Subsidiaries, and none of their respective employees, officers, directors, brokers or agents acting or benefiting in any capacity in connection with the selling of Commodities under the Facility or any other transaction under the Finance Documents, is an Embargoed Person or conducts any business or engages in making or receiving any contribution of funds, goods or services to or for the benefit of any Embargoed Person.
- (c) No part of the proceeds generated from selling Commodities under the Facility will be used, directly or indirectly, for any payments to any official or employee of any Governmental Authority, political party, official of a political party, candidate for political office, or anyone else acting in an official capacity, in order to obtain, retain or direct business or obtain any improper advantage, in violation of the United States Foreign Corrupt Practices Act of 1977, as amended (the "FCPA").

12. UNDERTAKINGS

12.1 Duration

The undertakings in this **clause 12** and **clause 13** remain in force from the date of this Agreement for so long as any amount is or may be outstanding under a Finance Document.

12.2 Financial Statements

- (a) Commencing on June 30, 2013, AIHL shall supply to the Investment Agent in sufficient copies for all the Participants:
 - (i) as soon as the same are available (and in any event within 120 days of the end of each of its financial years), its Consolidated Financial Statements;
 - (ii) as soon as the same are available (and in any event within 90 days of the end of the first half-year of each of its financial years), its unaudited consolidated interim balance sheet and the related statement of income for that half-year;
 - (iii) as soon as the same are available (and in any event within 45 days of the end of each quarter of each of its financial years), its unaudited consolidated interim balance sheet and the related statement of income for that quarter;
 - (iv) as soon as the same are available (and in any event within 30 days of the end of each month of each of its financial years), its unaudited consolidated interim balance sheet and the related statement of income for that month;
 - (v) as soon as the same are available (and in any event within one Business Day of filing such document with the Bankruptcy Court), its operating report for that month, and any other financial statements or operating summaries, in each case that are filed with the Bankruptcy Court;
 - (vi) if the Maturity Date is extended pursuant to **clause 2.3**, AIHL shall deliver (x) a report in connection with such extension and (y) after such extension, a quarterly report (to be delivered at the same time as the financial statements in **clause** (iii) above) in each case identifying the Reportable Investment Assets, in each case in line with AIHL's revaluation policy as outlined in the relevant Original Financial Statements; and
 - (vii) if during the period covered by the financial statements delivered under clause (i) through (iii) above there has been a dilutive change in AIHL's ownership of any Investment Company in excess of 5% from the ownership reflected in the Waterfalls delivered to the Investment Agent prior to the Effective Date (as the same may be updated by any waterfall provided pursuant to this clause 12.2(a)(vii)), an updated waterfall for such Investment Company reflecting changes made from the Waterfalls delivered to the Investment Agent prior to the Effective Date or from the last waterfall provided pursuant to this clause 12.2(a)(vii) after giving effect to such change in ownership interest.
- (b) AIHL shall procure that Arcapita Bank shall supply to the Investment Agent in sufficient copies for all the Participants:

- (i) as soon as the same are available (and in any event within 120 days of the end of each of its financial years) Arcapita Bank's audited balance sheet and the related statement of income;
- (ii) as soon as the same are available (and in any event within 90 days of the end of the first half-year of each of its financial years) Arcapita Bank's unaudited interim balance sheet and the related statement of income for that half year; and
- (iii) as soon as the same are available (and in any event within 45 days of the end of each quarter of each of its financial years), Arcapita Bank's unaudited interim balance sheet and the related statement of income for that quarter;

provided that, if such statements are not available, that Arcapita Bank will provide its management accounts for its financial year end, half year end and quarter year end.

12.3 Financial Statement Audit Rights

AIHL shall cause the Obligors to permit the Investment Agent and its advisers reasonable access to their books and records, during normal office hours, for the purposes of verifying compliance with the terms of this Agreement.

12.4 Investment Statements

Promptly upon the Investment Agent's request, AIHL shall supply to the Investment Agent such information or evidence in relation to each Investment Company in which AIHL Sub has a direct or indirect Equity Interest as the Investment Agent may reasonably require for the purposes of conducting periodic credit reviews.

12.5 [Reserved]

12.6 [Reserved]

12.7 Requirements as to Financial Statements

AIHL shall ensure that:

- (a) unless the Central Bank of Bahrain otherwise requires in respect of Arcapita Bank (in which case AIHL shall promptly notify those requirements to the Investment Agent) each set of its financial statements delivered by it pursuant to **clause 12.2** are prepared in accordance with IFRS;
- (b) each set of Consolidated Financial Statements and Financial Statements delivered is prepared on the same basis as was used in the preparation of the relevant Original Financial Statements unless, in relation to any set of financial statements it notifies the Investment Agent that there has been a change (A) in IFRS or the

requirements of the Central Bank of Bahrain in respect of Arcapita Bank (as contemplated in **clause 12.7(a)** above), (B) in the accounting practices or (C) in reference periods and its auditor delivers to the Investment Agent;

- (i) a description of any change necessary for those financial statements to reflect IFRS, accounting practices and reference periods upon which the relevant Original Financial Statements were prepared; and
- (ii) sufficient information, in form and substance as may be reasonably required by the Investment Agent, to enable the Finance Parties to determine whether **clause 13** has been complied with and make an accurate comparison between the financial position indicated in those financial statements and the relevant Original Financial Statements.

Any reference in this Agreement to those financial statements shall be construed as a reference to those financial statements as adjusted to reflect the basis upon which the relevant Original Financial Statements were prepared; and

(c) each set of financial statements delivered by it is certified by a duly authorized officer as giving a true and fair view of its financial condition as at the end of the period to which those financial statements relate and of the results of its operations during such period.

12.8 Information - Miscellaneous

AIHL shall supply to the Investment Agent (in sufficient copies for all the Participants if the Investment Agent so requests):

- (a) promptly upon a Responsible Officer becoming aware of them, except with respect to the Cases, the details of any litigation, arbitration or administrative proceedings which are current, threatened or pending against any member of the Group or any Investment Company, and which might reasonably be expected to have a Material Adverse Effect;
- (b) promptly, such further information regarding the financial condition, business and operations of any member of the Group or any Investment Company which the Investment Agent may reasonably request;
- (c) prompt written notice of any contemplated initial public offering of the Equity Interests of any Obligor or any of their respective Subsidiaries, together with such further information regarding such initial public offering which the Investment Agent may reasonably request;
- (d) promptly, upon a Responsible Officer becoming aware of them, written notice of the occurrence of any event of default under (however described, and the steps being taken to remedy it, if any), or any material amendment to, (i) any Financial Indebtedness of or owing to any Investment Company or any WCF and (ii) any

joint venture agreement to which any member of the Group or any Investment Company is a party;

- (e) prompt written notice of any action by any Person to (a) amend, cancel, modify, withdraw, rescind, revoke or change any proxy or any administration agreement, management agreement or similar agreement described in **clauses 11.19(a)** and **(c)**, or (b) alter the composition of any board of directors of any Syndication Company or PV such that AIHL employees do not constitute 100% of the board of directors of such Syndication Company or PV;
- (f) (i) at least 2 days prior to such filing of the same with the Bankruptcy Court in the Cases or in the Cayman Proceedings, or distribution, copies of all motions, applications, pleadings relating to the disposition of any Collateral or that would impair any Finance Party's rights or remedies under any Finance Document (in each case which must be in form and substance reasonably satisfactory to the Investment Agent) (other than (x) pleadings, motions applications or other filings which would reasonably expected to be immaterial to the Investment Agent and the Participants or (y) emergency pleadings, motions or other filings where, despite such Debtors' best efforts, such 2-day notice is impracticable) and (ii) at the time of such distribution by or on behalf of Arcapita Bank or any of its Subsidiaries to any Committee, the United States Trustee in the Cases or the Joint Provisional Liquidators, a copy of the Reorganization Plan and/or any disclosure statement related to such plan; and
- (g) at any time following a Default, such further information as the Investment Agent may require regarding the financial condition, business and operations of any Investment Company.

The Obligors shall permit the Investment Agent reasonable access to the Obligors' representatives and books and records during regular business hours to monitor financial performance, DIP Budget compliance and the Collateral.

12.9 Notification of Default

AIHL shall notify the Investment Agent of the occurrence of any Default (and the steps being taken to remedy it if any) promptly upon becoming aware of the occurrence and, promptly upon receipt of a written request to that effect from the Investment Agent, AIHL shall confirm to the Investment Agent that no Default is continuing (or if a Default is continuing specifying the Default and steps, if any, being taken to remedy it).

12.10 Authorizations

AIHL shall, and shall cause each Obligor to, obtain, comply with the terms of and do all that is necessary to maintain in full force and effect the Orders, the Cayman Validation Order and all authorizations, approvals, registrations, licenses and consents required in or by the laws and regulations of its jurisdiction of incorporation to enable it lawfully to enter into and perform its obligations under the Finance Documents to which it is a party

or to ensure the legality, validity, enforceability or admissibility in evidence in its jurisdiction of incorporation of such Finance Documents.

12.11 Pari-Passu Ranking

AIHL shall, and shall cause each Obligor to, procure that the obligations of each non-Debtor Obligor under the Finance Documents do and will rank at least pari passu with all its other present and future unsecured and unsubordinated Financial Indebtedness, except for liabilities to creditors whose claims are given preference by the laws of the jurisdiction of its incorporation.

12.12 Use of Proceeds Generated from Selling Commodities

- (a) No part of the proceeds generated from selling the Commodities shall:
 - (i) (x) be used, directly or indirectly, for any payments to any governmental official or employee, political party, official of a political party, candidate for political office, or anyone else acting in an official capacity, in order to obtain, retain or direct business or obtain any improper advantage, in violation of the United States Foreign Corrupt Practices Act of 1977, as amended or (y) be paid to (A) any Embargoed Person, (B) any agency of the government of any Sanctioned Country, (C) any organization controlled by a Sanctioned Country or (D) any Person resident in a Sanctioned Country, to the extent subject to a sanctions program administered by U.S. Department of the Treasury's Office of Foreign Assets Control; or
 - (ii) be used in any manner that would will violate the provisions of Regulation T, U or X of the Board of Governors of the United States Federal Reserve System.
- (b) Subject to **paragraph** (a) above, AIHL shall apply all proceeds generated from selling Commodities under the Facility as follows:
 - (i) first, in payment of fees, costs and expenses incurred by the Investment Agent in the negotiation, preparation and administration of the Finance Documents;
 - (ii) secondly, to satisfy any payment obligations in respect of all profits, fees and other expenses due and payable to the Investment Agent or the Participants under the Facility Documents; and
 - (iii) thereafter, for expenses and working capital, capital expenditures and other general corporate purposes consistent with the DIP Budget, subject to any variance permitted under **clause 13**.
- (c) No portion of the proceeds generated from selling Commodities shall be used (x) to make any Prepetition Payment, except as provided in the SCB Order or

otherwise consented to in writing by the Investment Agent or (y) for any act which has the effect of adversely modifying or compromising the rights and remedies of the Investment Agent or any Participant as set forth herein and in the other Facility Documents, or which results in the occurrence of a Default.

12.13 Financial Indebtedness

(a) Subject to **clause 12.13(b),** AIHL shall not, and shall not permit any member of the Group to incur or suffer or permit to exist any Financial Indebtedness (other than the Obligations) or any other liabilities or contingent liabilities other than as existing on the Effective Date and identified on the SOFA Schedules or the Disclosure Schedules; *provided* that the foregoing prohibition shall not apply to any obligations, costs or expenses arising under sections 327, 328, 330, 503 and 1103 of the Bankruptcy Code, including in connection with professional costs and expenses in the Cases, that are permitted under this Agreement and the Orders and payable in accordance with the DIP Budget.

(b) Clause 12.13(a) shall not apply to:

- ordinary course Inter-Obligor Indebtedness incurred after the Effective Date; *provided* that the intercompany Investments giving rise to such Financial Indebtedness are permitted by, and in compliance with, the DIP Budget, subject to any variance permitted under **clause 13**;
- (ii) intercompany Financial Indebtedness incurred in connection with Arcapita Bank's obligation to fund payments in connection with a sale-leaseback transaction entered into with Lusail Golf Development LLC, a Qatari limited liability company (the "Lusail Obligations"); provided that the obligation of Arcapita Bank to make such payments exists on the date hereof and such payments are in compliance with, the DIP Budget, subject to any variance permitted under clause 13; and
- (iii) Financial Indebtedness in an aggregate principal amount not to exceed \$5,000,000 owing by any non-Obligor member of the Group to any Obligor; *provided* that the intercompany Investments giving rise to such Financial Indebtedness are permitted by, and in compliance with, the DIP Budget, subject to any variance permitted under **clause 13**.

12.14 Negative Pledge

AIHL will not (and AIHL shall ensure that no member of the Group will) without the prior consent of the Investment Agent permit any Liens to subsist, arise or be created or extended over all or any part of its present or future undertakings, assets, rights or revenues to secure or prefer any present or future Financial Indebtedness of AIHL or any other Person other than Liens arising pursuant to the SCB Order and the Existing Security and Permitted Liens.

12.15 Disposals

- (a) AIHL shall not (and AIHL shall ensure that each member of the Group shall not) enter into a single transaction or a series of transactions (whether related or not) and whether voluntary or involuntary to sell, lease, transfer or otherwise dispose of any asset, unless an amount equal to the Attributable Amount of the Net Cash Proceeds from such sale, lease, transfer or disposal shall be paid to the Investment Agent on behalf of the Participants (or deposited in escrow) to the extent required under clause 9.1.4. AIHL agrees that it shall use its commercially reasonable efforts (consistent with its and its Affiliates' fiduciary duties) to ensure that no assets constituting all or substantially all of the assets of, or Equity Interests in, any Investment Company shall be sold, leased, transferred or otherwise disposed of in a transaction (or series of related transactions) without the prior written consent of the Investment Agent if the Net Cash Proceeds received from such sale, lease transfer or disposal would be less than 70% of the KPMG LLP midpoint current valuations for such assets delivered to the Investment Agent prior to the Effective Date.
- (b) Clause 12.15(a) above does not apply to:
 - (i) the Eurolog IPO;
 - (ii) distributions received from the Falcon Escrow Agreement; or
 - (iii) any sale, issuance or distribution of assets secured by the Existing Security, *provided* that the proceeds of such sale, issuance or distribution of assets described in this subsection (vii) are (x) applied to the payment of the Existing US\$ Facilities no later than the effective date of a Reorganization Plan, if then outstanding and (y) pending such application are held in escrow or other arrangements satisfactory to the Investment Agent which ensure that such proceeds are only utilized for the payment of the Existing US\$ Facilities and, after the Existing US\$ Facilities are paid in full, the Obligations in accordance with **clause 9.1.4**.

12.16 No Investments

AIHL will not (and shall ensure that no member of the Group shall) make any advance, loan, extension of credit (by way of guaranty or otherwise) or capital contribution to, or purchase any Equity Interests, bonds, notes, debentures or other debt securities of, or any assets constituting a business unit of, or make any other investment in, any Person (all of the foregoing, "Investments") other than Investments made in the ordinary course of its business and on arm's length terms consistent with past practice; *provided* that the following Investments shall be permitted: (a) Investments made by any member of the Group in any other member of the Group or in Investment Companies; *provided*, in each such case, such Investments are consistent with the DIP Budget, subject to any variance permitted under clause 13 and (b) Investments made in any Exit Plan Subsidiary to meet the minimum capitalization requirements of the jurisdiction of organization of such Exit Plan Subsidiary.

12.17 Know Your Customer Requirements

If:

- (a) the introduction of or any change in (or in the interpretation, administration or application of) any law or regulation made after the date of this Agreement;
- (b) any change in the status of AIHL or the Guarantors after the date of this Agreement; or
- (c) a proposed assignment or transfer by a Participant of any of its rights and obligations under any Finance Document to a party that is not a Participant prior to such assignment or transfer,

obliges the Investment Agent or any Participant (or, in the case of **paragraph (c)** above, any prospective new Participant) to comply with know your customer requirements, antimoney laundering rules and regulations or similar identification procedures, AIHL shall and shall procure that the Guarantors shall promptly upon the request of the Investment Agent or any Participant supply, or procure the supply of, such documentation and other evidence as is reasonably requested by the Investment Agent (for itself or on behalf of any Participant) or any Participant (for itself or, in the case of **paragraph (c)** above, on behalf of any prospective new Participant) in order for the Investment Agent, such Participant or, in the case of **paragraph (c)** above, any prospective new Participant to carry out and be satisfied it has complied with all necessary know your customer requirements, anti-money laundering rules and regulations or other similar checks under all applicable laws and regulations pursuant to this Agreement.

12.18 Insurance

AIHL shall (and AIHL shall ensure that each member of the Group (other than any Specified Non-Guarantor Subsidiary that is subject to bankruptcy or any liquidation proceedings) shall) effect and maintain insurance at its own expense on and in relation to its business and assets of an insurable nature with financially sound and reputable underwriters or insurance companies against those risks and to the extent as is usual and prudent for Shari'ah compliant Persons owning or possessing similar assets and/or carrying on the same or substantially the same business as AIHL and each member of the Group, in each case in such amounts, with such deductibles, covering such risks and otherwise on such terms and conditions as shall be customary for such Persons.

12.19 Taxes

Except as permitted not to be filed or paid under the Bankruptcy Code or the Cayman Proceedings, AIHL shall (and AIHL shall ensure that each member of the Group (other than any Specified Non-Guarantor Subsidiary that is subject to bankruptcy or any liquidation proceedings) shall) pay and discharge all Taxes and governmental charges payable by or assessed upon such Person prior to the date on which the same becomes overdue unless, and only to the extent that, (i) such Taxes and charges are not yet delinquent, (ii) such Taxes and charges shall be contested in good faith by appropriate

proceedings, pending determination of which payment may lawfully be withheld, and there shall be set aside adequate reserves with respect to any such Taxes or charges so contested in accordance with IFRS or (iii) such Taxes and charges could not reasonably be expected to have a Material Adverse Effect.

12.20 [Reserved]

12.21 Compliance with Laws

AIHL shall and shall procure that each member of the Group shall comply in all material respects with all laws, rules and regulations including (without limitation) applicable environmental laws and regulations, in each case applicable to or binding upon such Person or any of its assets or to which such Person or any of its assets are subject.

12.22 Change of Business or Group Structure

- (a) AIHL shall ensure that no substantial change is made to (i) the general nature of the business of AIHL or the Group, so that it continues to operate predominantly in the investment and investment management business, and (ii) the corporate structure of the Group and the Investment Companies (other than (x) the formation of any Exit Plan Subsidiaries and (y) as permitted in accordance with clauses 12.15 and 12.16), in each case from that carried on at the date of this Agreement.
- (b) AIHL shall ensure that no company other than a Guarantor owns (whether wholly or otherwise) any of AIHL's equity ownership interests in the Investment Companies.
- (c) AIHL shall not (and shall procure that no member of the Group shall) enter into any amalgamation, demerger, merger or corporate reconstruction without the prior written approval of the Investment Agent.
- (d) AIHL shall ensure that each long-term equity ownership interest owned by AIHL in an Investment Company is owned indirectly by AIHL Sub through its ownership of an LT Cayco.
- (e) AIHL shall ensure that:
 - (i) each Guarantor (other than Arcapita Bank) shall remain a wholly-owned Subsidiary of Arcapita Bank;
 - (ii) AIHL Sub will remain a wholly-owned Subsidiary of AIHL;
 - (iii) each LT CayCo will remain a wholly-owned Subsidiary of AIHL Sub;
- (f) AIHL shall not, and shall ensure that each Guarantor does not, without the prior consent of the Investment Agent, permit any Lien to subsist, arise or be created or extended over all or any part of the Collateral, including the relevant company's

present or future shares, undertakings, assets, rights or revenues other than Existing Security and Permitted Liens.

- (g) Notwithstanding any other provision of this Agreement, AIHL shall ensure that there is no change in the ownership of any Obligor, any Transaction Holdco or any other Investment Company (other than (x) changes of ownership in Transaction Holdcos and other Investment Companies resulting from transfers of Equity Interests by Persons other than an Obligor, an LT CayCo or a whollyowned Syndication Company or (y) as permitted in accordance with clauses 12.15 and 12.16), in each case without the prior consent of the Investment Agent.
- (h) AIHL will procure that no proxies are granted to it or any other Guarantor or any of their respective Subsidiaries by any LT CayCo (other than pursuant to a Finance Document and (to the extent the same relates to the Existing Security) the Existing US\$ Facilities) in relation to the shares such LT CayCo owns in any Holding Company of an Investment Company or by AIHL Sub in in relation to an LT Cayco.
- (i) AIHL will procure that no Obligor or (except for agreements described in items 6 and 7 of the Disclosure Schedule) Transaction Holdco is or will be bound by any agreement with Arcapita Bank (or any of its Subsidiaries) or any Third Party which permits Arcapita Bank (or any of its Subsidiaries (other than AIML) or Third Party) any management or administrative role in the business of any Obligor or any Transaction Holdco.
- (j) AIHL will procure that no agreement is entered into which would affect the validity or enforceability of all or any part of any istisna agreements, Nominee Declarations or Call Options upon the enforcement of any Lien created under or pursuant to the Finance Documents. AIHL will not enter into any agreement (other than in connection with the Existing Security) under which it is granted rights under any istisna agreement, Nominee Declaration or Call Option which would not pass to the Security Agent on the enforcement of the Cayman Charges.

12.23 Deposits

No member of the Group shall open a deposit account for, or accept any deposits from, any Person (other than another member of the Group) if by reason of opening such account or making such deposit, such Person's claim will by law have priority over that member of the Group's unsecured and unsubordinated creditors, including, for the avoidance of doubt, the Finance Parties; provided that, notwithstanding the foregoing, any member of the Group may open a retainer account with a professional or service provider.

12.24 Dividends

Neither AIHL nor any other Guarantor (other than AIHL Sub, the WCFs and the LT Caycos) will declare or make any dividend payment or other distribution of assets or

properties on account of any shares of any class of their respective Equity Interests or make any payment on account of, or set apart assets for a sinking or other analogous fund for, the purchase, redemption, defeasance, retirement or other acquisition of, any Equity Interests of AIHL or any other Obligor; *provided* that, any such payment or distribution may be made (a) by Subsidiaries of AIHL to AIHL and its Subsidiaries, and (b) by AIHL or by any other Guarantor to Arcapita Bank, so long as no Default has occurred and is continuing and the proceeds thereof are promptly used by Arcapita Bank to pay its operating expenses and any other corporate overhead costs and expenses (including payroll and legal and accounting expenses), in each case in the ordinary course of business and only to the extent permitted by, and in compliance with, the DIP Budget (subject to any variances permitted under clause 13).

12.25 Further Assurances

- (a) AIHL shall, and shall cause each other Obligor to, upon request by the Investment Agent (i) correct any material defect or error that may be discovered in any Finance Document or in the execution, acknowledgment, filing or recordation thereof, and (ii) do, execute, acknowledge, deliver, record, re-record, file, re-file, register and re-register any and all such further acts, documents, deeds, conveyances, assignments, approvals, transfers, certificates, assurances and other instruments as the Investment Agent may reasonably require from time to time in order to (x) facilitate a disposal of the Collateral or any Investment Company if an Event of Default is continuing, (y) carry out more effectively the purposes of the Finance Documents, and (z) perfect and maintain the validity, effectiveness and priority of any Liens intended to be created hereunder, under any other Finance Document or under the Orders
- (b) With respect to any assets otherwise constituting "Collateral" acquired after the Effective Date by any Obligor (other than any property described in **paragraph (c)** below) as to which the Security Agent does not have a perfected Lien, AIHL shall, or shall cause such Obligor to, promptly (i) execute and deliver to the Security Agent such amendments to the applicable Security Documents or such other documents as the Security Agent reasonably deems necessary or advisable to grant to the Security Agent a security interest in such assets and (ii) take all actions reasonably necessary or advisable to grant to the Security Agent a perfected first priority (subject to the Existing Security and Permitted Liens) Lien in such assets, including the filing of Uniform Commercial Code financing statements (or, to the extent they exist, similar documents in foreign jurisdictions) in such jurisdictions as may be required by any Security Document or by law or as may be reasonably requested by the Security Agent.
- (c) With respect to each new direct or indirect wholly-owned Subsidiary of any Obligor created or acquired after the Effective Date that is not an Exit Plan Subsidiary (including, but not limited to, each new LT CayCo and WCF (each, an "Additional Guarantor"), AIHL shall cause such Additional Guarantor to promptly (i) become an Obligor by executing and delivering a joinder to the Guarantee (or, at the Investment Agent's election, an Additional Guarantee in

substantially the same form as the Guarantee), and each such Person shall thereupon become a Guarantor hereunder and thereupon shall have all of the rights, benefits, duties, and obligations in such capacity under the Finance Documents, (ii) execute and deliver to the Investment Agent all of the documents and other instruments listed on Schedule 7, in form and substance satisfactory to the Investment Agent and the Security Agent, as applicable, (iii) execute and deliver to the Security Agent such amendments to the applicable Security Documents as the Security Agent reasonably deems necessary or advisable to grant to the Security Agent a perfected first-priority Lien in the Equity Interests of such Additional Guarantor that is owned by any Obligor, (iv) deliver to the Security Agent the certificates representing such Equity Interests, together with undated stock powers, in blank, executed and delivered by a duly authorized officer of the relevant Obligor, (v) cause such Additional Guarantor (A) to become a party to or enter into the applicable Security Documents and (B) to take such actions reasonably necessary or advisable to grant to the Security Agent a perfected first-priority Lien in the Collateral described in the applicable Security Documents with respect to such Additional Grantor, including the filing of Uniform Commercial Code financing statements (or, to the extent they exist, similar documents in foreign jurisdictions) in such jurisdictions as may be required by the applicable Security Documents or by law or as may be requested by the Security Agent.

- (d) The foregoing paragraphs (a) through (c) shall be subject to any applicable limitations set forth in the Security Documents. Notwithstanding anything to the contrary in this Agreement, if the Security Agent shall reasonably determine that the cost of obtaining a Lien in any assets otherwise required to be subject to a Lien hereunder or under the applicable Security Document is excessive in relation to the value of the Lien afforded thereby, neither the Obligors nor any Additional Guarantor shall have any obligations under clause 12.25(b) or clauses 12.25(c)(iii) through (v) solely with respect to such asset.
- (e) Everything any Obligor and any Additional Guarantor is required to do under this clause 12.25 shall be at such Obligor or such Additional Guarantor's expense. Each Obligor and each Additional Guarantor agrees to pay or reimburse the costs (including in connection with advisers) of the Finance Parties in connection with anything any Obligor or any Additional Guarantor is required to do under this clause 12.25.

12.26 Transactions with Affiliates

AIHL shall not, and will not permit any Obligor or any of their respective Subsidiaries, to enter into or permit to exist any transaction, including any purchase, sale, lease or exchange of property, the rendering of any service or the payment of any management, advisory or similar fees, with any Affiliate (other than another Obligor) unless such transaction is (a) otherwise permitted under this Agreement and (b) in the ordinary course of business and upon fair and reasonable terms no less favorable to the relevant

Group member than it would obtain in a comparable arm's length transaction with a Person that is not an Affiliate.

12.27 Exit Plan Subsidiaries

Without the prior written consent of the Investment Agent, which consent shall be given or withheld in the Investment Agent's sole discretion, no Exit Plan Subsidiary shall engage in any business or shall own any assets (other than assets contributed to such Exit Plan Subsidiary to meet any minimum capitalization requirements of its jurisdiction of organization and other than transactions occurring on the effective date of a Reorganization Plan).

12.28 Chapter 11 Claims

AIHL shall not, and will not permit any Debtor to incur, create, assume, suffer to exist or permit any other Superpriority Claim or Lien on any Collateral which is pari passu with or senior to the claims of the Investment Agent and the Participants granted pursuant to this Agreement, the Finance Documents and/or the Interim Order (or the Final Order, as applicable), except in each case for (i) the Carve-Out, (ii) the Superpriority Claims of SCB to the extent set forth in the SCB Order and (iii) Existing Security.

12.29 Cancellation of Indebtedness

No Obligor shall, or shall permit any of its Subsidiaries to, cancel any claim or debt owing to it, except for reasonable consideration negotiated on an arm's length basis and in the ordinary course of its business consistent with past practices.

12.30 No Impairment of Intercompany Transfers

No Obligor shall, or shall permit any of its Subsidiaries to, directly or indirectly enter into or become bound by any effective agreement, instrument, indenture or other obligation (other than this Agreement, the other Finance Documents and the SCB Order) that could directly or indirectly restrict, prohibit or require the consent of any Person with respect to the payment of dividends or distributions or the making or repayment of intercompany loans by a Subsidiary of any Obligor to any Obligor or between Obligors.

12.31 No Speculative Transactions

No Obligor shall engage in any transaction involving commodity options, futures contracts or similar transactions, except solely to hedge against fluctuations in the prices of commodities owned or purchased by it provided any such transaction is consistent with such Obligors' hedging policies existing as of the date hereof.

12.32 Changes Relating to Other Indebtedness and Material Contracts

No Obligor shall change, modify, amend, waive or consent to variation of the terms of (a) any document or agreement relating to any prepetition Financial Indebtedness (other than amendments to the Existing US\$ Facilities, solely to the extent such amendments

are made to conform the Existing US\$ Facilities to the terms of the SCB Order), (b) any shareholder or other related documents or agreement relating to the relationship described in items 6 and 7 of the Disclosure Schedule or relating to any similar co-investment arrangements, (c) any administration agreement, management agreement or similar agreement to which AIML is a party, (d) any credit, loan, finance or other funding agreement to which a WCF is a party (other than amendments, changes or modifications that only renew or extend the maturity date of such agreements), in the case of clauses (b) through (d) above, if the effect of such change, modification, amendment, waiver or consent could be adverse to the interests of any Finance Party, without in each case obtaining the prior written consent of the Investment Agent. Without limiting the foregoing, no Obligor shall cause or permit any change, modification, amendment, supplement to, or waiver of, any of its rights under any organizational or constitutional document of any Obligor, any Syndication Company or any Transaction Holdco, without in each case obtaining the prior written consent of the Investment Agent.

12.33 Repayment of Indebtedness

Except pursuant to (a) a confirmed Reorganization Plan or (b) as expressly required under the SCB Order, no Debtor shall, without the express prior written consent of the Investment Agent, make any payment or transfer with respect to any Lien or Financial Indebtedness incurred or arising prior to the Petition Date, whether by way of "adequate protection" under the Bankruptcy Code or otherwise, without the prior written consent of the Investment Agent.

12.34 Post-Closing Matters

No later than 30 days following the Effective Date (or such longer period as the Investment Agent may agree in its sole discretion), for each deposit account and securities account owned by any Obligor, such Obligor shall execute and deliver to the Investment Agent a control agreement in form and substance reasonably satisfactory to the Investment Agent, pursuant to which (a) the depositary bank or securities intermediary, as applicable, shall acknowledge the Lien of the Security Agent granted under the applicable Security Documents and (b) the depositary bank or securities intermediary, as applicable, shall agree to comply with instructions from the Investment Agent directing the disposition of funds from time to time credited to such deposit account, or entitlement orders directing transfer or redemption of financial assets carried in such securities account, without further consent of the Obligors.

13. BUDGET COVENANT

Funding & Expenses" and (b) "Net Disbursements" (each as defined below) (collectively, the "**Tested Budget Items**") for the four-week period ending on January 5, 2013 and for any successive four-week period thereafter (in each case calculated in a manner consistent with the DIP Budget) to exceed the budgeted amounts for "Deal Funding & Expenses" and "Net Disbursements", respectively,

during such four-week period set forth in the DIP Budget, by an amount in excess of 10%; *provided* that:

- (x) in the case of "Net Disbursements", amounts not utilized during any four-week period may be carried forward to the next four-week period on a last-out basis (and to the extent not utilized in the subsequent four-week period, such carried forward amount shall expire); and
- (y) in the case of the "Deal Funding & Expenses" line-item, the amount of such line-item contained in the DIP Budget for such four-week period and not utilized during such period may be carried forward to subsequent periods.

As used herein, (a) "Deal Funding & Expenses" means expenses calculated in a manner consistent with the "Total Deal Funding & Expenses" line-item in the DIP Budget and (b) "Net Disbursements" means the difference between (i) total expenses calculated in a manner consistent with the "Total Disbursements" line-item in the DIP Budget and (ii) Deal Funding & Expenses. Any reference to a DIP Budget line-item contained in this **clause 13** shall be a reference to the line-item as set forth in the consolidated tab of the DIP Budget.

13.2 No later than five Business Days after the last day of the four-week budget period ending on January 5, 2013 and each successive four-week period thereafter (and at such other times as the Investment Agent may reasonably request), AIHL shall deliver to the Investment Agent a variance report, reasonably satisfactory to the Investment Agent, showing on a line-item basis the percentage and dollar variance of actual cash disbursements and cash receipts for each Tested Budget Item (and its components) for the prior four-week period and for the period from December 9, 2012 to the last day of such four-week period from the amounts set forth for each such period in the DIP Budget, and a narrative analysis of each material variance for the prior four-week period.

14. DEFAULT AND REMEDIES

Each of the events set out in **clauses 14.1** through **14.16** is an Event of Default.

14.1 Non-Payment

AIHL or any of the Guarantors fails to pay when due, whether at scheduled maturity, by mandatory prepayment, by notice of voluntary prepayment, by acceleration, demand or otherwise, any amount, whether principal of or premium or profit on or any other amount payable in respect of the Obligations owing under the Finance Documents at the time, in the currency and in the manner specified in this Agreement unless (in the case of payments other than for the Deferred Purchase Price) payment is made within five Business Days after its due date.

14.2 Breach of Other Obligations

- (a) AIHL fails to perform or comply with any term, covenant, agreement or condition contained in clauses 9, 12.2 12.7 12.9 through 12.34 (inclusive), 13 and 18.6).
- (b) AIHL or any of the Guarantors fail to perform or comply with any term, covenant, agreement or condition contained herein or in any other Finance Document (other than those referred to in **clauses 14.1** and **14.2(a)**), and such failure shall not have been remedied or waived in accordance with the terms hereof within ten days after such failure.

14.3 Misrepresentation

Any representation, warranty, certification or other statement made or deemed to be made by any Obligor in any Finance Document or in any notice or other document, certificate or statement at any time delivered by any Obligor or any of its Subsidiaries pursuant hereto or thereto or in connection herewith or therewith is or proves to have been false, incorrect or misleading in any material respect when made or deemed to be made, except for representations qualified by materiality or Material Adverse Effect, in which case such representation, warranty, certification or other statement is or proves to have been false, incorrect or misleading in any respect when made or deemed to be made.

14.4 Cross-Default

- (a) Any Financial Indebtedness (other than the Obligations) (or, in the case of clause (iii) below, any commitment in respect of Financial Indebtedness) of any member of the Group (other than Financial Indebtedness of Specified Non-Guarantor Subsidiaries and, with respect to the Debtors, pre-petition Financial Indebtedness, the payment of which is subject to an effective stay in the Cases) or the Lusail Obligations:
 - (i) is not paid when due nor within any originally applicable grace period;
 - (ii) is or can be declared to be, or otherwise becomes, due and payable prior to its specified maturity as a result of an event of default (however described), early amortization event, termination event or other similar event; or
 - (iii) is accelerated, cancelled, suspended or terminated by a creditor of any member of the Group as a result of a default or an event of default (however described), early amortization event, termination event or other similar event;

provided that it shall not constitute an Event of Default under this **clause 14.4(a)** if the aggregate amount (or its equivalent in Dollars) of such Financial Indebtedness (or in the case of (iii) above, any commitment in respect of Financial Indebtedness) is less than \$10,000,000 or its equivalent in any other currencies.

(b) Any creditor of AIHL or any other member of the Group that is not a Specified Non-Guarantor Subsidiary (other than, with respect to the Debtors, any creditor

under pre-petition Financial Indebtedness, the payment of which is subject to an effective stay in the Cases) becomes entitled to declare any Financial Indebtedness (including the Lusail Obligations) of any member of the Group due and payable prior to its specified maturity as a result of a default or an event of default (however described), early amortization event, termination event or other similar event; *provided* that it shall not constitute an Event of Default under this **clause 14.4(b)** if the aggregate amount (or its equivalent in Dollars) of such Financial Indebtedness is less than \$10,000,000 or its equivalent in any other currencies.

(c) Any Debtor shall default on any of its obligations under the SCB Order, or, except with respect to matters not constituting an Event of Default under **clause 14.10** hereunder, SCB shall be permitted to enforce any of its rights under either Existing US\$ Facilities other than with respect to the adequate protection granted to SCB under the SCB Order.

14.5 Insolvency

- (a) Any member of the Group (other than a Debtor or any Immaterial Subsidiary) is unable or admits inability to pay its debts as they fall due, suspends making payments on any of its debts or, by reason of actual or anticipated financial difficulties, commences negotiations with one or more of its creditors with a view to rescheduling any of its indebtedness.
- (b) A moratorium is declared in respect of any indebtedness of any member of the Group (other than a Debtor or any Immaterial Subsidiary).

14.6 Insolvency Proceedings

Any corporate or other organizational action, legal proceedings or other procedure, action or step is taken in relation to:

- (a) the suspension of payments, a moratorium of any indebtedness, winding-up, dissolution, bankruptcy, insolvency, adjustment, dissolution, administration, reorganization (by way of voluntary arrangement, scheme of arrangement or otherwise), composition or other relief with respect to indebtedness, in each case of any member of the Group, other than any legal proceedings which are frivolous or vexatious or are discharged, stayed or dismissed within 30 days of commencement of such legal proceedings;
- (b) a composition, assignment or arrangement with or for the benefit of any creditor of any member of the Group;
- (c) the appointment of a liquidator, receiver, administrator, administrative receiver, trustee, custodian, conservator, compulsory manager or other similar officer or official in respect of any member of the Group or for all or any of its assets;
- (d) the issuance of a warrant of attachment, execution, distraint or similar process or action, or the enforcement of any Lien (other than a Lien granted pursuant to a

- Security Document), in each case against, over or in respect of any assets of any member of the Group;
- (e) a winding up order in respect of AIHL is made by the Courts of the Cayman Islands pursuant to Part V of the Companies Law (2012 revision) of the Cayman Islands or any subsequent revision or enactment thereof;
- (f) the Courts of the Cayman Islands shall grant leave for the commencement or continuance of any suit, action or other proceeding against AIHL, including any criminal proceedings,

or any analogous procedure or step is taken in any jurisdiction; *provided* that, (i) except with respect to **clauses** (e) and (f) above, this **clause 14.6** shall not apply to any Debtor solely with respect to the Cases and the Cayman Proceedings, in each case commenced prior to the Effective Date and (ii) this **clause 14.6** shall not apply to any Immaterial Subsidiary.

14.7 Creditors' Process

Any expropriation, attachment, sequestration, distress or execution affects any asset or assets of any member of the Group (other than a Debtor or an Immaterial Subsidiary).

14.8 Dismissal or Conversion

(i) Any of the Cases of the Debtors shall be dismissed or converted to a case under Chapter 7 of the Bankruptcy Code, (ii) any Debtor shall file a motion or other pleading seeking (x) the dismissal of any Case of any Debtor under Section 1112 of the Bankruptcy Code or otherwise or (y) the conversion of any Case of any Debtor to a case under Chapter 7 of the Bankruptcy Code, or (iii) a trustee (or comparable Person) under Chapter 7 or Chapter 11 of the Bankruptcy Code or a responsible officer or an examiner with enlarged powers relating to the operation of the business (powers beyond those set forth in Section 1106(a)(3) and (4) of the Bankruptcy Code) under Section 1106(b) of the Bankruptcy Code shall be appointed in any of the Cases of the Debtors.

14.9 Superpriority Claims

An order of the Bankruptcy Court shall be entered granting any Superpriority Claim or other claim or administrative expense (other than the Carve-Out and the Superpriority Claims of SCB granted in the SCB Order) in any of the Cases of the Debtors that is pari passu with or senior to the claims or Liens of the Investment Agent and the Participants against AIHL or any other Obligor hereunder or under any of the other Finance Documents, or any Debtor takes any action seeking or supporting the grant of any such claim or Lien, except as expressly permitted hereunder; or

14.10 Relief from the Automatic Stay

The Bankruptcy Court shall enter an order or orders granting relief from the automatic stay applicable under Section 362 of the Bankruptcy Code to the holder or holders of any Lien to permit foreclosure (or the granting of a deed in lieu of foreclosure or the like) on

any assets of any of the Debtors which have a value in excess of \$100,000,000 in the aggregate.

14.11 The Orders

- (a) The Interim Order Entry Date shall not have occurred by the date that is 2 days following the hearing of the Bankruptcy Court relating to the Bankruptcy Court's approval of this Agreement.
- (b) The Final Order Entry Date shall not have occurred by the earlier of (i) the date of the expiration of the Interim Order and (ii) the date that is 30 days following entry of the Interim Order.
- (c) Any Order shall cease to be in full force and effect or an order of the Bankruptcy Court shall be entered reversing, staying, vacating or (except as otherwise agreed to in writing by the Investment Agent in its sole discretion) otherwise amending, supplementing or modifying the Interim Order, the Final Order or any of the Finance Documents.
- (d) Any Debtor shall violate or fail to comply with any material term, provision or condition contained in (i) the Orders or the Cayman Validation Order or (ii) the SCB Order or any other order of the Bankruptcy Court relating to the Existing US\$ Facilities.

14.12 Government Intervention

By or under any Governmental Authorization:

- (a) the management of AIHL or any member of the Group (other than any Specified Non-Guarantor Subsidiary) is wholly or partially displaced or the authority of AIHL or any member of the Group (other than any Specified Non-Guarantor Subsidiary) in the conduct of its business is wholly or partially curtailed; or
- (b) any of the issued shares of AIHL or any member of the Group (other than any Specified Non-Guarantor Subsidiary) or all or a substantial part of their respective revenues or assets is seized, nationalized, expropriated or compulsorily acquired or an order is made to that effect.

14.13 Finance Documents; Repudiation

- (a) Any Finance Document shall cease, for any reason, to be in full force and effect or any material provision thereof ceases to be legal, valid, binding or enforceable in accordance with its terms against any Obligor that is a party thereto.
- (b) Any Lien purported to be created by the Orders or any Security Documents shall cease to be, or shall be asserted by any Obligor not to be, a valid and perfected Lien on any material portion of the Collateral, enforceable and of the same effect

- and of at least the priority purported to be created thereby other than by reason of the release thereof in accordance with the terms hereof or thereof.
- (c) Any Obligor denies that it has any or further liability or obligation under any Finance Document, or asserts, purports, does or causes to be done any act or thing evidencing an intention to repudiate, revoke, terminate or rescind any provision of any Finance Document, any guarantee contained in any Guarantee, or any Lien granted under the Security Documents or the Orders.

14.14 Illegality

At any time it is or becomes unlawful for any Obligor to perform or comply with any or all of such Obligor's material obligations under, or any term, covenant, agreement or condition contained in, any Finance Document, or any of such Obligor's material obligations under any Finance Document are not or cease to be, legal, valid and binding under the laws of any Governmental Authority.

14.15 Other Bankruptcy Items

The occurrence of any of the following in the Cases:

- (a) The entry of an order in any Case avoiding or requiring repayment of any portion of the payments made on account of the Obligations owing under this Agreement or the other Finance Documents;
- (b) (i) the filing of any Reorganization Plan or disclosure statement attendant thereto, or any direct or indirect amendment to such plan or disclosure statement, by a Debtor or any other Person or (ii) the entry of an order in any of the Chapter 11 Cases confirming a Reorganization Plan, in each case that either (x) is not acceptable to the Investment Agent in its sole discretion or (y) unless the Facility is converted to an exit facility in accordance with and pursuant to the terms and conditions set forth in the Commitment Letter, does not contain a provision for termination of all obligations of the Investment Agent and the Participants to provide the Facility and repayment in full in cash of all of the Obligations under this Agreement and the other Finance Documents on or before the effective date of such Reorganization Plan;
- (c) the sale without the Investment Agent's written consent of all or substantially all of the assets of the Debtors either through a sale under section 363 of the Bankruptcy Code, through a confirmed Reorganization Plan or otherwise that does not provide for payment in full in cash of the Obligations and the termination of all obligations of the Investment Agent and the Participants to provide the Facility; or
- (d) the allowance of any claim or claims under Section 506(c) of the Bankruptcy Code or otherwise against the Investment Agent, any Participant or any of the Collateral.

14.16 Material Adverse Effect

Any event or series of events occur, individually or in the aggregate, which may have or be reasonably likely to have a Material Adverse Effect.

14.17 Change of Control

A Change of Control shall occur.

14.18 Cessation and Change of Business

A substantial change shall occur or be made to the general nature of the business of AIHL and each member of the Group, individually and in the aggregate, from that carried on as of the Effective Date which may have or be reasonably likely to have a Material Adverse Effect.

14.19 Judgments; Etc.

- (a) Any judgment or judgments (other than, with respect to the Debtors, pre-petition judgments that are subject to an effective stay of enforcement) for the payment of money in excess of \$10,000,000 in the aggregate at any time are outstanding against one or more of the Obligors (which judgments are not covered by insurance policies as to which liability has been accepted by the insurance carrier), and the same are not, within thirty days after the entry thereof, discharged or execution thereof stayed or bonded pending appeal, or such judgments are not discharged prior to the expiration of any such stay.
- (b) Any assets of any Obligor with a fair market value of \$10,000,000 or more are attached, seized, levied upon or subjected to a writ or distress warrant, or come within the possession of any receiver, trustee, custodian or assignee for the benefit of creditors of any Obligor and such condition continues for thirty days or more.

14.20 Remedies

If an Event of Default has occurred and is continuing, the Investment Agent may, without further order of or application, motion or notice to, or hearing before or order from the Bankruptcy Court, and notwithstanding the provisions of Section 362 of the Bankruptcy Code:

- (a) cancel the Facility, whereupon the Facility Limit shall be reduced to zero; and/or
- (b) require AIHL immediately to pay the Deferred Sale Price in respect of each Purchase Contract and all Obligations and other amounts payable under any Finance Document, whereupon they shall become immediately due and payable.
- (c) Upon seven days' prior written notice given to AIHL, counsel to AIHL, counsel to the Committee and the United States Trustee after the occurrence of an Event of Default, and if an Event of Default has occurred and is continuing (such seven

day period herein referred to as the "Remedies Notice Period"), the Investment Agent may, notwithstanding the provisions of Section 362 of the Bankruptcy Code, without application, motion or notice to, hearing before, or order from, the Bankruptcy Court, take any or all of the following actions, at the same or different times, but subject to the restrictions set forth in the Interim Order (or Final Order, when applicable): (i) require that any or all of the Obligors sell or otherwise dispose of any or all of the Collateral on terms and conditions acceptable to the Investment Agent pursuant to Sections 363, 365 and other applicable provisions of the Bankruptcy Code and with respect to designation rights for assumption and rejection of leases and executor contracts, direct any Obligor to assume and assign any lease or executory contract included in the Collateral to Investment Agent's designees in accordance with and subject to Section 365 of the Bankruptcy Code), (ii) enter onto the premises of any Obligor in connection with an orderly liquidation or other disposition of the Collateral, or (iii) exercise any rights and remedies provided to the Investment Agent or any Participant under the Finance Documents or at law or equity, including all remedies provided under the Bankruptcy Code; provided, however, that during the Remedies Notice Period, the Obligors and the Committee shall be entitled to an emergency hearing before the Bankruptcy Court for the sole purpose of contesting the occurrence and/or continuance of an Event of Default. Unless the Bankruptcy Court determines during the Remedies Notice Period that an Event of Default has not occurred and/or is not continuing, pursuant to the Interim Order (or the Final Order, when applicable) the automatic stay of Section 362 of the Bankruptcy Code shall be modified and vacated to permit the Investment Agent and the Participants to exercise their remedies under this Agreement and the other Facility Documents, without further notice, application or motion to, hearing before, or order from, the Bankruptcy Court. Subject to the Carve-Out, during the Remedies Notice Period, the Obligors shall not use cash collateral (other than to pay payroll and other expenses critical to keep the business of the Obligors operating in accordance with the DIP Budget). Upon the occurrence of an Event of Default and the exercise by the Investment Agent or the Participants of their rights and remedies under this Agreement and the other Facility Documents, each of the Obligors shall assist the Investment Agent and the Participants in effecting a sale or other disposition of the Collateral upon such terms as are acceptable to the Investment Agent.

15. FEES AND EXPENSES

15.1 Fees and Expenses

- (a) AIHL shall promptly on demand pay the Investment Agent and the Arranger the amount of all costs and expenses (including legal fees) reasonably incurred by any of them in connection with the negotiation, preparation, printing, execution and syndication of:
 - (i) this Agreement and any other documents referred to in this Agreement; and

- (ii) any other Finance Documents executed after the date of this Agreement.
- (b) If an Obligor requests an amendment, waiver or consent AIHL shall, within three Business Days of demand, reimburse the Investment Agent for the amount of all costs and expenses (including legal fees) reasonably incurred by the Investment Agent in responding to, evaluating, negotiating or complying with that request or requirement.
- (c) AIHL shall, within three Business Days of demand, pay to each Finance Party the amount of all costs and expenses (including legal fees) incurred by that Finance Party in connection with the enforcement of, or the preservation of any rights under, any Finance Document.

15.2 Fee Letter

AIHL shall pay the fees in the amounts and at the times agreed in each Fee Letter.

16. INDEMNITIES

16.1 Indemnity

- (a) AIHL shall indemnify each Finance Party and their respective directors, officers, employees and agents (each, an "**Indemnified Party**") on demand against any actual costs, loss, liability or expense which the Indemnified Party has sustained or incurred as a consequence of:
 - (i) the failure of AIHL to make payment on the due date of any sum due to the Investment Agent under any Finance Document;
 - (ii) the failure of AIHL to accept the Investment Agent's offer to sell the Commodities on the date offered, whether or not such acceptance is required under clause 5.3 or sell the Commodities pursuant to clause 5.8 of this Agreement;
 - (iii) the occurrence of any Event of Default;
 - (iv) any actions, claims, proceedings, liabilities, losses, damages, penalties, judgments, suits, costs and expenses that may be imposed on, incurred by, asserted of or claimed by any Person and howsoever arising out of this Agreement, any of the other Finance Documents or any action taken or omitted by any of them (including, but not limited to, the sale, delivery, non-delivery, handling, storage, use, possession, seizure, forfeiture of, or in relation to, the Commodities), other than, solely with respect to this clause 16.1(a)(iv), any actions, claims, proceedings, liabilities, losses, damages, penalties, judgments, suits, costs and expenses arising from the ownership of the Commodities by any of the Indemnified Parties; or
 - (v) any prepayment not being made in accordance with **clause 9**;

provided that such indemnity shall not, as to any Indemnified Party, be available to the extent that such losses, damages, penalties, costs and expenses are determined by a final, non-appealable judgment of a court of competent jurisdiction to have resulted solely from the gross negligence or willful misconduct of such Indemnified Party. Subject to clause 6.5, AIHL shall make payment of any indemnity in favor of an Indemnified Party provided for in this clause 16.1(a) within 10 days of demand by such Indemnified Party.

- (b) AIHL shall promptly indemnify the Investment Agent against any reasonable cost, loss or liability incurred by the Investment Agent as a result of:
 - (i) investigating any event which it reasonably believes is a Default; or
 - (ii) acting or relying on any notice, request or instruction which it reasonably believes to be genuine, correct and appropriately authorized;

provided that such indemnity shall not, as to any Indemnified Party, be available to the extent that such costs, losses or liabilities are determined by a final, non-appealable judgment of a court of competent jurisdiction to have resulted solely from the gross negligence or willful misconduct of such Indemnified Party.

16.2 Documentary Taxes Indemnity

All stamp, documentary, registration or other like duties or Taxes, including any penalties, additions, fines, surcharges or late payment charges relating to those duties and Taxes, which are imposed or chargeable on or in connection with any Finance Document shall be paid by AIHL. The Investment Agent shall be entitled but not obliged to pay any such duties or Taxes (whether or not they are its primary responsibility). AIHL shall on demand indemnify the Finance Parties from and against those duties and Taxes and against any costs and expenses incurred by the Finance Parties in discharging them.

16.3 Currency Indemnity

(a) Any payment made to or for the account of or received by the Finance Parties in respect of any moneys or liabilities due, arising or incurred by AIHL to a Finance Party in a currency (the "Currency of Payment") other than the currency in which the payment should have been made (the "Currency of Obligation") in whatever circumstances (including as a result of a judgment against AIHL) and for whatever reason shall constitute a discharge by AIHL only to the extent of the Currency of Obligation amount which the Finance Party is able on the date of receipt of such payment (or if such date of receipt is not a Business Day, on the next succeeding Business Day) to purchase with the Currency of Payment amount at its Spot Rate of Exchange (as conclusively determined by the Investment Agent). If the amount of the Currency of Obligation which the Finance Party is so able to purchase falls short of the amount originally due to the Investment Agent, then AIHL shall immediately on demand indemnify the Finance Party against any loss or cost arising as a result of that shortfall by paying to the

- Finance Party that amount in the Currency of Obligation certified by the Finance Party as necessary so to indemnify it.
- (b) AIHL waives any right it may have in any jurisdiction to pay any amount under the Finance Documents in a currency other than that in which it is expressed to be payable.

16.4 Communications Indemnity

- (a) AIHL shall promptly indemnify the Investment Agent against any cost, loss or liability incurred by the Investment Agent (acting in good faith) as a result of acting or relying on any notice, request or instruction which it reasonably believes to be genuine, correct and appropriately authorized; *provided* that such indemnity shall not be available to the extent that such costs, losses or liabilities are determined by a final, non-appealable judgment of a court of competent jurisdiction to have resulted solely from the gross negligence or willful misconduct of the Investment Agent.
- (b) AIHL verifies that each Person AIHL has identified to the Investment Agent as authorized representatives are duly authorized to give or send instructions and other communications by telephone, facsimile transmission or letter.

16.5 Increased Costs

- (a) Subject to **clause 16.5(e)**, AIHL shall, within 3 Business Days of a demand by the Investment Agent, at any time after the Transaction Date, pay for the account of a Finance Party the amount of any Increased Costs incurred after the Transaction Date by that Finance Party or any of its affiliates.
- (b) In this Agreement "Increased Costs" means:
 - (i) a reduction in the rate of return from the Facility or on a Finance Party's (or its affiliate's) overall capital;
 - (ii) an additional or increased cost or
 - (iii) a reduction of any amount due and payable under any Finance Document,

which is incurred or suffered by a Finance Party or any of its affiliates as a result of (i) the introduction of or any change in (or in the interpretation, administration or application of) any law or regulation or (ii) compliance with any law or regulation made after the date of this Agreement to the extent that it is attributable to that Finance Party having entered into this Agreement, the Investment Agency Agreement or funding or performing its obligations under any Finance Document; provided that, notwithstanding anything herein to the contrary, (x) the Dodd-Frank Wall Street Reform and Consumer Protection Act and all requests, rules, guidelines or directives thereunder or issued in connection therewith, and (y) all requests, rules, guidelines or directives promulgated by the Bank for International

Settlements, the Basel Committee on Banking Supervision (or any successor or similar authority) or the United States or foreign regulatory authorities, in each case pursuant to Basel III, shall in each case be deemed to be an introduction of, or change in, a law or regulation for the purposes hereof, regardless of the date enacted, adopted or issued.

- (c) A Finance Party intending to make a claim for Increased Costs pursuant to clause 16.5(a) shall notify the Investment Agent of the event giving rise to the claim, following which the Investment Agent shall promptly notify AIHL.
- (d) Each Finance Party shall, as soon as practicable, provide a certificate confirming the amount of its Increased Costs.
- (e) Clause 16.5(a) does not apply to the extent any Increased Cost is:
 - (i) compensated for by the payment of Mandatory Cost;
 - (ii) compensated for by **clause 8.3** (or would have been compensated for under **clause 8.3** but was not so compensated solely because any of the exclusions in **paragraph (b)** of **clause 8.3** applied); or
 - (iii) attributable to the willful breach by the relevant Finance Party or its affiliates of any law or regulation.

16.6 General

The certificate of the Finance Party or other Indemnitee as to the amount of any loss or damage sustained or incurred by it shall be conclusive and binding on AIHL except for any manifest error.

17. WAIVERS, REMEDIES CUMULATIVE

The rights of the Finance Parties under the Finance Documents:

- (a) may be exercised as often as necessary;
- (b) are cumulative and not exclusive of its rights under the general law; and
- (c) may be waived only in writing and specifically.

Delay in exercising or non-exercise of any such right is not a waiver of that right.

18. MISCELLANEOUS

18.1 Severance

If any provision of any Finance Document is or becomes illegal, invalid or unenforceable in any jurisdiction, that shall not affect:

- (i) the legality, validity or enforceability in that jurisdiction of any other provision of the Finance Document; or
- (ii) the legality, validity or enforceability in any other jurisdiction of that or any other provision of the Finance Document.

18.2 Counterparts

This Agreement may be executed in any number of counterparts and this shall have the same effect as if the signatures on the counterparts were on a single copy of this Agreement. Delivery of an executed counterpart of a signature page of this Agreement by facsimile or electronic transmission via e-mailed pdf or other similar format shall be effective as delivery of a manually executed counterpart of this Agreement.

18.3 Documentation in English

Each Finance Document and each other agreement, certificate, document and instrument delivered or required to be delivered with respect to this Agreement or any other Finance Document shall be in the English language.

18.4 Amendments

Subject always to **clause 8.4** of the Investment Agency Agreement, this Agreement, each other Finance Document, and any of the terms hereof or thereof, may not be amended, changed, waived, discharged, modified, supplemented or terminated unless such amendment, change, waiver, discharge, modification, supplement or termination is in a writing signed by each of the parties to such Finance Document. No amendment to this Agreement or any other Finance Document to which a Debtor is a party shall be effective without the express approval of the Bankruptcy Court and, insofar as it relates to a disposition of AIHL's property, transfer of AIHL's shares or alterations in the status of AIHL's members, without a prior validation order obtained from the Grand Court of the Cayman Islands.

18.5 Patriot Act Notice

Each Participant that is subject to the Patriot Act and the Investment Agent (for itself and not on behalf of any Participant) hereby notifies AIHL that pursuant to the requirements of the Patriot Act, it is required to obtain, verify and record information that identifies AIHL and the other Obligors, which information includes the name, address and tax identification number of AIHL and the other Obligors and other information regarding AIHL and the other Obligors that will allow such Participant or the Investment Agent, as applicable, to identify AIHL and the other Obligors in accordance with the Patriot Act. This notice is given in accordance with the requirements of the Patriot Act and is effective as to the Participants and the Investment Agent.

18.6 Compliance with Anti-Terrorism Laws and FCPA; Embargoed Persons

- (a) Each Obligor and each of its Subsidiaries will comply with the Anti-Terrorism Laws and all applicable requirements of Governmental Authorities having jurisdiction over such Person and its assets, including those relating to money laundering and terrorism. The Investment Agent shall have the right to audit each Obligor's compliance with the Patriot Act and all applicable requirements of Governmental Authorities having jurisdiction over each Obligor and its assets, including those relating to money laundering and terrorism. In the event that any Obligor fails to comply with the Patriot Act or any such requirements of Governmental Authorities, then the Investment Agent may, at its option, exercise any remedies provided for or permissible under applicable law including, if permitted, causing such Obligor to comply therewith, and any and all costs and expenses incurred by the Investment Agent in connection therewith shall be immediately due and payable by AIHL.
- (b) AIHL covenants and agrees with the Investment Agent that, so long as this Agreement remains in effect and until the Obligations have been paid in full, no Obligor shall, nor shall it cause or permit any Subsidiary to:
 - directly or indirectly, (x) knowingly conduct any business or engage in making or receiving any contribution of funds, goods or services to or for the benefit of any Person described or referenced in **clauses 11.22 and 12.12(a)**, (y) knowingly deal in, or otherwise engage in any transaction relating to, any property or interests in property blocked pursuant to any Anti-Terrorism Law, or (z) knowingly engage in or conspire to engage in any transaction that evades or avoids, or has the purpose of evading or avoiding, or attempts to violate, any of the prohibitions set forth in any Anti-Terrorism Law (and the Obligors shall deliver to the Participants and the Investment Agent any certification or other evidence requested from time to time by any Participant or the Investment Agent, in its sole discretion, confirming the Obligors' compliance with this **clause 18.6**;
 - (ii) cause or permit any funds or proceeds of the Obligors generated from selling Commodities under the Facility or that are used to repay the Obligations to be derived from any unlawful activity with the result that the sale of Commodities under the Facility would be in violation of the FCPA or any other requirement of any Governmental Authority; or
 - (iii) cause or permit (x) any of the funds, proceeds or assets of the Obligors generated from selling Commodities under the Facility or that are used to repay the Obligations to constitute assets of, or be beneficially owned directly or indirectly by, any (1) Embargoed Person, or (2) by any Anti-Terrorism Law, any related enabling legislation or (y) any Embargoed Person to have any direct or indirect interest, of any nature whatsoever in the Obligors, with the result that the investment in the Obligors (whether directly or indirectly) is prohibited by a requirement of any Governmental

12-11076-shl Doc 690-2 Filed 12/04/12 Entered 12/04/12 18:01:19 Exhibit B - DIP Agreement Pg 79 of 97

Authority or the Obligations or the other transactions contemplated by the Finance Documents are in violation of a requirement of any Governmental Authority.

18.7 No Investment Agent Exclusivity

Arcapita Bank and AIHL's respective boards of directors (or its equivalent) may seek and receive proposals for debtor-in-possession financing or similar transactions during the pendency of the Cases from any third party and negotiate such proposals with any third party and provide due diligence information regarding the Obligors and their Affiliates to such third party; *provided* that nothing contained herein shall be deemed to permit any Financial Indebtedness otherwise prohibited hereunder or to affect AIHL's obligation to pay any fees payable to Fortress Credit Corp. or any of its affiliates pursuant to the Fee Letter or the Commitment Letter relating to this Facility.

18.8 Reinstatement

This Agreement shall remain in full force and effect and shall continue to be effective or to be reinstated, as the case may be, if at any time payment and performance of the Obligations, or any part thereof; is, pursuant to applicable law, rescinded or reduced in amount, or must otherwise be restored or returned by any obligee of the Obligations, whether as a "voidable preference," "fraudulent conveyance," or otherwise, all as though such payment or performance had not been made. In the event that any payment, or any part thereof, is rescinded, reduced, restored or returned, the Obligations shall be reinstated and deemed reduced only by such amount paid and not so rescinded, reduced, restored or returned.

18.9 No Strict Construction

The parties hereto have participated jointly in the negotiation and drafting of this Agreement and the other Finance Documents. In the event an ambiguity or question of intent or interpretation arises, this Agreement shall be construed as if drafted jointly by the parties hereto and no presumption or burden of proof shall arise favoring or disfavoring any party by virtue of the authorship of any provisions of this Agreement.

19. NOTICES

19.1 Giving of Notices

All notices or other communications under or in connection with this Agreement shall be given by letter, facsimile or other electronic transmission via e-mailed pdf or other similar format and shall be in English. Any such notice will be deemed to be given as follows:

(a) if by way of letter, when it has been left at the relevant address, provided such delivery was by way of a reputable courier company which returns proof of delivery; and

(b) if by facsimile or other electronic transmission via e-mailed pdf or other similar format, when received in legible form.

However, a notice given in accordance with the above but received on a non-working day or after business hours in the place of receipt will only be deemed to be given on the next working day in that place.

19.2 Addresses for Notices

(a) The address and facsimile number of AIHL for all notices under or in connection with this Agreement are:

Address:	Arcapita Investment Holdings Limited		
	[]	
Attention:	[]	
Fax:	[]	

or such other as AIHL may notify in writing to the Investment Agent by not less than 5 Business Days' prior notice.

(b) The address and facsimile number of the Investment Agent are:

Address: 1345 Avenue of the Americas

46th Floor

New York, NY 10105 United States of America

Attention: Dean Dakolias Fax number: +1.212.798.6099

or such other as the Investment Agent may notify in writing to AIHL by not less than 5 Business Days' prior notice.

19.3 Electronic Communication

- (a) Any communication to be made between the Investment Agent and AIHL under or in connection with this Agreement may be made by electronic mail or other electronic means, if the Investment Agent and AIHL:
 - (i) agree that, unless and until notified to the contrary, this is to be an accepted form of communication;
 - (ii) notify each other in writing of their electronic mail address and/or any other information required to enable the sending and receipt of information by that means; and

- (iii) notify each other of any change to their address or any other such information supplied by them.
- (b) Any electronic communication made between the Investment Agent and AIHL will be effective only when actually received in readable form and in the case of any electronic communication made by AIHL to the Investment Agent only if it is addressed in such a manner as the Investment Agent shall specify for this purpose.

20. ASSIGNMENTS AND TRANSFERS

20.1 Benefit of Agreement

This Agreement, the other Finance Documents, and all Liens and other rights and privileges created hereby or pursuant hereto or to any other Finance Document shall be binding upon AIHL, the estate of AIHL, and any trustee, other estate representative or any successor in interest of AIHL in any Case or any subsequent case commenced under Chapter 7 of the Bankruptcy Code, and shall not be subject to Section 365 of the Bankruptcy Code. This Agreement shall be binding upon, and inure to the benefit of, the successors of the Investment Agent and its permitted assigns, transferees and endorsees. The Liens created by this Agreement and the other Finance Documents shall be and remain valid and perfected in the event of the substantive consolidation or conversion of any Case or any other bankruptcy case of any Debtor to a case under Chapter 7 of the Bankruptcy Code or in the event of dismissal of any Case or the release of any Collateral from the jurisdiction of the Bankruptcy Court for any reason, without the necessity that the Investment Agent file financing statements or otherwise perfect its Liens under applicable law. AIHL may not assign, transfer, hypothecate or otherwise convey its rights, benefits, obligations or duties hereunder or under any of the other Finance Documents without the prior express written consent of the Investment Agent and each Participant. Any such purported assignment, transfer, hypothecation or other conveyance by AIHL without the prior express written consent of the Investment Agent and each Participant shall be void. The terms and provisions of this Agreement and the other Finance Documents are for the purpose of defining the relative rights and obligations of each Obligor, the Investment Agent and each Participant with respect to the transactions contemplated hereby and no Person shall be a third party beneficiary of any of the terms and provisions of this Agreement or any of the other Finance Documents. This Agreement shall be binding upon and inure to the benefit of each party and its permitted successors and assigns.

20.2 Assignments and Transfers by AIHL

AIHL shall not be entitled to assign or transfer any of its rights or obligations under any Finance Document without the prior express written consent of the Investment Agent and each Participant.

20.3 Assignments and Transfers by Investment Agent

The Investment Agent shall only be permitted to assign or transfer any of its rights and benefits under any Finance Document to another bank or other financial institution with the prior approval of AIHL (such approval not to be unreasonably withheld, delayed or conditioned) and further in accordance with the Investment Agency Agreement; *provided* that AIHL shall be deemed to have consented to any such assignment or transfer unless it shall object thereto by written notice to the Investment Agent within 10 Business Days after having received notice of such proposed assignment or transfer. No such approval of AIHL shall be required in respect of an assignment or transfer (i) when an Event of Default has occurred and is continuing, or (ii) to any then-existing Participant or their affiliates.

21. GOVERNING LAW & DISPUTE RESOLUTION

21.1 Governing Law

This Agreement and all obligations and rights arising out of or in connection with this Agreement shall be governed by and construed in accordance with the law of the State of New York, United States of America.

21.2 Jurisdiction

AIHL, THE INVESTMENT AGENT AND EACH OTHER PARTICIPANT IRREVOCABLY AND UNCONDITIONALLY SUBMITS, FOR ITSELF AND ITS PROPERTY, TO THE EXCLUSIVE JURISDICTION OF THE BANKRUPTCY COURT AND, IF THE BANKRUPTCY COURT DOES NOT HAVE (OR ABSTAINS FROM) JURISDICTION, OF THE COURTS OF THE STATE OF NEW YORK SITTING IN NEW YORK COUNTY AND OF THE UNITED STATES DISTRICT COURT OF THE SOUTHERN DISTRICT OF NEW YORK, AND ANY APPELLATE COURT FROM ANY THEREOF, IN ANY ACTION OR PROCEEDING ARISING OUT OF OR RELATING TO THIS AGREEMENT OR ANY OTHER FINANCE DOCUMENT, OR FOR RECOGNITION OR ENFORCEMENT OF ANY JUDGMENT, **EACH** OF THE **PARTIES HERETO IRREVOCABLY** UNCONDITIONALLY AGREES THAT ALL CLAIMS IN RESPECT OF ANY SUCH ACTION OR PROCEEDING MAY BE HEARD AND DETERMINED IN SUCH NEW YORK STATE COURT OR, TO THE FULLEST EXTENT PERMITTED BY APPLICABLE LAW, IN SUCH FEDERAL COURT. EACH OF THE PARTIES HERETO AGREES THAT A FINAL JUDGMENT IN ANY SUCH ACTION OR PROCEEDING SHALL BE CONCLUSIVE AND MAY BE ENFORCED IN OTHER JURISDICTIONS BY SUIT ON THE JUDGMENT OR IN ANY OTHER MANNER PROVIDED BY LAW. NOTHING IN THIS AGREEMENT OR IN ANY OTHER FINANCE DOCUMENT SHALL AFFECT ANY RIGHT TO BRING ANY ACTION OR PROCEEDING RELATING TO THIS AGREEMENT OR ANY OTHER FINANCE DOCUMENT AGAINST AIHL OR ANY OTHER OBLIGOR OR ITS PROPERTIES IN THE COURTS OF ANY JURISDICTION, INCLUDING THE RIGHT OF THE INVESTMENT AGENT TO BRING SUIT OR TAKE OTHER LEGAL ACTION IN ANY OTHER JURISDICTION TO REALIZE ON THE COLLATERAL OR ANY OTHER SECURITY FOR THE OBLIGATIONS, OR TO ENFORCE A JUDGMENT OR OTHER COURT ORDER IN FAVOR OF INVESTMENT AGENT. AIHL EXPRESSLY SUBMITS AND CONSENTS IN ADVANCE TO SUCH JURISDICTION IN ANY ACTION OR SUIT COMMENCED IN ANY SUCH COURT, AND AIHL HEREBY WAIVES ANY OBJECTION THAT SUCH OBLIGOR MAY HAVE BASED UPON IMPROPER VENUE OR FORUM NON CONVENIENS AND HEREBY CONSENTS TO THE GRANTING OF SUCH LEGAL OR EQUITABLE RELIEF AS IS DEEMED APPROPRIATE BY SUCH COURT.

21.3 Waiver of Jury Trial

EACH PARTY HERETO HEREBY WAIVES, TO THE FULLEST EXTENT PERMITTED BY APPLICABLE LAW, ANY RIGHT IT MAY HAVE TO A TRIAL BY JURY IN ANY LEGAL PROCEEDING DIRECTLY OR INDIRECTLY ARISING OUT OF OR RELATING TO THIS AGREEMENT, ANY OTHER FINANCE DOCUMENT OR THE TRANSACTIONS CONTEMPLATED HEREBY OR THEREBY (WHETHER BASED ON CONTRACT, TORT OR ANY OTHER THEORY). EACH PARTY HERETO (A) CERTIFIES THAT NO REPRESENTATIVE, AGENT OR ATTORNEY OF ANY OTHER PARTY HAS REPRESENTED, EXPRESSLY OR OTHERWISE, THAT SUCH OTHER PARTY WOULD NOT, IN THE EVENT OF LITIGATION, SEEK TO ENFORCE THE FOREGOING WAIVER AND (B) ACKNOWLEDGES THAT IT AND THE OTHER PARTIES HERETO HAVE BEEN INDUCED TO ENTER INTO THIS AGREEMENT BY, AMONG OTHER THINGS, THE MUTUAL WAIVERS AND CERTIFICATIONS IN THIS CLAUSE.

21.4 Conflict of Terms

Except as otherwise provided in this Agreement, in clauses 3.5 and 3.7 of the Investment Agency Agreement, or in any of the other Finance Documents by specific reference to the applicable provisions of this Agreement, if any provision contained in this Agreement conflicts with any provision in any of the other Finance Documents, the provision contained in this Agreement shall govern and control.

21.5 Service of Process

Without prejudice to any other mode of service allowed under any relevant law AIHL agrees that the process by which any suit, action or proceeding is begun may be served on it by being delivered to Corporation Service Company 1180 Avenue of the Americas, Suite 210 New York, New York 10036, and hereby appoints Corporation Service Company as its agent for such service of process. Each party hereto irrevocably consents to service of process in the manner provided for notices in **clause 19.2**. Nothing in this agreement will affect the right of any party hereto to serve process in any other manner permitted by applicable law.

12-11076-shl Doc 690-2 Filed 12/04/12 Entered 12/04/12 18:01:19 Exhibit B - DIP Agreement Pg 84 of 97

21.6 Immunity

To the extent that AIHL may in any jurisdiction claim for itself or its assets immunity from suit, execution, attachment (whether in aid of execution, before judgment or otherwise) or other legal process and to the extent that in any such jurisdiction there may be attributed to itself or its assets such immunity (whether or not claimed), AIHL hereby irrevocably agrees not to claim and hereby irrevocably waives such immunity to the full extent permitted by the laws of such jurisdiction.

THIS AGREEMENT is entered into by the parties on the date stated at the beginning of this Agreement.

SCHEDULE 1 CONDITION PRECEDENT DOCUMENTS²

The documents and other evidence referred to in **clause 3.1** (Conditions Precedent) are as follows, in each case in a form satisfactory to the Investment Agent:

1. AIHL AND GUARANTORS

- 1.1 A certified copy of the constitutional documents of each Obligor.
- 1.2 (a) A certified copy of the minutes of the meeting of the board of directors of each Obligor containing resolutions approving the terms of and the transactions contemplated by the Finance Documents to which that Obligor is a party and authorizing a specified Person or Persons to sign the Finance Documents to which it is party on its behalf, and (b) in respect of AIHL, written evidence of the approval by its Joint Provisional Liquidators (in accordance with the Protocol made between them and AIHL's directors) of the terms of, the transactions contemplated by and the execution by AIHL of the Finance Documents.
- 1.3 A certified specimen of the signature of the authorized signatory of each Obligor who will sign the Finance Documents on that Obligor's behalf.
- 1.4 A certificate of a director of each Guarantor:
 - (a) confirming that the guaranteeing of the Facility and the granting of the Liens pursuant to the Security Documents to which it is a party would not cause any guarantee or security limit binding on such Guarantor to be exceeded; and
 - (b) that (except for AIHL Sub, AEID II, RailInvest and WTHL), is not a Debtor, certifying, in form and substance reasonably satisfactory to the Investment Agent, that such Guarantor is solvent.
- 1.5 To the extent such approval is necessary or advisable for the enforceability of any Finance Document against any Obligor, a certified copy of resolutions of the shareholder(s) of such Obligor approving the entry by that Obligor into the Finance Documents to which it is a party.

2. **LEGAL OPINIONS**

- 2.1 A legal opinion of Gibson, Dunn & Crutcher LLP, legal advisers to AIHL and each Guarantor in New York.
- 2.2 A legal opinion of [], legal advisers to Arcapita Bank in the Kingdom of Bahrain.

² Subject to continuing review.

2.3	A legal opinion of [], legal advisers to AIHL and each Guarantor in the Cay	man
	Islands.		

3. FEES, COSTS AND EXPENSES

Evidence that the fees, costs and expenses then due from AIHL pursuant to this Agreement and under the Commitment Letter and the Fee Letter have been paid or will be paid by on the Transaction Date.

4. FINANCE DOCUMENTS

Each of the Finance Documents duly executed by each of the parties to them and all documents and other instruments to be delivered to the Investment Agent in accordance with the Finance Documents, including, but not limited to, the following, each in form and substance satisfactory to the Investment Agent:

1.	this Agreement, duly executed and delivered by AIHL and the investment Agent;
ii.	the Guarantee, dated as of December, 2012, duly executed and delivered by [] and the Investment Agent;
iii.	the U.S. Security Agreement, duly executed and delivered by [] Arcapita Inc., and the Security Agent;
iv.	the Cayman Charge, duly executed and delivered by Arcapita Bank and the Security Agent; and
v.	the Cayman Debentures, duly executed and delivered by [] and the Security Agent.

5. PROCESS AGENT

Evidence of the appointment of process agent for each Obligor.

6. OTHER DOCUMENTS AND EVIDENCE

- 6.1 Evidence of the approval of the Executive Committee of each Obligor's Shari'ah Advisory Board of the execution of the Finance Documents and the issuance of a "Fatwa" that the Finance Documents and the transactions contemplated therein are in compliance with Shari'ah principles.
- 6.2 The initial DIP Budget.
- 6.3 The Investment Agent shall have received (a) satisfactory evidence that the Obligors shall have obtained all required consents and approvals of all Persons including all requisite Governmental Authorities, in connection with the filing of the Cases and to the execution, delivery and performance of this Agreement and the other Finance Documents, the consummation of the transactions contemplated hereby and thereby, the

validity and perfection of the Liens of the Security Agent in the Collateral with the priority contemplated in the Finance Documents, and the exercise by the Finance Parties of the rights and remedies with respect thereto or (b) a certificate from a director of AIHL in form and substance reasonably satisfactory to the Investment Agent affirming that no such consents or approvals are required.

- 6.4 The Investment Agent shall have received a certified copy of a validation order granted in the Cayman Proceedings, pursuant to section 99 of the Companies Law (2012 revision) of the Cayman Islands, in respect of (a) any disposition of AIHL's property made under, in connection with or resulting from the transactions contemplated by the Finance Documents and (b) any transfer of AIHL's shares made by virtue of, or by way of perfection or enforcement of, the Lien over AIHL's shares to be created under the Security Documents.
- 6.5 The Investment Agent shall have received a certificate, dated the Effective Date and signed by an authorized officer of AIHL, certifying compliance with each of the conditions precedent set forth in **clause 3.2** have been satisfied.
- 6.6 The Security Agent shall have received (a) originals of all certificates (if any) representing the pledged Equity Interests constituting Collateral, together with undated stock powers executed in blank and instruments constituting Collateral indorsed in blank, and (b) for all Equity Interests constituting Collateral, a copy of the share register of each Person the Equity Interests of which constitute Collateral, notating the Lien granted to the Security Agent by the Obligor owning such Equity Interests, in the case of clauses (a) and (b), in accordance with the terms of the applicable Security Documents.

SCHEDULE 2 FORM OF TRANSACTION REQUEST

From:	Arcapita	Investment Holdings Lin	mited		
To:	Fortress	Credit Corp., as Investme	ent Agent		
Date:	[]			
DATE		EMBER	SSESSION MASTER M S AMENDED FROM		
	defined in defined in the defined in	•	e same meaning when us	ed in this documer	nt. This is a
We wi	ish to enter	r into a Purchase Contrac	et as follows:		
	(a)	Transaction Date:	[]	
	(b)	Commodity:	[]	
	(c)	Cost Price:	ſ	1	

The proposed Purchase Contract to be entered into pursuant to this Transaction Request [will be][will not be] a Subsequent Purchase Contract

[FOR PURCHASE CONTRACTS (OTHER THAN SUBSEQUENT PURCHASE CONTRACTS), INCLUDE THE FOLLOWING:] [We confirm that each of the representations and warranties contained in this Agreement and the other Finance Documents (subject to, after the Effective Date, any Permitted Representation Exceptions), is true and correct in all material respects, except for representations qualified by materiality or Material Adverse Effect, in which case such representation and warranty are true and correct in all respects, on the date of this Transaction Request (or such earlier date as may be expressly referenced in any such representation and warranty). We further confirm that each of the conditions precedent set forth in clause 3.2 of the Agreement have been satisfied on the date of this Transaction Request and will be satisfied on the proposed Transaction Date.]

[FOR SUBSEQUENT PURCHASE CONTRACTS ENTERED INTO ON A BRINGDOWN DATE INCLUDE THE FOLLOWING:] [We confirm that each of the representations and warranties contained in this Agreement and the other Finance Documents (subject to, after the Effective Date, any Permitted Representation Exceptions), is true and correct in all material respects, except for representations qualified by materiality or Material Adverse Effect, in which case such representation and warranty are true and correct in all respects, on the date of this Transaction Request (or such earlier date as may be expressly referenced in any such representation and warranty). We confirm that each of the conditions

12-11076-shl Doc 690-2 Filed 12/04/12 Entered 12/04/12 18:01:19 Exhibit B - DIP Agreement Pg 89 of 97

precedent set forth in **clause 3.2** of the Agreement (other than **clauses (e)**, **(g)** and **(i)**) have been satisfied on the date of this Transaction Request and will be satisfied on the proposed Transaction Date.]

[FOR SUBSEQUENT PURCHASE CONTRACTS NOT ENTERED INTO ON A BRINGDOWN DATE INCLUDE THE FOLLOWING:] [We confirm that each of the conditions precedent set forth in clause 3.2 of the Agreement (other than clauses (g) and (i)) have been satisfied on the date of this Transaction Request and will be satisfied on the proposed Transaction Date.]

authorised signatory
Arcanita Investment Holdings Limited

SCHEDULE 3 FORM OF OFFER LETTER AND ACCEPTANCE

From:	Fortress Credit Corp., as Investment Agent				
To:	Arcapita Investment Holdings Limite	ed Da	te:		
DATE	RPRIORITY DEBTOR-IN-POSSESSION D DECEMBER, 2012 AS AMENI EEMENT")				
Terms Offer I	defined in the Agreement have the same mea Letter.	ning when	used in	this document	. This is an
We off	Fer to sell you the following Commodities on t	the followi	ng terms	:	
A.	Transaction Date:	[]		
B.	Quantity and type of Commodities:	[]		
C.	Cost Price (per unit Commodity):	[]		
D.	Cost Price:	[]		
E.	Increased Costs Amount:	[]		
F.	Deferred Sale Price:	[]		
G.	Expressly without prejudice to clause 8.4(a) of the Agreement, Value Added Tax:	[]		
H.	Deferred Payment Date	[]		
clauses incorpo	sove offer is subject to the terms of the s 18 and 21 of the Agreement, which shorated into the Purchase Contract to be made fer Letter.	nall (muta	tis muta	indis) be deer	med to be
	ss Credit Corp., estment Agent				
Accept We acc	cance cept the offer in the above Offer Letter.				
	sed signatory ita Investment Holdings Limited			Date: []

SCHEDULE 4 DISCLOSURE SCHEDULE

- 1. Arcapita Bank B.S.C.(c) is the borrower under two facilities made available by Standard Chartered Bank: (i) a \$50 million facility dated May 30, 2011 (as the same may be amended, the "SCB May 2011 Facility"); and (ii) a \$50 million facility dated December 22, 2011 (as the same may be amended, the "SCB December 2011 Facility" and, together with the SCB May 2011 Facility).
- 2. The SCB May 2011 Facility is guaranteed by each of Arcapita Investment Holdings Limited ("AIHL"), Arcapita LT Holdings Limited ("AIHL Sub") and WindTurbine Holdings Limited ("WTHL"). These guarantees are secured by: (i) a first priority pledge of AIHL's shares in AIHL Sub; (ii) a first priority pledge of AIHL Sub's shares in WTHL; and (iii) a second priority pledge of AIHL Sub's shares in AEID II Holdings Limited ("AEID II") and RailInvest Holdings Limited ("RailInvest").
- 3. The SCB December 2011 Facility is guaranteed by each of AIHL, AIHL Sub, WTHL, AEID II, and RailInvest. These guarantees are secured by: (i) a second priority pledge of AIHL's shares in AIHL Sub; (ii) a first priority pledge of AIHL Sub's shares in AEID II and RailInvest; and (iii) a second priority pledge in AIHL Sub's shares in WTHL.
- 4. Arcapita Industrial Management Sarl and its subsidiaries are party to certain agreements entered into in connection with the pending EuroLog IPO.
- 5. Administration Agreements are referred to in lists as provided to the Investment Agent on October 24, 2012 and October 30, 2012.
- 6. HarborVest Agreements as provided to the Investment Agent on October 24, 2012 and separate disclosure as provided on November 1, 2012.
- 7. Share Purchase Agreements by Middle East institutional investor in connection with its confidential investments in (i) J. Jill Group, (ii) Viridian Group Holding Limited, (iii) CEPL Group, and (iv) Freightliner Group Limited. A Separate disclosure regarding the foregoing was provided to the Investment Agent on November 1, 2012.
- 8. \$1.1 Billion Master Murabaha Agreement (as amended and restated by an Amendment and Restatement Agreement dated April 11, 2007) dated March 28, 2007 between Arcapita Bank B.S.C.(c) and WestLB AG, London Branch, as Investment Agent.
- 9. [Lusail]

SCHEDULE 5 LIST OF ENTITIES

A. <u>LT Caycos</u>
B. <u>Transaction Holdcos</u>
C. WCFs
D. <u>PVs</u>
E. Syndication Companies
F. Non-Guarantor Subsidiaries
G. <u>Immaterial Subsidiaries</u>

12-11076-shl Doc 690-2 Filed 12/04/12 Entered 12/04/12 18:01:19 Exhibit B - DIP Agreement Pg 93 of 97

SCHEDULE 6 FORM OF ACCESSION LETTER

То	: Fortress Credit Corp., as Investment Agent
Fre	om: [Subsidiary] and Arcapita Investment Holdings Limited
Da	ted: [] 201[]
De	ar Sirs
	RCAPITA INVESTMENT HOLDINGS LIMITED — SUPERPRIORITY DEBTOR-IN- POSSESSION MASTER MURABAHA AGREEMENT DATED DECEMBER, 2012 (THE "AGREEMENT")
1.	We refer to the Agreement. This is an Accession Letter. Terms defined in the Agreemen have the same meaning in this Accession Letter unless given a different meaning in this Accession Letter.
2.	[Subsidiary] agrees to become an Additional Guarantor and to be bound by the terms of the Guarantee (or, at the election of the Investment Agent, an Additional Guarantee) as an Additional Guarantor pursuant to clause 12.25(c) of the Agreement.
3.	[Subsidiary] agrees to be bound by the terms of the applicable Security Documents (or, at the election of the Investment Agent, enter into a new Security Document) pursuant to clause 12.25(c) of the Agreement.
4.	[Subsidiary] agrees to be bound by those terms of the Investment Agency Agreement which are applicable to a Guarantor as if references therein to a Guarantor were also references to such Subsidiary, as an Additional Guarantor.
5.	[Subsidiary] is a company duly incorporated under the laws of [name of relevan jurisdiction].
6.	[Subsidiary's] administrative details are as follows:
	Address:
	Fax No: Attention:

7. This Accession Letter shall be governed by and construed in accordance with the law of the State of New York, United States of America.

12-11076-shl Doc 690-2 Filed 12/04/12 Entered 12/04/12 18:01:19 Exhibit B - DIP Agreement Pg 94 of 97

[SUBSIDIARY]

Signed by
duly authorized for and
on behalf of
[Subsidiary]
)

SCHEDULE 7 CONDITIONS PRECEDENT REQUIRED TO BE DELIVERED BY EACH ADDITIONAL GUARANTOR

- 1. An Accession Letter, duly executed by AIHL and the Additional Guarantor.
- 2. A certified copy of the constitutional documents of the Additional Guarantor, certified in accordance with paragraph 6.
- 3. A certified specimen of the signature of the authorized signatory of the Additional Guarantor who will sign the Finance Documents to which such Additional Guarantor is party on such Additional Guarantor's behalf.
- 4. A certified copy of the minutes of the meeting of the board of directors of the Additional Guarantor containing resolutions approving the terms of and the transactions contemplated by the Finance Documents to which such Additional Guarantor is a party and authorizing a specified Person or Persons to sign the Finance Documents to which such Additional Guarantor is party on such Additional Guarantor's behalf. The minutes and resolutions shall set out in detail the factors considered by the directors in determining the corporate benefit of the transactions for their respective companies.
- 5. A certificate of a director of the Additional Guarantor:
 - a. confirming that the guaranteeing of the Facility and the granting of the Liens pursuant to the Security Documents to which such Additional Guarantor is a party would not cause any guarantee or security limit binding on such Additional Guarantor to be exceeded; and
 - b. that is not a Debtor, certifying, in form and substance reasonably satisfactory to the Investment Agent, that such Additional Guarantor is solvent.
- 6. A certificate of the Additional Guarantor (signed by its authorized signatory) certifying that each document relating to it specified in this Schedule 7 is correct, complete and in full force and effect as at a date no earlier than the date of the Additional Guarantee.
- 7. A certificate of the Additional Guarantor (signed by authorized signatory) certifying that the Additional Guarantor is in compliance with all material regulatory or governmental requirements.
- 8. The latest audited financial statements of the Additional Guarantor.
- 9. A legal opinion of Gibson, Dunn & Crutcher LLP, legal advisers to the Investment Agent and the Additional Guarantor in New York.
- 10. A legal opinion of the legal advisers to the Investment Agent in the jurisdiction in which the Additional Guarantor is incorporated.

- 11. If the Additional Guarantor is incorporated in a jurisdiction other than New York, a letter to the Investment Agent from the process agent referred to in **clause 5** accepting its appointment by the Additional Guarantor.
- 12. Such documentation and other evidence as is reasonably requested by the Investment Agent (for itself or on behalf of a Participant), including as required pursuant to **clause 12.17**, in order for the Investment Agent (or that Participant) to carry out and be satisfied it has complied with all necessary "know your customer" or other similar checks under all applicable laws and regulations.
- 13. The Investment Agent shall have received (a) satisfactory evidence that the Additional Guarantor shall have obtained all required consents and approvals of all Persons including all requisite Governmental Authorities, in connection with the filing of the Cases, if applicable, and to the execution, delivery and performance of this Agreement and the other Finance Documents, the consummation of the transactions contemplated hereby and thereby, the validity and perfection of the Liens of the Security Agent in the Collateral with the priority contemplated in the Finance Documents, and the exercise by the Finance Parties of the rights and remedies with respect thereto or (b) a certificate from a director of such Additional Guarantor in form and substance reasonably satisfactory to the Investment Agent affirming that no such consents or approvals are required.
- 14. Any other agreements, documents and evidence as the Investment Agent considers necessary (if it has notified the Additional Guarantor accordingly) in connection with the entry into and performance of the transactions contemplated by this Agreement and the other Finance Documents to which the Additional Guarantor is party or the validity and enforceability of this Agreement and the other Finance Documents to which the Additional Guarantor is party.

EXECUTION PAGE OF SUPERPRIORITY DEBTOR-IN-POSSESSION MASTER MURABAHA AGREEMENT

AIHL

Signed by)
duly authorized for and)
on behalf of)
Arcapita Investment)
Holdings Limited)
)

INVESTMENT AGENT

Signed by
duly authorized for and
on behalf of
Fortress Credit Corp,
as Investment Agent

12-11076-shl Doc 690-3 Filed 12/04/12 Entered 12/04/12 18:01:19 Exhibit C - DIP Budget Pg 1 of 6

Exhibit C

DIP Budget

12-11076-shl Doc 690-3 Filed 12/04/12 Entered 12/04/12 18:01:19 Exhibit C - DIP Budget Pg 2 of 6

ARCAPITA

Cash Flow Forecast - CONSOLIDATED
For the Period Ending 3/30/13

For the Period Ending 3/30/13				_				_	_			_		_			
(\$'s in 000's)	Forecast 12/9/12 12/9/12	Forecast 12/16/12	Forecast 12/23/12	Forecast 12/30/12	Forecast 1/6/13	Forecast 1/13/13	Forecast 1/20/13	Forecast 1/27/13	Forecast 2/3/13	Forecast 2/10/13	Forecast 2/17/13	2/24/13	Forecast 3/3/13	Forecast 3/10/13	Forecast 3/17/13	Forecast 3/24/13	16 Weeks Ending
	3/24/13 12/15/12	12/22/12	12/29/12	1/5/13	1/12/13	1/19/13	1/26/13	2/2/13	2/9/13	2/16/13	2/23/13	3/2/13	3/9/13	3/16/13	3/23/13	3/30/13	3/30/13
Receipts																	
Proceeds from Exits	\$ -	\$ -	\$ - !	\$ 31,750 \$	- \$	- 5	\$ - \$; -	\$ -	\$ - \$	\$ -	\$ -	\$ -	\$ -	\$ -	\$ -	\$ 31,750
Deal Company Murabahas	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-
Management Fees	-	-	-	998	-	125	-	-	-	-	-	-	172	-	-	250	1,544
Yield from Deal Companies	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-
All Other		-	-	19	-						-			-			19
Total Receipts		-	-	32,767		125	<u>-</u>	<u>-</u>		<u>-</u>		-	172		-	250	33,314
<u>Disbursements</u>																	
G&A Expenses																	
Atlanta	82	70	6	110	6	147	6	110	6	84	75	92	24	6	213	42	1,081
Bahrain	1,399	659	680	251	209	535	215	181	228	463	237	289	302	462	354	563	7,025
Hong Kong	25	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	25
London	13	1,639	379	713	34	18	12	607	27	14	12	606	12	30	58	952	5,124
Singapore	515	70	9	47	157	38	7	47	104	79	11	47	7	168	23	64	1,392
Point Park Properties	250	250	250	250	150	150	150	150	150	150	150	150	150	150	150	150	2,800
Total G&A Expenses	2,284	2,687	1,324	1,371	555	888	389	1,095	516	791	484	1,184	495	816	799	1,771	17,448
Staff Expenses																	
Atlanta	151	33	-	151	-	151	33	151	-	144	33	144	-	144	33	144	1,313
Bahrain	89	-	2,381	-	-	145	869	1	-	200	802	1	-	-	50	783	5,322
Hong Kong	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-
London	-	346	-	-	-	25	346	-	11	-	304	-	11	-	-	304	1,348
Singapore	-	190	-	-	-	190	-	-	-	190	-	-	-	-	266	-	834
Point Park Properties										-							
Total Staff Expenses	240	569	2,381	151	-	510	1,249	152	11	534	1,139	145	11	144	349	1,231	8,817
Other Expenses																	
AHQ Payments	_		_	_	_	_	_	_		_	_		_	_	_	_	_
Other	100	53	-	-	-	-	-	150	-	-	20	-	-	-	-	-	323
Total Other Expenses	100	53						150			20						323
	100	33						130			20						323
Deal Funding & Expenses																	
Private Equity	175	75	-	-	89	-	-	-	39	-	-	-	89	-	-	-	468
Real Estate	1,054	750	429	11,425	99	-	-	2,606	-	-	-	-	250	10,000	-	-	26,613
Infrastructure	3,526	750	-	1,322	-	-	-	1,234	-	-	-	-	-	-	-	-	6,832
Venture Capital		-	-	-	1,500	-	-		-	-	-	-	-	-	-		1,500
Total Deal Funding & Expenses	4,755	1,575	429	12,747	1,688	-	-	3,841	39	-	-	-	339	10,000	-	-	35,414
Total Disbursements	7,379	4,884	4,134	14,269	2,244	1,398	1,638	5,238	567	1,324	1,643	1,329	846	10,960	1,147	3,003	62,002
Operating Cash Flow	(7,379)	(4,884)	(4,134)	18,498	(2,244)	(1,273)	(1,638)	(5,238)	(567)	(1,324)	(1,643)	(1,329)	(674)	(10,960)	(1,147)	(2,753)	(28,689)
Operating cash flow	(1,313)	(4,004)	(4,134)	10,430	(2,244)	(1,273)	(1,030)	(3,230)	(307)	(1,324)	(1,043)	(1,323)	(074)	(10,500)	(1,147)	(2,733)	(20,003)
Restructuring																	
Restructuring Fees	6,603	5,111	-	2,205	3,479	122	-	51	6,434	-	122	-	5,604	-	122	4,092	33,945
Critical / Foreign Vendor Payments	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-
Ordinary Course Professionals	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-
Payroll Adjustments	167	564	511	-	-	80	156	-	1,110	790	669	-	-	2,143	-	1,278	7,469
Other Restructuring Costs		-	-	-	-	-	-		-	-	-	-	-	-	-		
Total Restructuring	6,770	5,675	511	2,205	3,479	202	156	51	7,544	790	791	-	5,604	2,143	122	5,371	41,414
Deht Service																	
<u>Debt Service</u> Financing Cost - SCB			_	6,759	_	_	_	500			_	500				1,000	8,759
DIP Interest & Fees	2,000	1,750	-	496	-	-	750	937	-	-	-	1,104	-	-	-	1,380	8,416
DIP Receipts/Payments	(25,000)		-	31,750	-	-	(25,000)	23/	-	-	-	1,104	-	-	-	1,360	(118,250)
Restructuring Accrual / Completion Fees Adjustme		(100,000)	-	31,730	-	-	(23,000)	-	-	-	-	-	-	-	-	27,656	27,656
Total Debt Service	(23,000)	(98,250)		39,005			(24,250)	1,437				1,604		-		30,036	(73,419)
Net Cash Flow	8,851	87,691	(4,645)	(22,712)	(5,723)	(1,475)	22,456	(6,726)	(8,110)	(2,114)	(2,434)	(2,933)	(6,278)	(13,104)	(1,270)	(38,159)	3,316
Intercompany Transfers																	
Cash Inflows (Source of Cash)	167	4,540	4,548	2,300	3,150	1,569	2,393	1,100	6,927	2,280	2,664	1,350	5,050	2,840	1,816	8,451	51,143
Cash Outflows (Use of Cash)	167	4,540	4,548	2,300	3,150	1,569	2,393	1,100	6,927	2,280	2,664	1,350	5,050	2,840	1,816	8,451	51,143
Total Intercompany Transfers	- 107				-	-,,,,,,	-,,,,,			- 2,200	2,004	-	3,030	2,040			31,143
												·					
Balances & Liquidity																	
Beginning Book Balance	51,368	60,219	147,910	143,265	120,554	114,831	113,356	135,811	129,086	120,975	118,861	116,428	113,495	107,217	94,113	92,843	51,368
	8,851	87,691	(4,645)	(22,712)	(5,723)	(1,475)	22,456	(6,726)	(8,110)	(2,114)	(2,434)	(2,933)	(6,278)	(13,104)	(1,270)	(38,159)	3,316
Net Receipts, Disbursements & Transfers Ending Book Balance	60,219	87,691 147,910	(4,645) 143,265	(22,712) 120,554	(5,723) 114,831	113,356	22,456 135,811	(6,726) 129,086	(8,110) 120,975	(2,114) 118,861	(2,434) 116,428	(2,933) 113,495	107,217	(13,104) 94,113	92,843	(38,159) 54,684	3,316 54,684
criting book balance	60,219	147,910	143,265	120,554	114,831	113,356	133,811	129,086	120,975	116,861	110,428	113,495	107,217	94,113	92,843	54,684	54,684
Float/Foreign Exchange	500	500	500	500	500	500	500	500	500	500	500	500	500	500	500	_	
	500	300	300	200	200	200	300	300	300	300	200	200	500	500	500	-	-
			_	_	_	_	_	_		_						_	
Interbank Transfers	(35,000)	(35,000)	(35,000)	(35,000)	(35,000)	(35,000)	(35,000)	(35,000)	(35,000)	(35,000)	(35,000)	(35,000)	(35,000)	(35,000)	(35,000)	(35,000)	(35,000)
	(35,000) \$ 25,719	(35,000)	(35,000)	(35,000) \$ 86,054 \$	(35,000) 8 80,331 \$	(35,000) 78,856	(35,000) \$ 101,311 \$	(35,000)	(35,000) \$ 86,475	(35,000) \$ 84,361 \$	(35,000) \$ 81,928	(35,000)	(35,000)	(35,000)	(35,000)	(35,000) \$ 19.684	(35,000)

12-11076-shl Doc 690-3 Filed 12/04/12 Entered 12/04/12 18:01:19 Exhibit C - DIP Budget Pg 3 of 6

ARCAPITA

Cash Flow Forecast - TOTAL DEBTOR
For the Period Ending 3/30/13

Property	For the Period Ending 3/30/13	Forecast	Forecast	Forecast	Forecast	Forecast	Forecast	Forecast	Forecast	Forecast	Forecast	Forecast	Forecast	Forecast	Forecast	Forecast	Forecast	16 Weeks
Marcial Content	(\$'s in 000's) 12																	
Procession for the property of the process of the p		24/13 12/15/12	12/22/12	12/29/12	1/5/13	1/12/13	1/19/13	1/26/13	2/2/13	2/9/13	2/16/13	2/23/13	3/2/13	3/9/13	3/16/13	3/23/13	3/30/13	3/30/13
Segregation of the segretary of the seg				<u></u>	. 24.750		*	_	_			<u></u>	<u>_</u>		•			ć 24.750
Management (a)		\$ -	\$ -	\$ -:	31,/50 \$	- :	\$ -	\$ -	\$ -	\$ -	> -:	> -	\$ -	\$ -	\$ -	\$ -	\$ -	\$ 31,750
Miniman		-	-	-	740	-	-	-	-	-	-	-	-	-	-	-	-	740
A CHOUNG TOWARD AND MATERIAL AN		-	-	-	740	-	-	-	-	-	-	-	-	-	-	-	-	746
The Protect of the Pr					10													10
Part																		
Control Cont	rotal necespto				52,517													32,317
Part	<u>Disbursements</u>																	
See Topical Management 139	G&A Expenses																	
Second	Bahrain	1,399	659	680	251		535	215	181	228	463			302	462	354		
Part	Total G&A Expenses	1,399	659	680	251	209	535	215	181	228	463	237	289	302	462	354	563	7,025
Part	Staff Evnences																	
Treal Plant Response		80	_	2 381	_	_	1/15	860	1		200	802	1	_	_	50	783	5 322
Control									1				1					
Control Charles	Total Staff Expenses	03		2,501			143	003	-		200	002	-			30	,03	3,322
Total Deference 100 51 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1																		
Trais other Expenses Provide Equity 157 87 87 87 88 88 81 81 81 81 81 81 81 81 81 81 81		-		-	-	-	-	-		-	-		-	-	-	-	-	-
Private function 175 75 125				-	-	-	-	-		-	-		-	-	-	-	-	
Private Equity 175	Total Other Expenses	100	53	-	-	-	-	-	150	-	-	20	-	-	-	-	-	323
Private Equity 175	Deal Funding & Expenses																	
Part		175	75	_	-	89	-	_	_	39	_	-	-	89	_	-	-	468
Part				429	11.425		-	_	2.606	-	_	-	-		10.000	-	-	
Part				_		-	-	_		_	_	-	-	-	-	-	-	
Total Deal Funding & Lypenes		-	-	-	-	1,500	-	-	, .	-	-	-	-	-	-	-	-	
Part Disbursements 6,343 2,287 3,498 12,998 1,877 1,978 1,084 4,173 2,677 16,681 1,085 1,085 1,085 1,082 1,042 10,462 10,462 10,462 1,047 1,045 1,		4,755	1,575	429	12,747	1,688	-	-	3,841	39	-	-	-	339	10,000	-	-	35,414
Persistrication Control Contro			2 207	2 400	43.000	4.007	670	4 004	4.472	267		4.050	200	C42	10.153	***	4 247	
Restructuring Fees	Total Disbursements	6,343	2,287	3,490	12,998	1,897	6/9	1,084	4,173	267	663	1,058	290	642	10,462	404	1,347	48,084
Restructuring Fees 6,609 5,111 - 2,205 3,479 122 - 51 6,434 - 122 - 5,604 - 122 4,092 33,945 Cricial Foreign Verdor Payments 6 - 5 - 5 - 5 - 5 - 5 - 5 - 5 - 5 - 5 -	Operating Cash Flow	(6,343)	(2,287)	(3,490)	19,519	(1,897)	(679)	(1,084)	(4,173)	(267)	(663)	(1,058)	(290)	(642)	(10,462)	(404)	(1,347)	(15,567)
Restructuring Fees 6,609 5,111 - 2,205 3,479 122 - 51 6,434 - 122 - 5,604 - 122 4,092 33,945 Cricial Foreign Verdor Payments 6 - 5 - 5 - 5 - 5 - 5 - 5 - 5 - 5 - 5 -	Postructuring																	
Conticial Foreign Vendor Payments		6 602	E 111		2 205	2 470	122		E1	6 121		122		E 604		122	4.002	22.045
Payroll Adjustments		0,003	3,111	-	2,203	3,473	122	-		0,434	-		-	3,004	-	122		33,543
Total Restructuring Costs Total Restructuring Cost			564					113		833								2 860
Polis			504	113			_	113		-		-			133	_	720	2,000
Politi Service Politic No.		6.603	5.675	113	2.205	3.479	122	113	51	7.267	-	477	-	5.604	153	122	4.820	36.806
Financing Cost - CSG IP Intercemptary Transfers IR Cash Inflows (Source of Cash) IR Cash Inflows (Source	•		5,5.5		_,	-,				.,							,,,,,	
DiP Interers & Fees																		
DiP Receipts/Payments (25,000) (100,000) . 31,750		-		-		-	-			-	-	-		-	-			
Restructuring Accrual / Completion Fees Adjustment				-		-	-		937	-	-	-	1,104	-	-	-	1,380	
Total Debt Service (23,000) (98,250) 39,085 39,005 (24,250) 1,437 1,604 30,036 (73,419)		(25,000)	(100,000)	-	31,750	-	-	(25,000)	-	-	-	-	-	-	-	-	27.656	
Net Cash Flow 10,054 90,288 (3,604) (21,691) (5,376) (802) 23,052 (5,661) (7,534) (663) (1,536) (1,894) (6,246) (10,615) (526) (36,203) 21,066 Cash Inflows (Source of Cash)		(22,000)	(00.350)		20.005	-		(24.250)	1 427		-		1.004					
Intercompany Transfers	Total Debt Service	(23,000)	(98,230)		39,005			(24,250)	1,437				1,604				30,036	(73,419)
Cash Inflows (Source of Cash) Cash Outflows (Use of Cash)	Net Cash Flow	10,054	90,288	(3,604)	(21,691)	(5,376)	(802)	23,052	(5,661)	(7,534)	(663)	(1,536)	(1,894)	(6,246)	(10,615)	(526)	(36,203)	21,046
Cash Inflows (Source of Cash) Cash Outflows (Use of Cash)																		
Cash Outflows (Use of Cash) Total Intercompany Transfers 167 4,540 4,548 2,300 3,150 1,569 2,393 1,100 6,927 2,280 2,664 1,350 5,050 2,800 1,816 8,451 51,143 104 104 104 104 104 104 104 104 104 104																		
Total Intercompany Transfers (167) (390) (1,198) (550) (200) (619) (1,143) (650) (477) (1,430) (1,114) (1,000) - (2,100) (1,066) (2,101) (14,293)		-																
Balances & Liquidity Beginning Book Balance 46,758 56,645 146,544 141,743 119,502 113,926 112,505 134,415 128,104 120,093 118,000 115,351 112,457 106,211 93,406 91,814 46,758 Net Receipts, Disbursements & Transfers 9,887 89,899 (4,801) (22,241) (5,576) (1,421) 21,910 (6,311) (8,011) (2,092) (2,650) (2,894) (6,246) (12,805) (1,592) 38,303) 6,753 Ending Book Balance 56,645 146,544 141,743 119,502 113,926 112,505 134,415 128,104 120,093 118,000 115,351 112,457 106,211 93,406 91,814 46,758 Float/Foreign Exchange 40 411,743 119,502 113,926 112,505 134,415 128,104 120,093 118,000 115,351 112,457 106,211 93,406 91,814 53,511 53,511 53,511 53,511 53,511 53,406 91,814 53,511																		
Beginning Book Balance 46,758 56,645 146,544 141,743 119,502 113,926 112,505 134,415 128,104 120,093 118,000 115,351 112,457 106,211 93,406 91,814 46,758 Net Receipts, Disbursements & Transfers 9,887 89,899 (4,801) (22,241) (5,576) (1,421) 21,910 (6,311) (8,011) (2,092) (2,650) (2,894) (6,246) (12,805) (1,592) (38,303) 6,753 (1,904)	Total Intercompany Transfers	(167)	(390)	(1,198)	(550)	(200)	(619)	(1,143)	(650)	(477)	(1,430)	(1,114)	(1,000)		(2,190)	(1,066)	(2,101)	(14,293)
Net Receipts, Disbursements & Transfers 9,887 89,899 (4,801) (22,241) (5,576) (1,421) 21,910 (6,311) (8,011) (2,092) (2,650) (2,894) (6,246) (12,805) (1,592) (38,303) 6,753 (1,916) (Balances & Liquidity																	
Net Receipts, Disbursements & Transfers 9,887 89,899 (4,801) (22,241) (5,576) (1,421) 21,910 (6,311) (8,011) (2,092) (2,650) (2,894) (6,246) (12,805) (1,592) (38,303) 6,753 (1,916) (Beginning Book Balance	46.758	56,645	146,544	141,743	119,502	113,926	112,505	134,415	128,104	120,093	118,000	115,351	112,457	106,211	93,406	91,814	46,758
Ending Book Balance 56,645 146,544 141,743 119,502 113,926 112,505 134,415 128,104 120,093 118,000 115,351 112,457 106,211 93,406 91,814 53,511 53,511 Float/Foreign Exchange 400 400 400 400 400 400 400 400 400 40																		
Interbank Transfers																		
Interbank Transfers			•															 ,
Less: Placements Held at Banks (35,000) (35,000) (35,000) (35,000) (35,000) (35,000) (35,000) (35,000) (35,000) (35,000) (35,000) (35,000) (35,000) (35,000) (35,000) (35,000) (35,000)		400	400	400	400	400	400	400	400	400	400	400	400	400	400	400	-	-
		-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-
Ending Bank Balance \$ 22,045 \$ 111,944 \$ 107,143 \$ 84,902 \$ 114,326 \$ 112,905 \$ 134,815 \$ 128,504 \$ 120,493 \$ 118,400 \$ 115,751 \$ 112,857 \$ 106,611 \$ 93,806 \$ 92,214 \$ 53,511 \$ 18,511																		
	Ending Bank Balance	\$ 22,045	\$ 111,944	\$ 107,143	\$ 84,902	114,326	\$ 112,905	\$ 134,815	\$ 128,504	\$ 120,493	\$ 118,400	\$ 115,751	\$ 112,857	\$ 106,611	\$ 93,806	\$ 92,214	\$ 53,511	\$ 18,511

12-11076-shl Doc 690-3 Filed 12/04/12 Entered 12/04/12 18:01:19 Exhibit C - DIP Budget Pg 4 of 6

ARCAPITA

Cash Flow Forecast - NON-DEBTOR For the Period Ending 3/30/13

(\$'s in 000's)	Forecast 12/9/12 12/9/12	Forecast 12/16/12				Forecast 1/13/13	Forecast 1/20/13	Forecast 1/27/13	Forecast 2/3/13		Forecast 2/17/13	Forecast 2/24/13	Forecast 3/3/13	Forecast 3/10/13	Forecast 3/17/13	Forecast 3/24/13	16 Weeks Ending
	3/24/13 12/15/12	12/22/12	12/29/12	1/5/13	1/12/13	1/19/13	1/26/13	2/2/13	2/9/13	2/16/13	2/23/13	3/2/13	3/9/13	3/16/13	3/23/13	3/30/13	3/30/13
Receipts																	
Proceeds from Exits	\$ -	\$ - \$	- \$	- \$	- \$	- \$	- \$	- \$	- \$	- \$	- \$	- \$	-	\$ - \$	- \$	-	\$ -
Deal Company Murabahas	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-
Management Fees	-	-	-	250	-	125	-	-	-	-	-	-	172	-	-	250	797
Yield from Deal Companies	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-
All Other		-	-	-	-	-	-	-	-	-	-	-	-	-	-		
Total Receipts		-	-	250	-	125	-	-	-	-	-	-	172	-	-	250	- 797
<u>Disbursements</u> G&A Expenses																	
Atlanta	82	70	6	110	6	147	6	110	6	84	75	92	24	6	213	42	1,081
Hong Kong	25	70	U	110	U	147	U	110	U	04	/3	32	24	U	213	42	25
London	13	1,639	379	713	34	18	12	607	27	14	12	606	12	30	58	952	5,124
Singapore	515	70	9	47	157	38	7	47	104	79	11	47	7	168	23	64	1,392
Point Park Properties	250	250	250	250	150 347	150	150	150	150 288	150	150 247	150	150	150	150	150	2,800
Total G&A Expenses	885	2,028	644	1,120	347	354	175	913	288	328	247	895	192	355	445	1,208	10,423
Staff Expenses																	
Atlanta	151	33	-	151	-	151	33	151	-	144	33	144	-	144	33	144	1,313
Hong Kong	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-
London	-	346	-	-	-	25	346	-	11	-	304	-	11	-	-	304	1,348
Singapore	-	190	-	-	-	190	-	-	-	190	-	-	-	-	266	-	834
Point Park Properties		<u> </u>				<u> </u>		<u> </u>	<u> </u>	<u> </u>	<u>-</u> _				<u> </u>	-	
Total Staff Expenses	151	569	-	151	-	365	379	151	11	334	337	144	11	144	299	448	3,495
Other Expenses																	
Other Total Other Expenses		-	-	-	-	-	-	-	-	-	-	-	-	-	-		
	-			-				-		-	-	-	-	-	-	-	-
Deal Funding & Expenses																	
Private Equity	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-
Real Estate	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-
Infrastructure	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-
Venture Capital	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-
Total Deal Funding & Expenses	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	
Total Disbursements	1,036	2,597	644	1,271	347	719	554	1,064	299	661	584	1,039	204	499	744	1,656	13,918
Operating Cash Flow	(1,036)	(2,597)	(644)	(1,021)	(347)	(594)	(554)	(1,064)	(299)	(661)	(584)	(1,039)	(32)	(499)	(744)	(1,406)	(13,121)
Danturaturia	-																
Restructuring																	
Payroll Adjustments	167	-	398	-	-	80	43	-	277	790	314	-	-	1,990	-	551	4,609
Other		-	-	-	-	-	-	-				-	-		-		
Total Restructuring	167		398			80	43	-	277	790	314		-	1,990		551	4,609
<u>Debt Service</u> Financing Cost - SCB																	_
DIP Interest		_		_	-		_	_			_	_	_	_	_		
DIP Receipts/Payments	_	_	_	_	_		_	_	_	_	_	_	_	_	_		_
Restructuring Accrual / Completion Fees Adjustmen	+																
Total Debt Service	-							-	-	-	-	-	-		-		
Net Cash Flow	(1,203)	(2,597)	(1,041)	(1,021)	(347)	(674)	(597)	(1,064)	(576)	(1,451)	(898)	(1,039)	(32)	(2,489)	(744)	(1,957)	(17,730)
Net Cash Flow	(1,203)	(2,597)	(1,041)	(1,021)	(347)	(0/4)	(597)	(1,004)	(5/6)	(1,451)	(696)	(1,039)	(32)	(2,469)	(744)	(1,957)	(17,730)
Intercompany Transfers																	
Cash Inflows (Source of Cash)	167	390	1,198	550	200	619	1,143	650	477	1,430	1,114	1,000	-	2,190	1,066	2,101	14,293
Cash Outflows (Use of Cash)		-		-	-		, .	-	_	-	´ -	-	_	-	-		-
Total Intercompany Transfers	167	390	1,198	550	200	619	1,143	650	477	1,430	1,114	1,000		2,190	1,066	2,101	14,293
Balances & Liquidity				1 522	1,052	905	850	1,396	982	883	861	1,077	1,038	1,006	707	1,029	4,610
	4,610	3.574	1.367					1,550	JUL	000			1,000	1,000	,		
Beginning Book Balance	4,610 (1,036)	3,574 (2,208)	1,367 156	1,523 (471)			546	(414)	(99)	(22)	216	(39)	(32)	(299)	322	144	(3,437)
	4,610 (1,036) 3,574	3,574 (2,208) 1,367	1,367 156 1,523	(471) 1,052	(147) 905	(54) 850	546 1,396	(414) 982	(99) 883	(22) 861	216 1,077	(39) 1,038	(32) 1,006	(299) 707	322 1,029	144 1,173	(3,437) 1,173
Beginning Book Balance Net Receipts, Disbursements & Transfers Ending Book Balance	(1,036) 3,574	(2,208) 1,367	156 1,523	(471) 1,052	(147) 905	(54) 850	1,396	982	883	861	1,077	1,038	1,006	707	1,029		
Beginning Book Balance Net Receipts, Disbursements & Transfers Ending Book Balance Float/Foreign Exchange	(1,036)	(2,208)	156	(471)	(147)	(54)											
Beginning Book Balance Net Receipts, Disbursements & Transfers Ending Book Balance Float/Foreign Exchange Interbank Transfers	(1,036) 3,574	(2,208) 1,367	156 1,523	(471) 1,052	(147) 905	(54) 850	1,396	982	883	861	1,077	1,038	1,006	707	1,029		(3,437) 1,173
Beginning Book Balance Net Receipts, Disbursements & Transfers Ending Book Balance Float/Foreign Exchange Interbank Transfers Less: Placements Held at Banks	(1,036) 3,574 100	(2,208) 1,367 100 -	156 1,523 100 -	(471) 1,052 100	905 100	(54) 850 100	1,396	100	100	100	1,077 100	1,038	1,006 100	100	1,029 100	1,173	1,173
Beginning Book Balance Net Receipts, Disbursements & Transfers Ending Book Balance Float/Foreign Exchange Interbank Transfers	(1,036) 3,574	(2,208) 1,367 100 -	156 1,523 100 -	(471) 1,052	(147) 905	(54) 850	1,396	100	100	861	1,077	1,038	1,006 100	100	1,029 100	1,173	

12-11076-shl Doc 690-3 Filed 12/04/12 Entered 12/04/12 18:01:19 Exhibit C - DIP Budget Pg 5 of 6

ARCAPITA

Cash Flow Forecast - DEBTOR (Bahrain)
For the Period Ending 3/30/13

(5's in 000's)	12/9/12	Forecast 12/9/12	Forecast 12/16/12	Forecast 12/23/12	Forecast 12/30/12	Forecast 1/6/13	Forecast 1/13/13	Forecast 1/20/13	Forecast 1/27/13	Forecast 2/3/13	Forecast 2/10/13	Forecast 2/17/13	Forecast 2/24/13	Forecast 3/3/13	Forecast 3/10/13	Forecast 3/17/13	Forecast 3/24/13	16 Weeks Ending
(+ ,	3/24/13	12/15/12	12/22/12	12/29/12	1/5/13	1/12/13		1/26/13	2/2/13	2/9/13	2/16/13	2/23/13	3/2/13	3/9/13	3/16/13	3/23/13	3/30/13	3/30/13
Receipts																		
Proceeds from Exits	:	\$ - !	\$ -	\$ -	\$ -	\$ - \$	- \$	- \$	-	\$ -	\$ - \$	- \$	- \$	\$ -	\$ -	\$ -	\$ -	\$ -
Deal Company Murabahas		-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-
Management Fees Yield from Deal Companies		-	-	-	748	-	-	-	-	-	-	-	-	-	-	-	-	748
All Other		-	-	-		-	-	-	-	-	-	-		-	-	-	-	-
Total Receipts	-				748				-				-					748
•	_																	
<u>Disbursements</u> G&A Expenses																		
Bahrain	_	1,399	659	680	251	209	535	215	181	228	463	237	289	302	462	354	563	7,025
Total G&A Expenses		1,399	659	680	251	209	535	215	181	228	463	237	289	302	462	354	563	7,025
Staff Expenses																		
Bahrain	_	89	_	2,381	_	-	145	869	1	_	200	802	1	_	-	50	783	5,322
Total Staff Expenses		89	-	2,381	-	-	145	869	1	-	200	802	1	-	-	50	783	5,322
Other Expenses																		
AHQ Payments		-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-
Other	_	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	
Total Other Expenses		-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-
Deal Funding & Expenses																		
Private Equity		-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-
Real Estate		-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-
Infrastructure		-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-
Venture Capital	_	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-		
Total Deal Funding & Expenses		-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-
Total Disbursements	=	1,488	659	3,061	251	209	679	1,084	183	228	663	1,038	290	302	462	404	1,347	12,347
Operating Cash Flow	-	(1,488)	(659)	(3,061)	497	(209)	(679)	(1,084)	(183)	(228)	(663)	(1,038)	(290)	(302)	(462)	(404)	(1,347)	(11,600)
	_																	
Restructuring																		
Restructuring Fees		5,435	4,601	-	2,205	2,686	122	-	51	5,641	-	122	-	4,811	-	122	4,092	29,888
Critical / Foreign Vendor Payments		-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-
Payroll Adjustments		-	564	113	-	-	-	113	-	833	-	355	-	-	153	-	728	2,860
Other Total Restructuring	-	5,435	5,165	113	2,205	2.686	122	113	51	6,474		477	-	4,811	153	122	4,820	32,748
	-	3,433	3,103	113	2,203	2,080	122	113	31	0,474		4//		4,011	133	122	4,820	32,748
<u>Debt Service</u>																		
Financing Cost - SCB		-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-
DIP Interest & Fees DIP Receipts/Payments		-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-
Restructuring Accrual / Completion Fees Adjustmen	nt	-	-	-		-	-	-	-	-	-	-		-	-	-	-	-
Total Debt Service	_				-				-				-					
	_																	
Net Cash Flow	=	(6,923)	(5,823)	(3,175)	(1,708)	(2,894)	(802)	(1,198)	(234)	(6,702)	(663)	(1,516)	(290)	(5,113)	(615)	(526)	(6,167)	(44,348)
Intercompany Transfers																		
Cash Inflows (Source of Cash)		-	4,150	3,350	1,750	2,950	950	1,250	450	6,450	850	1,550	350	5,050	650	750	6,350	36,850
Cash Outflows (Use of Cash)	_	21	190	180	-	-	213	-	-	-	213	-	-	-	26	266	157	1,265
Total Intercompany Transfers	_	(21)	3,960	3,170	1,750	2,950	737	1,250	450	6,450	637	1,550	350	5,050	624	484	6,193	35,585
Balances & Liquidity																		
Beginning Book Balance		44,951	38,008	36,145	36,140	36,182	36,237	36,173	36,226	36,442	36,190	36,165	36,199	36,258	36,195	36,204	36,162	44,951
Net Receipts, Disbursements & Transfers		(6,943)	(1,863)	30,143	42	56	(64)	50,173	216	(252)	(25)	30,103	60	(63)	30,133	(42)		(8,763)
Ending Book Balance	_	38,008	36,145	36,140	36,182	36,237	36,173	36,226	36,442	36,190	36,165	36,199	36,258	36,195	36,204	36,162	36,188	36,188
-	_																	
Float/Foreign Exchange		400	400	400	400	400	400	400	400	400	400	400	400	400	400	400	-	-
Interbank Transfers		-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-
Less: Placements Held at Banks	-	(35,000) \$ 3,408	(35,000)	(35,000) \$ 1,540	(35,000) \$ 1,582	(35,000) \$ 1,637 \$	(35,000) 1,573 \$	(35,000) 1,626 \$	(35,000)	(35,000) \$ 1,590	(35,000) \$ 1,565 \$	(35,000) 1,599 \$	(35,000)	(35,000)	(35,000) \$ 1,604	(35,000) \$ 1,562		(35,000) \$ 1,188
Ending Bank Balance	_	3,4U8	1,545	ə 1,540	ş 1,582	\$ 1,637 \$	1,5/3 \$	1,626 \$	1,842	ə 1,590	\$ 1,565 \$	1,599 \$	1,658 \$	1,595	ş 1,604	ş 1,562	\$ 1,188	\$ 1,188

12-11076-shl Doc 690-3 Filed 12/04/12 Entered 12/04/12 18:01:19 Exhibit C - DIP Budget Pg 6 of 6

ARCAPITA

Cash Flow Forecast - AIHL (DEBTOR)
For the Period Ending 3/30/13

For the Feriou Enamy 3/30/13	Forecast	Forecast	Forecast	Forecast	Forecast	Forecast	Forecast	Forecast	Forecast	Forecast	Forecast	Forecast	Forecast	Forecast	Forecast	Forecast	16 Weeks
	12/9/12 12/9/12	12/16/12	12/23/12	12/30/12	1/6/13	1/13/13	1/20/13	1/27/13	2/3/13	2/10/13	2/17/13	2/24/13	3/3/13	3/10/13	3/17/13	3/24/13	Ending
	3/24/13 12/15/12	12/22/12	12/29/12	1/5/13	1/12/13	1/19/13	1/26/13	2/2/13	2/9/13	2/16/13	2/23/13	3/2/13	3/9/13	3/16/13	3/23/13	3/30/13	3/30/13
Receipts Proceeds from Exits	\$	- \$ -	\$ - \$	31,750 \$	s - \$	- Ś	_	ė	s -	s -	\$ -	ė	Ś -	\$ - 5	5		\$ 31,750
Deal Company Murabahas	ş	- , -		31,730 ;	, - ,	- ,	-	, .	-	-	-	-	, .		, - ,	-	3 31,730
Management Fees			-	-	-	_	_	_	_	_	_	_	_	-	_	_	_
Yield from Deal Companies			-	_	_	-	-	-	_	_	_	-	_	-	-	-	_
All Other			-	19	-	-	-	-	-	-	-	-	-	-	-	-	19
Total Receipts				31,769		-		-	-	-					-	-	31,769
<u>Disbursements</u>																	
G&A Expenses Total G&A Expenses	-		-	-	-	-	-	-	-	-	-	-	-	-	-	-	-
Staff Expenses																	
Total Staff Expenses			-	-	-	-	-	-	-	-	-	-	-	-	-	-	-
Other Expenses																	
Other	10		-	·	-	-	-	150	-	-	20	-	-	-	-	-	323
Total Other Expenses	10	53	-	-	-	-	-	150	-	-	20	-	-	-	-	-	323
Deal Funding & Expenses																	
Private Equity	17	5 75	-	-	89	-	-	-	39	-	-	-	89	-	-	-	468
Real Estate	1,05	4 750	429	11,425	99	-	-	2,606	-	-	-	-	250	10,000	-	-	26,613
Infrastructure	3,52	5 750	-	1,322	-	-	-	1,234	-	-	-	-	-	-	-	-	6,832
Venture Capital			-	-	1,500	-	-	-	-	-	-	-	-	-	-		1,500
Total Deal Funding & Expenses	4,75	5 1,575	429	12,747	1,688	-	-	3,841	39	-	-	-	339	10,000	-	-	35,414
Total Disbursements	4,85	5 1,628	429	12,747	1,688		-	3,991	39	-	20		339	10,000	-		35,737
Operating Cash Flow	(4,85	5) (1,628)	(429)	19,022	(1,688)	-	-	(3,991)	(39)	-	(20)	-	(339)	(10,000)	-	-	(3,968)
Restructuring																	
Restructuring Fees	1,16	3 510		_	793	_	_	_	793		_	_	793		_	_	4,057
Critical / Foreign Vendor Payments	1,10	. 510			755				,,,,				755				4,037
Ordinary Course Professionals		_	_		_	_	_	_			_	_	_	_	_	_	_
Payroll Adjustments			-	_	-	_	_	_	_	_	_	_	_	-	-	_	
Other			-	-	-	_	_	_	_	_	_	_	_	-	_	_	_
Total Restructuring	1,16	3 510	-	-	793	-	-	-	793	-	-	-	793	-	-		4,057
Debt Service																	
Financing Cost - SCB			-	6,759	-	-	-	500	-	-	-	500	-	-	-	1,000	8,759
DIP Interest & Fees	2,00		-	496	-	-	750	937	-	-	-	1,104	-	-	-	1,380	8,416
DIP Receipts/Payments	(25,00	0) (100,000)	-	31,750	-	-	(25,000)	-	-	-	-	-	-	-	-	-	(118,250)
Restructuring Accrual / Completion Fees Adjustment			-	-	-	-	-	-	-	-	-	-	-	-	-	27,656	27,656
Total Debt Service	(23,00	0) (98,250)	-	39,005			(24,250)	1,437		-		1,604	-	-		30,036	(73,419)
Net Cash Flow	16,97	7 96,112	(429)	(19,982)	(2,481)	-	24,250	(5,427)	(832)	-	(20)	(1,604)	(1,132)	(10,000)	-	(30,036)	65,394
Intercompany Transfers																	
Cash Inflows (Source of Cash)			-	-	-	-	-	_	-	_	_	-	-	-	-		
Cash Outflows (Use of Cash)	14	7 4,350	4,367	2,300	3,150	1,357	2,393	1,100	6,927	2,067	2,664	1,350	5,050	2,814	1,550	8,293	49,878
Total Intercompany Transfers	(14		(4,367)	(2,300)	(3,150)	(1,357)	(2,393)	(1,100)	(6,927)	(2,067)	(2,664)	(1,350)	(5,050)	(2,814)	(1,550)	(8,293)	(49,878)
Balances & Liquidity																	
Beginning Book Balance	1,80	7 18,637	110,399	105,603	83,320	77,689	76,332	98,189	91,662	83,903	81,836	79,152	76,198	70,016	57,202	55,652	1,807
Net Receipts, Disbursements & Transfers	1,80		(4,796)	(22,282)	(5,631)	(1,357)	21,857	(6,527)	(7,760)	(2,067)	(2,684)	(2,954)	(6,182)	(12,814)	(1,550)	(38,329)	1,807
Ending Book Balance	18,63		105,603	83,320	77,689	76,332	98,189	91,662	83,903	81,836	79,152	76,198	70,016	57,202	55,652	17,323	17,323
• • • • • • •		-,	,	,-	,	-,	,		,	. ,	-, ,-	.,	-,			,	
Float/Foreign Exchange			-	-	-	-	-	-	-	-	-	-	-	-	-	-	-
Interbank Transfers			-	-	-	-	-	-	-	-	-	-	-	-	-	-	-
Less: Placements Held at Banks	¢ 10.63	7 \$ 110,399	\$ 105,603 \$	83,320	77,689 \$	76,332 \$	98,189	\$ 91,662	\$ 83,903	\$ 81,836	\$ 79,152	\$ 76,198	\$ 70,016	\$ 57,202 \$	5 55,652 \$	17,323	\$ 17,323
Ending Bank Balance	\$ 18,63	, à 110,399	3 105,003 S	03,320 \$, //,069 \$	/0,332 \$	30,109	91,00Z	و معروه د	9 61,63b	\$ 19,152	→ /0,198	\$ /U,U16	\$ 57,202	, 55,052 \$	17,323	ş 17,523

Exhibit D

Makuch Declaration

GIBSON, DUNN & CRUTCHER LLP

Michael A. Rosenthal (MR-7006) Craig H. Millet (admitted *pro hac vice*) Matthew J. Williams (MW-4081) Joshua Weisser (JW-0185) 200 Park Avenue New York, New York 10166-0193 Telephone: (212) 351-4000

Telephone: (212) 351-4000 Facsimile: (212) 351-4035

Attorneys for the Debtors and Debtors in Possession

UNITED STATES BANKRUPTCY COURT

SOUTHERN DISTRICT OF NEW YORK	
	X
IN RE:	Chapter 11
ARCAPITA BANK B.S.C.(c), et al.,	: Case No. 12-11076 (SHL)
Debtors.	: Jointly Administered
	:

DECLARATION OF JOHN MAKUCH IN SUPPORT OF
DEBTORS' MOTION FOR THE ENTRY OF INTERIM AND FINAL ORDERS
PURSUANT TO 11 U.S.C. §§ 105, 362, 363(b)(1), 363(m), 364(c)(1), 364(c)(2), 364(c)(3),
AND 364(e) AND BANKRUPTCY RULES 4001 AND 6004 (I) AUTHORIZING
DEBTORS (A) TO ENTER INTO AND PERFORM UNDER DIP AGREEMENT, AND
(B) TO OBTAIN CREDIT ON A SECURED SUPERPRIORITY BASIS, (II)
SCHEDULING FINAL HEARING PURSUANT TO BANKRUPTCY RULES
4001(b) AND (c) AND (III) GRANTING RELATED RELIEF

I, John Makuch, hereby declare as follows:

1. I am a Managing Director of Alvarez & Marsal North America, LLC (together with employees of its affiliates (all of which are wholly-owned by its parent company and employees), its wholly owned subsidiaries, and its independent contractors collectively, "A&M"), a professional services firm that has been retained by the Debtors (defined below) in the above-captioned chapter 11 cases (the "Chapter 11 Cases"). My business address is 3424 Peachtree Road NE, Suite 1500, Atlanta, Georgia 30326. I am one of the A&M professionals in

12-11076-shl Doc 690-4 Filed 12/04/12 Entered 12/04/12 18:01:19 Exhibit D - Makuch Declaration Pg 3 of 10

charge of A&M's engagement by Arcapita Bank B.S.C.(c) ("Arcapita") and its affiliated debtors and debtors in possession (collectively, with Arcapita, the "Debtors" and each a "Debtor").

- A&M's engagement by Arcapita, I am duly authorized to make this Declaration on behalf of A&M in support of the Debtors' Motion (the "*DIP Motion*")¹ for entry of interim and final orders pursuant to 11 U.S.C. §§ 105, 362, 363(b)(1), 363(m), 364(c)(1), 364(c)(2), 364(c)(3), and 364(e) and Bankruptcy Rules 4001 and 6004 (i) authorizing the Debtors (a) to enter into and perform under the DIP Agreement, and (b) to obtain credit on a secured superpriority basis, (ii) scheduling a final hearing pursuant to Bankruptcy Rules 4001(b) and (c) and (iii) granting related relief, all under the terms and conditions set forth in the DIP Motion and the DIP Agreement annexed as Exhibit B to the DIP Motion. The purpose of this Declaration is to summarize the Debtors' current available liquidity as well as their need for additional cash moving forward, both on an interim and final basis. I have reviewed the DIP Motion and the DIP Agreement and, as a result, have a good understanding of the DIP Transaction structure.
- 3. Except as otherwise indicated herein, all facts set forth in this Declaration are based upon my personal knowledge, information learned from my review of relevant documents and information supplied to me by A&M professionals who are acting under my supervision. If called upon to testify, I could and would testify competently to the facts set forth herein.

2

All capitalized terms used but otherwise not defined herein shall have the meanings ascribed to them in the DIP Motion.

STATEMENT

I. A&M's Budget Work during the Chapter 11 Cases

- 4. The Debtors have been in bankruptcy for approximately eight months. Prior to the date hereof, the Debtors have not sought post-petition financing. Instead, they have managed their businesses and funded general and administrative expenses, staff costs, professional fees and deal funding costs (as described below) with cash on hand.
- 5. I and A&M professionals supervised by me have developed an in-depth understanding of the Debtors' revenues and expenditures in connection with our work structuring (i) Interim Cash Management Budgets and (ii) the DIP Budget (each, as defined below). In my opinion, this work makes me uniquely qualified to evaluate the adequacy of the proposed liquidity to be made available under the DIP Transaction.
- the Debtors' Motion for Interim and Final Orders (A) Authorizing Debtors To (I) Continue

 Existing Cash Management System, Bank Accounts, and Business Forms and (II) Continue

 Ordinary Course Intercompany Transactions; and (B) Granting an Extension of Time To

 Comply with the Requirements of Section 345(b) of the Bankruptcy Code [Dkt. No. 12]

 (the "Cash Management Motion"). Prior to such filing, I reviewed the Cash Management

 Motion. To date, the Court has entered ten interim orders in connection with the Cash

 Management Motion. Each order provides the Debtors with authority to make intercompany

 transfers in accordance with the interim cash budget annexed thereto (each, an "Interim Cash

 Management Budget" and collectively, the "Interim Cash Management Budgets"). Each

 Interim Cash Management Budget outlines expected cash receipts and disbursements for the succeeding interim period. I have reviewed and approved each Interim Cash Management

12-11076-shl Doc 690-4 Filed 12/04/12 Entered 12/04/12 18:01:19 Exhibit D - Makuch Declaration Pg 5 of 10

Budget prior to its submission to the Court. In addition, where necessary, I and my colleagues have engaged in substantive discussions with professionals for the official committee of unsecured creditors in the Chapter 11 Cases and the joint provisional liquidators in the Cayman Islands liquidation of AIHL (the "*Cayman Proceedings*") in connection with specific contested payments reflected in the Interim Cash Management Budgets.

- 7. The Interim Cash Management Budgets reflect a bottom up analysis of projected cash inflows and outflows. The budgets broadly project: (i) receipts from deal exits, Murabaha repayments, yields, management fees and other sources; (ii) disbursements for staff expenses; (iii) disbursements for general and administrative expenses; (iv) disbursements for deal funding; (v) disbursements for restructuring fees, including professional fees and expenses in the Chapter 11 Cases and the Cayman Proceedings; and (vi) disbursements for other miscellaneous items. Each Interim Cash Management Budget segregates receipts and disbursements between non-Debtor and Debtor entities and differentiates between the receipts and disbursements of AIHL and those of the other Debtors.
- 8. To draft each Interim Cash Management Budget, A&M professionals supervised by me have worked with designated individuals in each Arcapita office. Designated investment team members, for example, have assisted with forecasting investment receipts and deal fundings. In addition, A&M professionals have periodically solicited projections of fees and expenses from the other retained professionals and have tracked fee statements and applications to ensure cost forecasts are as accurate as possible. Finally, A&M has established procedures to review proposed cash disbursements.
- 9. <u>DIP Budget</u>. In addition to the Interim Cash Management Budgets, A&M professionals have prepared, with the assistance of Arcapita personnel, a weekly cash flow

12-11076-shl Doc 690-4 Filed 12/04/12 Entered 12/04/12 18:01:19 Exhibit D - Makuch Declaration Pg 6 of 10

forecast through the end of the first quarter of 2013, by which time the Debtors project to emerge from chapter 11 (the "*DIP Budget*"). The DIP Budget (which is annexed to the DIP Motion as Exhibit C) reflects the same types of receipts and disbursements as are in the Interim Cash Management Budgets. I supervised preparation of the DIP Budget. In my opinion, the DIP Budget represents a reasonable projection of cash flows upon the assumption that the Debtors will continue to manage cash in a careful and responsible manner, utilizing their business judgment to minimize cash uses while protecting and maximizing asset values.

II. The Debtors' Current Liquidity Position

- 10. The Interim Cash Management Budgets and DIP Budget evidence the substantial costs the Debtors have incurred and expect to incur during the Chapter 11 Cases. Deal funding expenses used to support the value of the Debtors' assets comprise a large percentage of these expenditures. Since the Petition Date, the Debtors have incurred approximately \$137.0 million in total cash costs. Of that amount, approximately \$63.2 million (or 46%) comprised deal funding costs. Total projected disbursements for the period covered by the Tenth Interim Cash Management Order (through December 15) equal approximately \$18.6 million. Of that amount, approximately 22% comprises deal funding expenses.
- 11. As a result of these expenditures, and despite the Debtors' diligent cash management efforts, the Debtors' liquidity has decreased. The Debtors' available cash on hand, including non-Debtor subsidiary cash, as of the commencement of the Chapter 11 Cases (the "*Petition Date*"), was approximately \$120.1 million. By contrast, as of November 17, 2012, the Debtors had \$19.0 million of cash and cash equivalents and the non-Debtor subsidiaries held an additional \$4.6 million, yielding total cash and cash equivalents of \$23.6 million.

12-11076-shl Doc 690-4 Filed 12/04/12 Entered 12/04/12 18:01:19 Exhibit D - Makuch Declaration Pg 7 of 10

- 12. Excluded from the calculated amounts above is \$35 million held in placement accounts in Bahrain. I understand that this cash may not be accessed without prolonged litigation against parties which may not have minimum contacts with the United States. I also understand that the Debtors' attempts to negotiate for release of this cash have been unsuccessful. Accordingly, A&M professionals excluded the placements accounts as a source of cash in the DIP Budget. For similar reasons and to avoid confusion, A&M professionals have segregated this cash from "available cash" in each of the monthly operating reports filed in the Chapter 11 Cases.
- assets sales which have either occurred, are likely to occur and/or are under contract. Because the DIP Budget is intended to represent a conservative view of future cash flows, the budget does not reflect a large number of sales. Notably, I understand that various Arcapita Group Investments and/or assets are being marketed for sale or may be marketed for sale between now and March 2013. Potential resulting sale proceeds, however, are not reflected in the DIP Budget because consummation of such sales is subject to a myriad of issues outside of the Debtors' control including: (i) favorable market conditions; (ii) the ability to reach an acceptable agreement with a potential purchaser on terms or conditions for the sale of an asset or investment; and (iii) the ability of a potential purchaser to close on a purchase within an acceptable timeframe.

III. The Need for Additional Liquidity

14. <u>The Debtors' General Cash Needs</u>. Based on the DIP Budget, I estimate that current available cash resources are insufficient for the Debtors to operate in chapter 11 and to fund further investments in Arcapita Group Investments, which as I understand it, comprise

12-11076-shl Doc 690-4 Filed 12/04/12 Entered 12/04/12 18:01:19 Exhibit D - Makuch Declaration Pg 8 of 10

the actual value of the estates. As noted above, as of November 17, 2012, the Debtors and their non-Debtor subsidiaries had access to approximately \$23.6 million in cash and cash equivalents. If the Debtors enter into the DIP Transaction and, thereby, obtain an additional \$125 million to \$150 million of liquidity, the Debtors will have between \$148.6 million and \$173.6 million of total liquidity.

- subsidiaries will require \$153.7 million to operate from November 18, 2012 (the beginning of the forecast period in the DIP Budget) through March 30, 2013. The \$153.2 million represents the net cash outflow amount, after \$33.3 million of cash receipts, relating to the following expenditures: (i) \$19.5 million for general and administrative expenses; (ii) \$10.4 million in staff expenses; (iii) \$35.4 million of deal funding expenses; (iv) \$72.7 million of restructuring expenses, separation payments and completion fees; (v) \$8.4 million of fees and costs relating to the Proposed Transaction; (vi) \$8.8 million in interest and costs for SCB; and (vii) a \$31.8 million pay down of the DIP balance (using proceeds from exits). The current availability of \$23.6 million is insufficient to fund the foregoing.
- 16. Based on the foregoing analysis, the Debtors, with counsel from their advisors, including A&M, have determined that approximately \$150 million of additional financing will be sufficient to sustain their businesses during the Chapter 11 Cases and maximize the value of their assets. I support this assessment.
- 17. The Debtors' Immediate Need for Cash. The Debtors, with counsel from A&M professionals, have determined that the Debtors require \$25 million (referred to in the DIP Motion as the "*Interim Amount*") immediately in order to satisfy the Debtors' short-term financing needs and that approval of such Interim Amount pending entry of the final order is

12-11076-shl Doc 690-4 Filed 12/04/12 Entered 12/04/12 18:01:19 Exhibit D - Makuch Declaration Pg 9 of 10

necessary to avoid immediate irreparable harm to the estates. I support this assessment as well. This determination was based on A&M's review and monitoring of weekly cash budgets (itemizing receipts and disbursements) through the Interim Hearing and the Final Hearing. Moreover, pursuant to the DIP Agreement, the minimum amount the Debtors can draw with respect to the DIP Facility is \$25 million.

- expenditures including restructuring costs, DIP fees, deal funding and expenses and debt service payments to SCB. If the Debtors do not receive the Interim Amount, the damage to the viability of the Debtors' businesses and the endurance of the Debtors' operations will be imminent and irreparable. The Debtors' current cash on hand is insufficient to maintain their businesses, to satisfy short term financing needs, including restructuring costs, or to fund other investment related expenses. Additionally, the Debtors will be unable to adequately fund Arcapita Group operations and thus unable to maintain their equity position in, and the financial health of, Arcapita Group Investments. This could result in certain Arcapita Group Investments being forced to cease operations, which would eviscerate Arcapita's equity value in such entities and threaten the Debtors' ability to reorganize. Thus, it is my opinion that, the Debtors will suffer immediate and irreparable harm if the interim relief requested in the DIP Motion, upon entry of an Interim Order, is not granted.
- 19. Consequently, based upon my assessment of the Debtors' liquidity position and financing needs and upon my review of the DIP Agreement, it is my opinion that:
 - i. the Debtors require additional financing in order to continue to fund Arcapita Group operations and bridge the Debtors to emergence;
 - ii. the Debtors require the Interim Amount immediately to avoid irreparable harm to the estates;

12-11076-shl Doc 690-4 Filed 12/04/12 Entered 12/04/12 18:01:19 Exhibit D - Makuch Declaration Pg 10 of 10

- iii. the DIP Agreement represents the best available offer to provide the necessary financing; and
- iv. as a result of the foregoing, the relief requested in the DIP Motion is in the best interests of the Debtors, their estates, creditors and other stakeholders.

Pursuant to 28 U.S.C. § 1746, I declare under penalty of perjury under the laws of the United States of America that the foregoing is true and correct.

Executed on this 4th day of December, 2012.

/s/ John Makuch

John Makuch, Managing Director, Alvarez & Marsal North America, LLC

Exhibit E

Parkhill Declaration

GIBSON, DUNN & CRUTCHER LLP

Michael A. Rosenthal (MR-7006) Craig H. Millet (admitted pro hac vice) Matthew J. Williams (MW-4081) Joshua Weisser (JW-0185) 200 Park Avenue New York, New York 10166-0193 Telephone: (212) 351-4000

Telephone: (212) 351-4000 Facsimile: (212) 351-4035

Attorneys for the Debtors and Debtors in Possession

UNITED STATES BANKRUPTCY COURT

SOUTHERN DISTRICT OF NEW YORK	
	X
IN RE:	: : Chapter 11
ARCAPITA BANK B.S.C.(c), et al.,	: Case No. 12-11076 (SHL)
Debtors.	Jointly Administered
	·

DECLARATION OF HOMER PARKHILL IN SUPPORT OF
DEBTORS' MOTION FOR THE ENTRY OF INTERIM AND FINAL ORDERS
PURSUANT TO 11 U.S.C. §§ 105, 362, 363(b)(1), 363(m), 364(c)(1), 364(c)(2), 364(c)(3),
AND 364(e) AND BANKRUPTCY RULES 4001 AND 6004 (I) AUTHORIZING
DEBTORS (A) TO ENTER INTO AND PERFORM UNDER DIP AGREEMENT, AND
(B) TO OBTAIN CREDIT ON A SECURED SUPERPRIORITY BASIS, (II)
SCHEDULING FINAL HEARING PURSUANT TO BANKRUPTCY RULES
4001(b) AND (c) AND (III) GRANTING RELATED RELIEF

- I, Homer Parkhill, hereby declare as follows:
- 1. I am a Managing Director of Rothschild Inc. (together with its affiliate N M Rothschild & Sons Limited, "*Rothschild*"), a financial advisory services and investment banking firm. The principal office of Rothschild Inc. is located at 1251 Avenue of the Americas, 51st Floor, New York, New York 10020. N M Rothschild & Sons Limited has its principal office at New Court, St. Swithin's Lane, London, UK, EC4N 8AL. I am in charge of

12-11076-shl Doc 690-5 Filed 12/04/12 Entered 12/04/12 18:01:19 Exhibit E - Parkhill Declaration Pg 3 of 9

Rothschild's engagement by Arcapita Bank B.S.C.(c) ("*Arcapita*") and its affiliated debtors and debtors in possession (collectively, with Arcapita, the "*Debtors*" and each, a "*Debtor*").

- As a Managing Director of Rothschild and the person responsible for Rothschild's engagement by Arcapita, I am duly authorized to make this Declaration on behalf of Rothschild in support of the Debtors' Motion (the "DIP Motion")¹ for entry of interim and final orders pursuant to 11 U.S.C. §§ 105, 362, 363(b)(1), 363(m), 364(c)(1), 364(c)(2), 364(c)(3), and 364(e) and Bankruptcy Rules 4001 and 6004 (i) authorizing the Debtors (a) to enter into and perform under the DIP Agreement, and (b) to obtain credit on a secured superpriority basis, (ii) scheduling a final hearing pursuant to Bankruptcy Rules 4001(b) and (c) and (iii) granting related relief, all under the terms and conditions set forth in the DIP Motion and the DIP Agreement annexed as Exhibit B to the DIP Motion. The purpose of this Declaration is to summarize Rothschild's efforts to solicit and identify potential DIP providers and the negotiation process leading to the Debtors' entry into the Fortress Commitment Letter (as defined below). I was actively involved in the solicitation process and have reviewed the DIP Motion and the DIP Agreement and, as a result, have a good understanding of the structure and terms of the DIP Transaction.
- 3. Except as otherwise indicated herein, all facts set forth in this Declaration are based upon my personal knowledge, information learned from my review of relevant documents, including the DIP Agreement and the Fortress Commitment Letter, and information supplied to me by Rothschild professionals who are acting under my supervision. If called upon to testify, I could and would testify competently to the facts set forth herein.

All capitalized terms used but otherwise not defined herein shall have the meanings ascribed to them in the DIP Motion.

12-11076-shl Doc 690-5 Filed 12/04/12 Entered 12/04/12 18:01:19 Exhibit E - Parkhill Declaration Pg 4 of 9

- 4. Previously, during the Chapter 11 Cases, I submitted a declaration in connection with the *Debtors' Motion for Entry of an Order Authorizing the Debtors To Enter into a Financing Commitment Letter and Incur Related Fees, Expenses and Indemnities* [Dkt. No. 513] (the "*Original Commitment Letter Motion*"). My declaration in support of the Original Commitment Letter Motion was annexed thereto as <u>Exhibit D</u> (the "*Parkhill Commitment Letter Declaration*").
- 5. Prior to submitting the Parkhill Commitment Letter Declaration, I reviewed and analyzed monthly interim budgets and the DIP Budget prepared by the Debtors' financial advisor, Alvarez & Marsal North America, LLC ("A&M"), during the Chapter 11 Cases. Based on that review, in my previous declaration, I concluded that then current liquidity was insufficient to fund the Debtors through their projected emergence from chapter 11. Based on my further review of updated budgets, I still believe this to be the case. By contrast, if approved and consummated fully, the proposed DIP Transaction will provide the Debtors with sufficient liquidity to pay projected administrative expenses and deal funding costs, enabling the Debtors to preserve and to monetize their assets at the best available value.

SOLICITATION PROCESS

6. In connection with the DIP Transaction, Rothschild professionals supervised by me engaged in an extensive multi-stage solicitation. The result of that solicitation process is the Debtors' entry into the Fortress Commitment Letter and successful negotiation of the terms of the DIP Agreement. The Debtors' and their advisors' mutual goal with respect to the solicitation process was to obtain sufficient financing to fund projected deal funding costs and the Debtors' emergence from chapter 11 (taking into account other sources of liquidity), to do so on the best terms available, to ensure the selected providers of debtor in possession

12-11076-shl Doc 690-5 Filed 12/04/12 Entered 12/04/12 18:01:19 Exhibit E - Parkhill Declaration Pg 5 of 9

financing had the wherewithal to consummate the DIP Transaction and ultimately, to maximize value for the benefit of the Debtors and their estates.

- 7. During the initial phase of the solicitation process, Rothschild professionals commenced a rigorous screening process to identify potential DIP providers. I reviewed the results of the screening process. At my direction, Rothschild professionals selected potential DIP providers based on a number of factors, including their (i) ability to provide Shari'ah-compliant financing, (ii) experience providing DIP financings and (iii) familiarity with the Arcapita Group. Additionally, Rothschild professionals permitted third-parties, who expressed unsolicited interest, to participate in the process.
- 8. On or around September 9, 2012, the Debtors received two separate offers for post-petition financing. After examining the submitted proposals, the Debtors, with advice from Rothschild and the Debtors' other professionals, selected Silver Point Finance, LLC to serve as investment agent and DIP Participant (the "Original DIP Provider" and the agreement between the Debtors and the Original DIP Provider, the "Original Commitment Letter"). On September 25, the Debtors filed the Original Commitment Letter Motion, by which they sought Court authority to enter into the Original Commitment Letter. I am generally knowledgeable regarding the terms of the Original Commitment Letter.
- 9. A hearing on the Original Commitment Letter Motion took place on October 9, 2012 (the "*Initial Hearing*"). Prior to ruling, the Court adjourned the Initial Hearing to afford the parties an opportunity to attempt to reach a consensual resolution of disputed issues and to address the Court's stated concerns with respect to the Original Commitment Letter.

² An in depth summary of the solicitation process leading to the selection of the Original DIP Participant can be found in the Parkhill Commitment Letter Declaration.

12-11076-shl Doc 690-5 Filed 12/04/12 Entered 12/04/12 18:01:19 Exhibit E - Parkhill Declaration Pg 6 of 9

- 10. Both prior to and after the adjournment of the Initial Hearing, the Debtors, as advised by Rothschild professionals and other advisors, continued negotiations and extensive due diligence with multiple parties, including the Original DIP Provider, in an effort to obtain a firm commitment for additional liquidity on the best available terms. To that end, over the next few weeks, the Debtors administered an open auction for the DIP Transaction. The auction was ultimately a success. The Debtors received, reviewed, compared and negotiated various competing financing proposals. The Committee, the JPLs and their respective advisors also performed key roles in the auction. The Debtors, I and other Rothschild professionals consulted with Committee and JPL advisors and solicited their input regarding the terms of the proposed transaction documents.
- 11. In connection with the solicitation and negotiation process, I, and Rothschild professionals supervised by me, were actively involved in the negotiation of financing proposals, including the Fortress Commitment Letter. Specifically, we compared the terms of the various offers received by the Debtors and directly negotiated with potential DIP providers. Finally, I and Rothschild professionals supervised by me played a key role in the due diligence process.
- Corp. ("Fortress") as the DIP Participant. On or around October 18, 2012, the Debtors received an initial proposal from Fortress to provide debtor in possession financing (the "Fortress Commitment" and the commitment letter in respect thereof, as later revised, the "Fortress Commitment Letter"). From that date until the hearing on the Fortress Commitment Letter, Fortress and the Debtors engaged in extensive good faith arms' length negotiations relating to the terms of the financing. Concurrently, the Debtors continued to (a) engage in active negotiations

12-11076-shl Doc 690-5 Filed 12/04/12 Entered 12/04/12 18:01:19 Exhibit E - Parkhill Declaration Pg 7 of 9

with other potential DIP providers and (b) regularly confer with advisors to the Committee, the JPLs and Standard Chartered Bank regarding the terms of the competing proposals.

- Motion for Entry of an Order Authorizing the Debtors To Enter into a Financing Commitment

 Letter and Incur Related Fees, Expenses and Indemnities in Connection Therewith [Dkt. No.
 610] (the "Commitment Letter Motion Supplement," and together with the Original

 Commitment Letter Motion, the "Commitment Letter Motion"). An updated and revised version of the Fortress Commitment Letter was annexed to the Commitment Letter Motion Supplement.

 The Debtors filed the Commitment Letter Motion Supplement seeking Court authority to enter into the Fortress Commitment Letter in anticipation of a hearing on November 5, 2012.
- 14. The Debtors' efforts to negotiate the terms of the DIP Transaction continued after the filing of the Commitment Letter Motion Supplement. On the evening of November 1, 2012, the Debtors and their advisors received an alternative financing offer from a competing potential DIP provider. Immediately, the Debtors, acting with advice of their advisors, including me, engaged in a dual track solicitation, preparing for a hearing on approval of the Fortress Commitment Letter, on the one hand, and commencing substantive negotiations with the other potential DIP provider, on the other hand. It is my understanding that the Debtors and the Committee agreed to adjourn the November 5, 2012 hearing and provide potential DIP providers with the opportunity to submit their respective "last and best" offers by 12:00 p.m. (prevailing U.S. Eastern Time) on November 6, 2012.
- 15. The Debtors considered the "last and best" offers received during the adjournment, and determined that the terms of a revised offer from Fortress, improving upon the originally filed Fortress Commitment Letter, was superior to any other offer received by the

12-11076-shl Doc 690-5 Filed 12/04/12 Entered 12/04/12 18:01:19 Exhibit E - Parkhill Declaration Pg 8 of 9

Debtors. Whereas the filed Fortress Commitment Letter provided for a reduction in the initial commitment from \$100 million to \$80 million, during the adjournment, Fortress agreed to forgo that mandatory reduction. As a result, the Debtors elected to move forward with the Fortress Commitment Letter. At the November 7, 2012 hearing on the Commitment Letter Motion, the Court authorized the Debtors' entry into the Fortress Commitment Letter. Just prior to the hearing, it is my understanding that Fortress agreed to increase the initial commitment (not subject to confirmatory due diligence) under the Fortress Commitment Letter to \$125 million, subject to a \$25 million commitment reduction after 120 days. I view this as a major concession by Fortress.

- Agreement as well as the terms of other offers received by the Debtors during the financing solicitation process. In my opinion, the Fortress Commitment Letter and DIP Agreement represents the best financing proposal available and is the most favorable to these estates and the Debtors' principal restructuring goals, as it provides, *inter alia*:
 - the Debtors with a firm commitment not subject to due diligence contingencies or credit committee approval of \$125 million DIP financing;
 - the possibility of upsizing the Fortress Commitment to \$150 million, subject to confirmatory due diligence;
 - the Debtors with the opportunity to seek out alternative proposals; and
 - a lower commitment fee, a lower profit and a substantially lower unused line fee than those contemplated in the Original Commitment Letter.
- 17. In my opinion, based on my review of the various proposals, the DIP Agreement is the best offer available for post-petition financing in the Chapter 11 Cases. The competing proposals were either (i) for a significantly lesser amount which, the Debtors' advisors, including Rothschild, determined would not be sufficient to preserve estate values, (ii)

12-11076-shl Doc 690-5 Filed 12/04/12 Entered 12/04/12 18:01:19 Exhibit E - Parkhill Declaration Pg 9 of 9

on less favorable and more restrictive terms, or (iii) more conditional, resulting in greater execution risk.

18. As a result of the foregoing, I believe that the relief sought in the DIP Motion is in the best interests of the Debtors, their estates, creditors and other stakeholders.

Pursuant to 28 U.S.C. § 1746, I declare under penalty of perjury under the laws of the United States of America that the foregoing is true and correct.

Executed on this 4th day of December, 2012.

/s/ Homer Parkhill

Homer Parkhill Managing Director, Rothschild Inc.

Exhibit F

Investment Agency Agreement

DATED [] 2012

ARCAPITA INVESTMENT HOLDINGS LIMITED

- and -

FORTRESS CREDIT CORP.

as Investment Agent

- and -

FORTRESS CREDIT CORP.

as Security Agent

- and -

FORTRESS CREDIT CORP.

as Arranger

- and -

THE GUARANTORS AS SPECIFIED IN PART A OF SCHEDULE 1

- and -

THE PARTICIPANTS AS SPECIFIED IN PART B OF SCHEDULE 1

INVESTMENT AGENCY AGREEMENT

12-11076-shl Doc 690-6 Filed 12/04/12 Entered 12/04/12 18:01:19 Exhibit F - Investment Agency Agreement Pg 3 of 34

CONTENTS

CLAUSE		PAGE
1.	DEFINITIONS AND INTERPRETATION	1
2.	APPOINTMENT OF THE INVESTMENT AGENT AND SECURITY AGENT	· 3
3.	PARTICIPATION IN THE PURCHASE CONTRACT	8
4.	PREPAYMENT AND INCREASED COSTS	9
5.	PAYMENTS BY THE PARTICIPANTS	10
6.	PAYMENTS BY THE INVESTMENT AGENT	
7.	REFUND, SHARING AND FURTHER PAYMENTS	12
8.	AMENDMENTS; DEFAULTING PARTICIPANT	15
9.	SECURITY AGENT	17
10.	COMMISSIONS, COSTS AND EXPENSES	18
11.	ASSIGNMENT	18
12.	GENERAL	21
13.	NOTICES	23
14.	CONFIDENTIALITY	
15.	GOVERNING LAW	25
16.	DISPUTE RESOLUTION	25
SCHEE	OULE 1 THE PARTICIPANTS AND GUARANTORS	27
SCHEE	OULE 2 FORM OF TRANSFER CERTIFICATE	28
EXECU	JTION PAGE OF INVESTMENT AGENCY AGREEMENT	31

THIS INVESTMENT AGENCY AGREEMENT is dated [] 2012.

BETWEEN:

- (1) **Arcapita Investment Holdings Limited,** an exempted company incorporated with limited liability under the laws of the Cayman Islands with company number 78594, as debtor-in-possession under chapter 11 of title 11 of the United States code in a case in the Bankruptcy Court for the Southern District of New York ("**AIHL**"); and
- (2) Fortress Credit Corp. in its capacity as Investment Agent (the "Investment Agent").
- (3) The Guarantors as specified below;
- (4) **The Participants** as defined below;
- (5) **Fortress Credit Corp.** as security agent for and on behalf of the Participants (the "**Security Agent**"); and
- (6) **Fortress Credit Corp.** in its capacity as arranger (the "**Arranger**").

RECITALS

- A By a master murabaha agreement (the "Master Murabaha Agreement") dated on or about the date of this Agreement, a US Dollar term murabaha facility has been made available to AIHL subject to the terms and conditions of the Finance Documents.
- B The Participants have agreed to appoint the Investment Agent as their agent in the provision of that facility in accordance with the following terms and conditions.

IT IS AGREED:

1. DEFINITIONS AND INTERPRETATION

1.1 Definitions

Terms defined in the Master Murabaha Agreement shall, unless otherwise defined in this Agreement or the context does not permit, have the same meanings when used in this Agreement and, in addition:

"Condor Agreements" means:

- (a) the Letter of Understanding dated on or about the date of this Agreement and made between Condor Trading Limited and AIHL; and
- (b) the letter from Condor Trading Limited to AIHL dated on or about the date of this Agreement relating to such Letter of Understanding.

[&]quot;Agent Professionals" has the meaning given in clause 2.1(d).

[&]quot;Agents" means the collective reference to the Investment Agent and the Security Agent.

12-11076-shl Doc 690-6 Filed 12/04/12 Entered 12/04/12 18:01:19 Exhibit F - Investment Agency Agreement Pg 5 of 34

- "Contribution" means in relation to a Participant, the contributions (in an aggregate amount outstanding at any time not to exceed the amount set forth opposite its name in Part B of Schedule 1) to be placed by a Participant with the Investment Agent pursuant to clause 3.1 as the same may be increased or decreased by assignments or transfers in accordance with the provisions of clause 11.2.
- "Discharged Amount" has the meaning given in clause 7.4(a)(i).
- "Excess Amount" has the meaning given in clause 7.1.
- "Facility Commitment" means, with respect to any Participant at any time, such Participant's Relevant Percentage of the Facility Limit at such time.
- "Guarantors" means those entities listed in Part A of Schedule 1, and any Additional Guarantors from time to time party to a Guarantee.
- "Indemnified Costs" has the meaning given in clause 2.2(a).
- "Investment" means, at any time, the aggregate of the Contributions of all Participants at such time (which shall not exceed the Facility Limit at any time).
- "Majority Participants" means a group of Participants the aggregate of whose Participations exceed 50% of all Participations or in the event that no Contributions have been made a group of Participants the aggregate of whose Relevant Percentages exceed 50% of all Contributions to be made.
- "Participants" means those banks and financial institutions listed in Part B of Schedule 1 or any assignee or transferee which has become a Participant in accordance with clause 11.2.
- "Participation" means at any time, in relation to a Participant and save as otherwise provided in this Agreement, the aggregate amount of the Contributions actually made by it under this Agreement at such time, as the same may be increased or decreased by assignments or transfers in accordance with the provisions of clause 11.2.
- "Related Parties" has the meaning given in clause 2.1(d).
- "Relevant Payment" has the meaning given in clause 7.1.
- "Relevant Percentage" means in relation to a Participant, the percentage set opposite its name in Part B of Schedule 1 as such percentage may change from time to time in order to reflect the actual percentage of its Participation to the aggregate amount of all Participations.
- "Relevant Receipt" has the meaning given in clause 7.1.
- "Remittance" means any payment made or owing under the Master Murabaha Agreement or any other Finance Document by AIHL or the Guarantors to the Investment Agent due and payable to, and for the account of, the Participants, excluding for the avoidance of doubt:
- (a) those payments referred to in **clause 10.1**; and

12-11076-shl Doc 690-6 Filed 12/04/12 Entered 12/04/12 18:01:19 Exhibit F - Investment Agency Agreement Pg 6 of 34

(b) any other amounts received by the Investment Agent as reimbursement for its costs and expenses, save to the extent that the Investment Agent has been reimbursed for such costs and expenses by the Participants in accordance with **clause 10.2**.

"Restructuring" has the meaning given in clause 7.3.

"Secured Parties" has the mean given in clause 2.1(a).

"Sharing Participant" has the meaning given in clause 7.4(a).

"Transfer Certificate" has the meaning given in clause 11.5(a)(i).

"Transfer Date" means the date set out in the Transfer Certificate.

1.2 Construction of Certain Terms

The rules of construction and interpretation set out in clause 1.2 of the Master Murabaha Agreement apply to this Agreement as though they were set out in full in this Agreement, except that references to the Master Murabaha Agreement are to be construed as references to this Agreement.

1.3 Third Party Rights

Unless expressly provided to the contrary in this Agreement, a person who is not a party to this Agreement may not enforce any of its terms and shall have no rights with respect to the undertakings of the parties hereunder.

1.4 Headings

The headings in this Agreement are for convenience only and shall not affect the construction of this Agreement.

2. APPOINTMENT OF THE INVESTMENT AGENT AND SECURITY AGENT

2.1 Appointment; Etc.

(a) Each Participant hereby irrevocably appoints and designates each of the Investment Agent and the Security Agent as its agent and authorizes each Agent to take such actions on its behalf, including execution of the Finance Documents to which such Agent is a party and, in the case of the Security Agent, acting as agent for purposes of perfection, and to exercise such powers as are delegated to the such Agent by the terms of the Finance Documents, together with such actions and powers as are reasonably incidental thereto. Each Participant agrees that any action taken by either Agent or the Majority Participants in accordance with the provisions of the Finance Documents, and the exercise by either Agent or the Majority Participants of any rights or remedies set forth therein, together with all other powers reasonably incidental thereto, shall be authorized by and binding upon all the Agents, the Participants (collectively, the "Secured Parties"). Without limiting the generality of the foregoing, (i) the Investment Agent shall have the sole and exclusive authority to act as the disbursing and collecting agent for the Participants with respect to all payments and collections arising in connection with the Finance Documents; (ii) the Security Agent shall have the sole and exclusive authority to

12-11076-shl Doc 690-6 Filed 12/04/12 Entered 12/04/12 18:01:19 Exhibit F - Investment Agency Agreement Pg 7 of 34

- (A) act as security agent for the Secured Parties for purposes of perfecting and administering Liens granted to the Security Agent under the Finance Documents, and for all other purposes stated therein, (B) manage, supervise or otherwise deal with the Collateral; and (C) take any enforcement action or otherwise exercise any rights or remedies with respect to any Collateral under the Finance Documents, applicable law or otherwise and subject to the terms of the Finance Documents.
- (b) Each Person serving as an Agent hereunder shall have the same rights and powers in its capacity as a Participant as any other Participant and may exercise the same as though it were not an Agent hereunder, and such Person and its Affiliates may accept deposits from, lend money to and generally engage in any kind of business with any Obligor or any Affiliate thereof as if it were not an Agent hereunder.
- No Agent shall have any duties or obligations except those expressly set forth in the (c) Finance Documents. Without limiting the generality of the foregoing, (i) neither Agent shall be subject to any fiduciary or other implied duties, regardless of whether a Default has occurred and is continuing, (ii) neither Agent shall have any duty to take any discretionary action or exercise any discretionary powers, except discretionary rights and powers expressly contemplated by the Finance Documents that such Agent is required to exercise in writing as directed by the Majority Participants (or such other number or percentage of the Participants as shall be necessary under the circumstances as provided in clause 8), and (iii) except as expressly set forth in the Finance Documents, neither Agent shall have any duty to disclose, and shall not be liable for the failure to disclose, any information relating to any Obligor or any of their respective Subsidiaries that is communicated to or obtained by the Person serving as either such Agent or any of its Affiliates in any capacity. Neither Agent shall be liable for any action taken or not taken by it with the consent or at the request of the Majority Participants (or such other number or percentage of the Participants as shall be necessary under the circumstances as provided in clause 8) or in the absence of its own gross negligence or willful misconduct as finally determined in a final, non-appealable judgment of a court of competent jurisdiction. No Agent shall be deemed to have knowledge of any Default unless and until written notice thereof is given to such Agent by an Obligor or a Participant, and no Agent shall be responsible for or have any duty to ascertain or inquire into (A) any statement, warranty or representation made in or in connection with any Finance Document, (B) the contents of any certificate, report or other document delivered hereunder or in connection with any Finance Document, (C) the performance or observance of any of the covenants, agreements or other terms or conditions set forth in any Finance Document, (D) the validity, enforceability, effectiveness or genuineness of any Finance Document or any other agreement, instrument or document, (E) the creation, perfection or priority of Liens on the Collateral or the existence of the Collateral, or (F) the satisfaction of any condition set forth in any Finance Document, other than to confirm receipt of items expressly required to be delivered to such Agent. If any Participant acquires knowledge of a Default or failure of such conditions, it shall promptly notify the Investment Agent and the other Participants thereof in writing. Each Participant agrees that, except as otherwise provided in any Finance Document or with the written consent of the Investment Agent and Majority Participants, it will not take any enforcement action with respect to the Collateral or exercise any right that it might otherwise have under applicable law to credit bid at foreclosure sales, UCC sales or other similar dispositions of Collateral or to assert any rights relating to any Collateral.

12-11076-shl Doc 690-6 Filed 12/04/12 Entered 12/04/12 18:01:19 Exhibit F - Investment Agency Agreement Pg 8 of 34

- (d) The rights and remedies conferred upon the Agents under the Finance Documents may be exercised without the necessity of joinder of any other party, unless required by applicable law. Either Agent may request instructions from Majority Participants or other Secured Parties with respect to any act (including the failure to act) in connection with any Finance Documents, and may seek assurances to its satisfaction from Secured Parties of their indemnification obligations against all claims that could be incurred by such Agent in connection with any such act (or failure to act). Each Agent shall be entitled to refrain from any act until it has received such instructions or assurances, and no Agent shall incur liability to any Person by reason of so refraining. Instructions of the Majority Participants shall be binding upon all Secured Parties, and no Participant shall have any right of action whatsoever against either Agent as a result of such Agent acting or refraining from acting in accordance with the instructions of the Majority Participants. Notwithstanding the foregoing, instructions by and consent of specific Persons shall be required to the extent provided in clause 8. In no event shall either Agent be required to take any action that, in its opinion, is contrary to applicable law or any Finance Document or could subject such Agent (or its Affiliates or the respective directors, officers, trustees, employees, agents, controlling persons, representatives, partners, members, sub-agents, and advisors of such Agent and Affiliates (collectively, "Related **Parties**") to personal liability.
- (e) Each Agent shall be entitled to rely upon, and shall not incur any liability for relying upon, any notice, request, certificate, consent, statement, instrument, document, electronic transmission or other writing believed by it in good faith to be genuine and to have been signed or sent by the proper Person. Each Agent also may rely upon any statement made to it orally or by telephone and believed by it in good faith to be made by the proper Person, and shall not incur any liability for relying thereon. Each Agent shall have a reasonable and practicable amount of time to act upon any instruction, notice or other communication under any Finance Document, and shall not be liable for any delay in acting.
- (f) Each Agent may perform any and all of its duties and exercise its rights and powers by or through any one or more sub-agents or nominees appointed by such Agent. Each Agent and any such sub-agent and nominee may perform any and all of its duties and exercise its rights and powers through their respective Related Parties. The exculpatory provisions of the preceding paragraphs shall apply to any such sub-agents and nominees and to their respective Related Parties, and shall apply to their respective activities as an Agent hereunder. Each Agent may consult with and employ attorneys and other professionals and experts (collectively, "Agent Professionals"), and shall be entitled to act upon, and shall be fully protected in any action taken in good faith reliance upon, any advice given by an any such Person. No Agent shall be responsible for the negligence or misconduct of any agents, employees or Agent Professionals selected by it with reasonable care.
- (g) Subject to the appointment and acceptance of a successor Investment Agent or Security Agent, as the case may be, as provided in this paragraph (g), either Agent may resign at any time by notifying the Participants and AIHL. Upon any such resignation, the Majority Participants shall, with (if no Event of Default is then continuing) the consent of AIHL (not to be unreasonably withheld, delayed or conditioned), have the right to appoint a successor Investment Agent or Security Agent, as the case may be, which may be a Participant or an Affiliate of a Participant. Upon the acceptance by a successor Agent of an appointment as the Investment Agent or Security Agent hereunder, or upon appointment of the Majority Participants as such successor Agent, such successor shall

succeed to and become vested with all the rights, powers, privileges and duties of the retiring Agent, and the retiring Agent shall be discharged from its duties and obligations hereunder. After such Agent's resignation hereunder, the provisions of this **clause 2** and **clause 8** shall continue in effect for the benefit of such retiring Agent, its sub-agents, their nominees and their respective Related Parties in respect of any actions taken or omitted to be taken by any of them while it was acting as an Agent. Any successor to Fortress Credit Corp. by merger or acquisition of stock shall continue to be Investment Agent and Security Agent hereunder without further act on the part of the parties hereto, unless such successor resigns as provided above.

- (h) Each Participant acknowledges that it has, independently and without reliance upon either Agent or any other Participant and based on such documents and information as it has deemed appropriate, made its own analysis and decision to enter into this Agreement and to participate in the transactions under the Master Murabaha Agreement. Each Participant also acknowledges that it will, independently and without reliance upon either Agent or any other Participant and based on such documents and information as it shall from time to time deem appropriate, continue to make its own decisions in taking or not taking action under or based upon this Agreement, any other Finance Document or related agreement or any document furnished hereunder or thereunder.
- (i) If either Agent believes that it may be limited in the exercise of any rights or remedies under the Finance Documents due to any applicable law, such Agent may appoint an additional Person who is not so limited, as a separate co-agent. If either Agent so appoints a co-agent, each right and remedy intended to be available to such Agent under the Finance Documents shall also be vested in such separate agent. The Secured Parties shall execute and deliver such documents as either Agent deems appropriate to vest any rights or remedies in such agent. If any co-agent shall die or dissolve, become incapable of acting, resign or be removed, then all the rights and remedies of such agent, to the extent permitted by applicable law, shall vest in and be exercised by the relevant Agent until appointment of a new agent.

2.2 Indemnification by Participants

(a) Each Participant severally agrees to indemnify each Agent (to the extent not promptly reimbursed by AIHL) from and against such Participant's ratable share (determined as provided below) of any and all liabilities, obligations, losses, damages, penalties, actions, judgments, suits, costs, expenses or disbursements of any kind or nature whatsoever that may be imposed on, incurred by, or asserted against such Agent, arising solely in such Agent's capacity as an Agent hereunder and under the other Finance Documents, or any action taken or omitted by such Agent solely in its capacity as an Agent hereunder or under the Finance Documents (collectively, the "Indemnified Costs"); provided, however, that no Participant shall be liable for any portion of such liabilities, obligations, losses, damages, penalties, actions, judgments, suits, costs, expenses or disbursements resulting from any Agent's gross negligence or willful misconduct as determined in a final, non-appealable judgment by a court of competent jurisdiction. Without limitation of the foregoing, each Participant agrees to reimburse each Agent promptly upon demand for its ratable share of any costs and expenses (including, without limitation, fees and expenses of counsel) payable by any Obligor under any Finance Document, to the extent that such Agent, acting solely in its capacity as an Agent hereunder or under any other Finance Document, is not promptly reimbursed for such costs and expenses by Obligors. In the case of any investigation, litigation or proceeding giving rise to any Indemnified Costs, this **clause 2.2** applies whether any such investigation, litigation or proceeding is brought by any Participant or any other Person.

- (b) For purposes of this **clause 2.2**, each Participant's ratable share of any amount shall be determined, as at the incurrence of the relevant Indemnified Costs, according to its share of the aggregate amount of the Participations outstanding at such time. The failure of any Participant to reimburse any Agent promptly upon demand for its ratable share of any amount required to be paid by the Participants to such Agent as provided herein shall not relieve any other Participant of its obligation hereunder to reimburse such Agent for its ratable share of such amount, but no Participant shall be responsible for the failure of any other Participant to reimburse any Agent for such other Participant's ratable share of such amount. Without prejudice to the survival of any other agreement of any Participant hereunder, the agreement and obligations of each Participant contained in this **clause 2.2** shall survive the payment in full of the Obligations.
- (c) Neither Agent shall be liable to any Secured Party for any action taken or omitted to be taken under the Finance Documents, except for losses directly and solely caused by such Agent's gross negligence or willful misconduct as determined in a final, non-appealable judgment by a court of competent jurisdiction. Neither Agent assumes any responsibility for any failure or delay in performance or any breach by any Obligor, Participant or other Secured Party of any obligations under the Finance Documents. Neither Agent makes any express or implied representation, warranty or guarantee to the Secured Parties with respect to any Obligations, Collateral, Finance Documents or Obligors. No Agent nor any Related Party shall be responsible to the Secured Parties for any recitals, statements, information, representations or warranties contained in any Finance Documents; the execution, validity, genuineness, effectiveness or enforceability of any Finance Documents; the genuineness, enforceability, collectability, value, sufficiency, location or existence of any Collateral, or the validity, extent, perfection or priority of any Lien therein; the validity, enforceability or collectability of any Obligation; or the assets, liabilities, financial condition, results of operations, business, creditworthiness or legal status of any Obligor. No Agent nor any Related Party shall have any obligation to any Secured Party to ascertain or inquire into the existence of any Default or Event of Default.

2.3 No Third Party Beneficiaries

This **clause 2** is an agreement solely among the Participants and the Agents, and shall survive payment and satisfaction in full of the Obligations. Except solely to the extent of AIHL's rights to consent pursuant to and subject to the conditions in **clause 2.1(g)**, this **clause 2** does not confer any rights or benefits upon any Obligor or any other Person. As between the Obligors, on the one hand, and the Agents, on the other hand, any action that either Agent may take under any Finance Documents or with respect to any Obligations shall be conclusively presumed to have been authorized and directed by the Participants.

2.4 Partnership

None of this Agreement, the Contributions and the Participations shall, nor shall they be construed so as to, constitute a partnership between the Investment Agent and the Participants or an assignment (at law or in equity) of all or any part of the Remittances or of all or any of the Investment Agent's rights under the Finance Documents.

2.5 Role of the Arranger

Except as specifically provided in the Finance Documents, the Arranger does not have any obligations of any kind to any other party under or in connection with any Finance Document.

3. PARTICIPATION IN THE PURCHASE CONTRACT

3.1 Contribution Notification

Following receipt by the Investment Agent of a duly completed Transaction Request, the Investment Agent shall:

- (a) promptly provide a copy of such Transaction Request to each Participant; and
- (b) notify each Participant of that Participant's required participation in the Purchase Contract.

3.2 Contribution Payment

Each Participant shall make available to the Investment Agent its required participation in the Purchase Contract by making payment in Dollars in immediately available funds to the Investment Agent of an amount equal to its Relevant Percentage of the Cost Price specified in the Transaction Request on the Transaction Date in accordance with **clause 5.1**, provided that (i) all conditions to the funding of such Purchase Contract shall have been satisfied and (ii) after giving effect thereto, the aggregate outstanding contributions of such Participant shall not exceed such Participant's Facility Commitment.

3.3 Deferred Sale Price Entitlement

Upon payment of its Contribution as and when it is due, each Participant shall be entitled to receive, subject to the terms of the Netting Letter, its Relevant Percentage of the Deferred Sale Price due and payable to, and for the account of, the Participants in accordance with the Master Murabaha Agreement.

3.4 Unconditional Contribution

Each Participant's obligation to pay its Contribution is unconditional in circumstances where the Investment Agent is required to make the Cost Price payment pursuant to the Finance Documents.

3.5 Finance Parties' Rights and Obligations

The rights and obligations of each Finance Party under the Finance Documents are several. No Finance Party is responsible for the obligations of any other Participant, and a failure by a Finance Party to perform its rights or obligations under this Agreement (including without limitation a failure by a Participant to pay its Contribution to the Investment Agent in accordance with **clause 3.2**) shall not affect the rights or obligations of any other party.

3.6 Separate and Independent Obligations

Notwithstanding any other term of this Agreement, the interests of the Investment Agent and the Participants are several and each amount due to the Investment Agent and to each Participant is a separate and independent obligation.

3.7 Failure to Pay Contribution

In the event of a failure by a Participant to duly perform or comply with its obligation to pay its Contribution under this Agreement, AIHL and each Participant irrevocably and unconditionally agree that:

- (a) the Investment Agent shall no longer be obliged on the Transaction Date to pay or cause to be paid the full amount of the Cost Price, but it shall purchase a reduced amount of Commodities from the Seller for an amount equal to the portion of the Cost Price actually received from the Participants and shall accordingly only be obliged to pay or cause to be paid such reduced amount of the Cost Price;
- (b) AIHL shall have no recourse to the Investment Agent or to any non-defaulting Participant for such failure by a Participant; and
- (c) AIHL shall have recourse only to the defaulting Participant in respect of such failure.

4. PREPAYMENT AND INCREASED COSTS

4.1 Illegality and Change in Control

If:

- (a) it becomes unlawful in any applicable jurisdiction for a Participant to perform any of its obligations as contemplated by this Agreement or to fund or maintain its Participation or any obligation of a Participant under a Finance Document ceases to be legal, valid and binding on such Participant and that Participant promptly notifies the Investment Agent upon becoming aware of that event; or
- (b) following a Change of Control, any Participant notifies the Investment Agent that it no longer wishes to participate in the Facility,

the Investment Agent shall promptly notify AIHL pursuant to clause 9.1 of the Master Murabaha Agreement.

4.2 Increased Costs, Tax Credit and Mitigation

Each Participant shall comply with any obligation expressed to be assumed by it under clauses 8.2, 10 and 16.5 of the Master Murabaha Agreement.

4.3 Investment Agent's Distribution of Increased Costs; etc.

Any amount in respect of Increased Costs, Taxes or VAT received by the Investment Agent as part of the Deferred Sale Price or pursuant to **clause 8.3, 8.4** or **16.5** of the Master Murabaha Agreement shall be paid by the Investment Agent to the relevant Participant which has incurred

12-11076-shl Doc 690-6 Filed 12/04/12 Entered 12/04/12 18:01:19 Exhibit F - Investment Agency Agreement Pg 13 of 34

such Increased Cost (and the Participants who have not incurred the Increased Cost, Taxes or VAT shall not be entitled to any Relevant Percentage of such Increased Cost).

5. PAYMENTS BY THE PARTICIPANTS

5.1 Payments

The Contributions and any other sums to be paid by the Participants to the Investment Agent pursuant to this Agreement shall be made available in Dollars in immediately available funds as notified by the Investment Agent on the Transaction Date to such account as the Investment Agent shall have notified to the Participants for this purpose.

5.2 No set-off or Counterclaim

All payments by the Participants to the Investment Agent under this Agreement shall be made without set-off or counterclaim.

6. PAYMENTS BY THE INVESTMENT AGENT

6.1 Payments to Participants

- (a) Subject to **clauses 6.1(b)** and **6.2**, the Investment Agent will pay the Participants amounts in respect of their Participations on the dates and otherwise on the terms set out in this **clause 6**.
- (b) The Investment Agent shall apply each repayment towards the obligations of AIHL under the Finance Documents in the following order:
 - (i) **first**, in or towards payment pro rata of any unpaid fees, costs and expenses of the Arranger, the Investment Agent and the Security Agent under the Finance Documents;
 - (ii) **secondly**, in or towards payment pro rata of any unpaid fees, costs and expenses of the Participants under the Finance Documents;
 - (iii) **thirdly**, in or towards payment of any Profit Amount due but unpaid under the Finance Documents;
 - (iv) **fourthly**, in or towards payment of any element of the Deferred Sale Price (other than the Profit Amount) due but unpaid under the Finance Documents; and
 - (v) **fifthly**, in or towards payment of any other costs due but unpaid under the Finance Documents.
- (c) Notwithstanding anything to the contrary contained in this Agreement, if any Participant becomes a Defaulting Participant, then, until such time as such Participant is no longer a Defaulting Participant, any payment of any amounts received by either Agent for the account of such Defaulting Participant shall be applied at such time or times as may be determined by the Investment Agent as follows: first, to the payment of any amounts owing by such Defaulting Participant to the Investment Agent hereunder; second, to the extent no Default has occurred and is continuing, to the funding of any Participation that such Defaulting Participant has failed to fund, as determined by the Investment Agent;

12-11076-shl Doc 690-6 Filed 12/04/12 Entered 12/04/12 18:01:19 Exhibit F - Investment Agency Agreement Pg 14 of 34

third, to be held in a deposit account and released pro rata in order to satisfy such Defaulting Participant potential future funding obligations with respect to Participations under this Agreement; fourth, to the payment of any amounts owing to any Participant as a result of any judgment of a court of competent jurisdiction obtained by any Participant against such Defaulting Participant as a result of such Defaulting Participant's breach of its obligations under this Agreement; fifth, so long as no Default has occurred and is continuing, to the payment of any amounts owing to AIHL as a result of any judgment of a court of competent jurisdiction obtained by AIHL against such Defaulting Participant as a result of such Defaulting Participant's breach of its obligations under this Agreement; and sixth, to such Defaulting Participant or as otherwise directed by a court of competent jurisdiction; provided that if such payment is a payment of Cost Price under any Purchase Contract in respect of which such Defaulting Participant has not fully funded its appropriate share, such payment shall be applied solely to pay the Cost Price components of all non-defaulting Participant on a pro rata basis prior to being applied to the payment of any Cost Price component owed to a Defaulting Participant.

6.2 Limited Recourse

- (a) The obligation of the Investment Agent to pay the Participants the Relevant Percentage of the Deferred Sale Price and/or any other Remittance pursuant to this Agreement is conditional upon it having received the corresponding payment from an Obligor pursuant to the relevant Finance Document.
- (b) The Participants shall have no recourse to the Investment Agent in the event of any failure by any Obligor to make any such payments.
- (c) The Participants acknowledge and agree that the primary credit risk assumed by the Participants in relation to their Participations in the Investment (and their entry into this Agreement) is that of the Obligors. Each of the Participants also acknowledges and agrees that payment by any Obligor to the Investment Agent of the Deferred Sale Price (or any part thereof) in accordance with the relevant Finance Documents will irrevocably and unconditionally satisfy the relevant Obligor's obligations to pay the Deferred Sale Price (or such amount as is actually paid in accordance with the relevant Finance Documents).

6.3 Assumed Receipt

- (a) Where a sum is to be paid to the Investment Agent under the Finance Documents for another party to this Agreement, the Investment Agent may pay that sum to that other party, but the Investment Agent is not obliged to pay that sum to that other party until it has been able to establish to its satisfaction that it has actually received that sum.
- (b) If the Investment Agent pays an amount to another party to this Agreement and it proves to be the case that the Investment Agent had not actually received that amount, then the party to whom that amount was paid by the Investment Agent shall on demand refund the same to the Investment Agent.
- (c) Unless the Investment Agent shall have received notice from a Participant prior to the proposed Transaction Date that such Participant will not make available to the Investment Agent such Participant's Participation, the Investment Agent may assume that such Participant has made Participation available on such date in accordance with the terms of

12-11076-shl Doc 690-6 Filed 12/04/12 Entered 12/04/12 18:01:19 Exhibit F - Investment Agency Agreement Pg 15 of 34

this Agreement and may, in reliance upon such assumption, make a corresponding amount available to AIHL. In such event, if a Participant has not in fact made its Participation available to the Investment Agent, then the applicable Participant and AIHL severally agree to pay to the Investment Agent forthwith on demand (without duplication) such corresponding amount with Profit thereon, for each day from and including the date such amount is made available to AIHL. If such Participant pays such amount to the Investment Agent, then such amount shall constitute such Participant's Participation included in such Purchase Contract. Nothing herein shall be deemed to relieve any Participant from its obligation to fulfill its obligations hereunder or to prejudice any rights which the Investment Agent or AIHL may have against any Participant as a result of any default by such Participant hereunder.

6.4 Distribution

Subject to **clauses 6.2** and **6.3**, whenever the Investment Agent is satisfied that firstly it has actually received a payment in respect of a Remittance and secondly it is entitled to apply the amount received in such manner, the Investment Agent shall promptly pay to each Participant its Relevant Percentage of such payment.

6.5 Application of Moneys

For the purposes of this **clause 6**, if the Investment Agent obtains a partial or total payment of any amount due from any Obligor in respect of any Remittance by virtue of its being entitled to a set-off, banker's lien, counterclaim or any security or other payment and actually applies the amount of such payment in or towards satisfaction of amounts due to the Investment Agent in respect of that Remittance, the amount of such partial or total payment so applied will be treated for the purposes of this **clause 6** as if an actual payment of such amount had been received from an Obligor, but without prejudice to **clause 7**. Any such sums received by the Investment Agent for application in accordance with this **clause 6**.5 may be credited to a suspense account and held by the Investment Agent for the benefit of the Participants *pro rata* according to their interests hereunder but shall not be treated as having been received by the Investment Agent for the purposes of this **clause 6** unless and until such sums are appropriated by the Investment Agent in or towards payment in respect of that Remittance.

6.6 Reduction in Payments

Notwithstanding any provisions herein to the contrary, if an Obligor fails to pay any amount due to the Investment Agent pursuant to the relevant Finance Document in full on the due date, for any reason whatsoever (but without affecting the obligations of such Obligor to pay that amount), the obligation of the Investment Agent to make the payments referred to in this **clause 6** to the Participants shall be reduced *pro rata* by the shortfall in the payment received by the Investment Agent.

7. REFUND, SHARING AND FURTHER PAYMENTS

7.1 Refund of Payments

If at any time the Investment Agent:

12-11076-shl Doc 690-6 Filed 12/04/12 Entered 12/04/12 18:01:19 Exhibit F - Investment Agency Agreement Pg 16 of 34

- (a) becomes obliged by any law, rule or regulation to repay to any liquidator, trustee or other person all or part of an amount previously paid to the Investment Agent (the "**Relevant Receipt**") by an Obligor; or
- (b) is required to indemnify any liquidator, trustee or other person in respect of a Relevant Receipt.

being in any such case an amount which the Investment Agent shall have paid, or, but for the provisions of this **clause 7.1** would become liable to pay, to the Participants (the "**Relevant Payment**") pursuant to this Agreement, then:

- (i) the Investment Agent shall promptly notify each Participant of the relevant circumstances and of the amount to be repaid by the Investment Agent or, as the case may be, to be paid by way of indemnity by the Investment Agent (the "Excess Amount");
- (ii) each Participant shall on demand pay to the Investment Agent an amount equal to its Relevant Percentage of the Excess Amount, together with an amount equal to its Relevant Percentage of any compensation, costs, charges or expenses which the Investment Agent shall have become liable to pay in respect of such Excess Amount. Any demand under this **paragraph** (ii) shall be accompanied by a certificate (such certificate being conclusive evidence of the liability incurred by the Investment Agent) addressed to such Participant and signed by the Investment Agent setting out the Excess Amount and any compensation, costs, charges or expenses claimed together with any accompanying documentation if available; and
- (iii) each Obligor shall jointly and severally indemnify each Participant, for the Relevant Percentage of the Excess Amount which it has paid to the Investment Agent.

7.2 Late Payment Charge

If a Participant fails for any reason to pay any amount to the Investment Agent on demand under clause 3 or clause 7.1 or otherwise pursuant to this Agreement it undertakes to pay a late payment charge to the Investment Agent on such amount in respect of each day from the date of such demand until the date of actual payment to the Investment Agent (after as well as before judgment) at the rate per annum which the Investment Agent certifies to such Participant as being the reasonable and actual cost to the Investment Agent due to such delay from the date on which such payment was due to the date of actual receipt thereof by the Investment Agent. However, the Investment Agent shall only be entitled to retain such late payment charge as is equal to its actual out-of-pocket costs and expenses (not to include opportunity costs) incurred due to such failure by a Participant, and shall pay the balance on behalf of that Participant, as soon as fully and finally received, to such charitable foundations or Islamic scientific or medical institutions as it may select in its absolute discretion.

7.3 Insolvency

(a) If, in connection with any moratorium, rescheduling, refinancing, suspension of payments or other similar arrangement or circumstance (a "**Restructuring**") affecting any Remittance:

12-11076-shl Doc 690-6 Filed 12/04/12 Entered 12/04/12 18:01:19 Exhibit F - Investment Agency Agreement Pg 17 of 34

- (i) the Remittance (or equivalent amount) is paid in whole or in part but the obligation of the Obligor in respect of the amount paid is substituted by any other payment obligation; and/or
- (ii) any sum is paid into a blocked account or in non-convertible currency in or towards discharge or purported discharge of the Remittance or any part thereof; and/or
- (iii) the Investment Agent is obliged to provide funds in addition to the amount of the Investment, whether to AIHL or any other person,

then, subject to clause 7.3(b):

- (1) in the case of **paragraphs (i)** and **(ii) of clause 7.3(a)**, no such payment shall be deemed to have been received by the Investment Agent for the purposes of **clause 6**; and
- (2) in the case of **paragraph** (iii) of clause 7.3(a), each Participant shall be obliged to pay to the Investment Agent on demand an amount equal to its Relevant Percentage of such additional funds by way of further deposits in accordance with the provisions, *mutatis mutandis*, of clause 3 and clause 5.

(b) In relation to clause 7.3(a):

- (i) any such payment obligation as is referred to in **paragraph (i)** or **paragraph (iii)** of clause 7.3(a) shall be treated as between the Participants and the Investment Agent in the same way as the relevant Remittance, as if such obligation had been originally contained in the relevant Finance Document for the purpose of ascertaining the right (if any) of the Participants to receive subsequent payments under clause 6; and
- (ii) the Investment Agent will (at the request and cost of a Participant) assign to the Participants the Relevant Percentage of the Investment Agent's rights to any such blocked account or non-convertible currency as is referred to in **paragraph** (ii) of clause 7.3(a).1

7.4 Sharing by Participants

- (a) If any amount owing by an Obligor under the Finance Documents to a Participant (the "Sharing Participant") is discharged by voluntary or involuntary payment, set-off or any other manner other than through the Investment Agent (in accordance with this Agreement), then:
 - (i) the Sharing Participant shall within 3 Business Days notify the Investment Agent of the amount discharged (the "**Discharged Amount**") and the manner of its receipt or recovery;

_

Subject to Cayman counsel review.

12-11076-shl Doc 690-6 Filed 12/04/12 Entered 12/04/12 18:01:19 Exhibit F - Investment Agency Agreement Pg 18 of 34

- (ii) the Sharing Participant shall pay the Investment Agent an amount equal to the Discharged Amount within 1 Business Day of demand by the Investment Agent;
- (iii) the Investment Agent shall distribute the Discharged Amount in accordance with this Investment Agency Agreement; and
- (iv) the amount owed by AIHL under the Finance Documents shall be adjusted accordingly.
- (b) Reserved
- (c) If any Discharged Amount subsequently has to be wholly or partly refunded to the relevant Obligor by a Sharing Participant which has paid an amount equal to that Discharged Amount to the Investment Agent under clause 7.4(a), each Participant to which any part of that amount was distributed pursuant to clause 7.4(a)(iii) shall on request from the Sharing Participant repay to the Sharing Participant that Participant's proportionate share of the amount which has to be so refunded by the Sharing Participant. The Sharing Participant's rights of subrogation under clause 7.4(d) in respect of the refunded amount shall be cancelled and the relevant Obligor will be liable to the refunding Finance Party for the amount to refunded.
- (d) On a distribution by the Investment Agent under **clause 7.4(a)(iii)** the Sharing Participant will be subrogated to the rights of the Finance Parties which have shared in the redistribution.
- (e) Reserved
- (f) Each Participant shall on request supply to the Investment Agent such information as the Investment Agent may from time to time request for the purpose of this **clause 7.4**.

7.5 Partial Payments

If the Investment Agent receives a payment that is insufficient to discharge all the amounts then due and payable by AIHL under the Finance Documents, the Investment Agent shall apply that payment towards the obligations of AIHL under the Finance Documents in the order set forth in clause 6.1.

- 8. AMENDMENTS; DEFAULTING PARTICIPANT
- 8.1 Reserved
- 8.2 Reserved
- 8.3 Approval of Majority Participants

Subject to **clause 8.4**, neither this Agreement nor any other Finance Document nor any provision hereof or thereof may be waived, amended or modified except (i) in the case of this Agreement, pursuant to an agreement or agreements in writing entered into by AIHL and the Majority Participants (or the Investment Agent with the consent of the Majority Participants), or (ii) in the case of any other Finance Document (other than any such amendment to effectuate any modification thereto expressly contemplated by the terms of such other Finance Documents),

12-11076-shl Doc 690-6 Filed 12/04/12 Entered 12/04/12 18:01:19 Exhibit F - Investment Agency Agreement Pg 19 of 34

pursuant to an agreement or agreements in writing entered into by the Investment Agent, with the consent of the Majority Participants, and the Obligors party thereto. The Investment Agent may not make any material amendments or modifications, or waive any material provisions, which (in any case) are prejudicial to the position of the Participants to the ►[DD&Co Ltd Agreements, the Condor Agreements or the Netting Letter] without, in each case, first obtaining the consent of the Majority Participants and, in the case of the Condor Agreements, AIHL; provided that, for the avoidance of doubt, the Investment Agent and the Obligors may agree to technical, ministerial and other immaterial amendments and modifications to this Agreement and the other Finance Documents, in each case which are not prejudicial to the position of the Participants, without the consent of any Participant.

8.4 Amendments requiring Consent of all Participants

- (a) Reserved.
- (b) The Investment Agent shall not amend, modify, supplement or waive any provisions of this Agreement or any of the Finance Documents without obtaining the unanimous written consent of all affected Participants (or, in the case of clauses (i), (iv), (v), (vii), (viii) or (x), the written consent of all Participants) if the effect thereof would be to:
 - (i) change the currency of payment hereunder or under any Finance Document;
 - (ii) extend or defer the required date of payment of any amount payable by an Obligor under any Finance Document;
 - (iii) in relation to any Participant, increase the amount of its Contributions and/or Participation;
 - (iv) release all or substantially all of the Collateral;
 - (v) release all or substantially all of the Guarantors from their respective under the Guarantees:
 - (vi) modify the definition of the Majority Participants;
 - (vii) amend clauses 3.5 (Finance Parties' Rights and Obligations), 3.6 (Separate and Independent Obligations), or 8.4 (Amendments requiring Consent of all Participants) of this Agreement;
 - (viii) change the amount of or the method of calculation of the Deferred Sale Price;
 - (ix) reduce the Margin of the Facility or reduce the amount of any payment of principal, profit, fees or commission payable; or
 - (x) amend any provision which expressly requires the consent of all Participants.

8.5 Defaulting Participants

(a) Any Participant who has failed to duly perform or comply with its obligations under this Agreement shall not be included in any decision-making process for the purposes of clauses 4.1 (Illegality and Change in Control), 8.2 (Discretion of the Investment Agent), 8.3 (Approval of Majority Participants), 8.4 (Amendments requiring Consent of

12-11076-shl Doc 690-6 Filed 12/04/12 Entered 12/04/12 18:01:19 Exhibit F - Investment Agency Agreement Pg 20 of 34

all Participants) and 11.9 (Transfers by the Investment Agent) whilst such failure is outstanding and until such time as such failure has been remedied by such Participant to the reasonable satisfaction of the Investment Agent.

- (b) In the circumstances envisaged under **clause 8.5(a)**, the consent or non-approval of such defaulting Participant shall not be necessary in determining whether a decision has been made by:
 - (i) the Majority Participants; or
 - (ii) all the Participants (as appropriate),

and any such decision shall be made as if such defaulting Participant was not a Participant for the purposes of such decision-making process.

(c) AIHL shall not assume any liability or responsibility to any Finance Party as a direct result of any Participant failing to fulfill its obligations under or in connection with the Finance Documents.

9. SECURITY AGENT

9.1 Reserved

9.2 Security Agent as Proprietor

Each other Finance Party confirms that it does not wish to be registered as a joint proprietor of any Lien created pursuant to any Security Document and accordingly:

- (a) authorizes the Security Agent to hold such Security in its sole name as trustee for the Finance Parties; and
- (b) requests any relevant registry to register the Security Agent as a sole proprietor of any such Lien.

9.3 Investments

Except to the extent that a Security Document otherwise requires, any moneys received by the Security Agent under or pursuant to a Security Document may be:

- (a) invested in investments which it may select and which are authorized by applicable law; or
- (b) placed on deposit at any bank or institution (including itself),

in each case in the name or under the control of the Security Agent, and those moneys, together with any accrued income (net of any applicable Tax) shall be held by the Security Agent to the order of the Investment Agent, and shall be payable to the Investment Agent on demand; provided to the extent such monies are being held as Collateral but not for impending distribution to the Participants, such monies shall be invested in Shari'ah compliant investments (in in deposits on Shari'ah compliant terms which the Investment Agent may select.

10. COMMISSIONS, COSTS AND EXPENSES

10.1 No Sharing

Subject to any written agreement between either Agent and the Participants, it is confirmed that the Participants shall not be entitled to, and neither Agent shall be obliged to account to the Participants for, all or any part of any arranging, structuring, retainer, management or agency fees paid to, and for the account of, such Agent under or in respect of the Finance Documents.

10.2 Reimbursement of Costs and Expenses

Each Participant shall reimburse each Agent within 3 Business Days of production of an appropriate statement of costs and expenses prepared by the Investment Agent an amount equal to its Relevant Percentage of all costs, expenses (including legal expenses) and disbursements which may be incurred or made by such Agent in connection with its obligations under the Finance Documents to the extent that such costs and expenses are not reimbursed by the Obligors.

10.3 Participants' Indemnity to the Investment Agent

Each Participant shall (in proportion to its Relevant Percentage) indemnify each Agent, within three Business Days of demand, against any actual loss or liability (including, without limitation, stamp duty, documentary, or like Taxes) incurred by such Agent (otherwise than by reason of such Agent's gross negligence or willful misconduct as determined in a final, non-appealable judgment of a court of competent jurisdiction) in acting as Investment Agent or Security Agent (as the case may be) under the Finance Documents, unless such Agent has been reimbursed by an Obligor pursuant to a Finance Document.

11. ASSIGNMENT

11.1 Benefit and Burden

This Agreement shall be binding upon, and inure for the benefit of, the Participants, the Investment Agent and the Security Agent and their respective successors.

11.2 Assignment by the Participants

(a) Subject to **clause 11.5**, each Participant may at any time assign, transfer or otherwise dispose of, or offer or grant any interest in, the whole or any part of its rights and obligations under this Agreement together with the whole or any part of its rights and obligations under the Guarantee and the other Finance Documents, in each case with the consent of the Investment Agent and, so long as no Event of Default has occurred and is continuing, AIHL (which consent shall not be unreasonably withheld, delayed or conditioned, provided that AIHL shall be deemed to have consented to such assignment, transfer or disposition unless it shall have objected thereto in writing to the Investment Agent within 10 Business Days after receiving notice of such proposed assignment, transfer or other disposition). Any assignment by a Participant of less than all of its interests in the Purchased Contracts shall be deemed an as assignment of an equal proportion of each Purchase Contract then outstanding.

12-11076-shl Doc 690-6 Filed 12/04/12 Entered 12/04/12 18:01:19 Exhibit F - Investment Agency Agreement Pg 22 of 34

- (b) The Participant may disclose to any of its Affiliates and any other person:
 - (i) to (or through) whom the Participant assigns or transfers (or may potentially assign or transfer) in accordance with **clause 11.2(a)**;
 - (ii) with (or through) whom the Participant enters into (or may potentially enter into) any sub-participation in relation to, or any other transaction under which payments are to be made by reference to this Agreement or AIHL;
 - (iii) with (or through) whom the Participant enters into (or may potentially enter into) any securitization (or similar transaction of broadly equivalent economic effect) of that Participant's rights or obligations under the Finance Documents; or
 - (iv) to whom, and to the extent that, information is required to be disclosed by any applicable law or regulation,

any information about any of AIHL or any other member of the Group and the Finance Documents as the Participant shall consider appropriate, provided such person shall have agreed to be bound by the duty of confidentiality set out in **clause 14**.

11.3 Amended Schedule

Following any assignment or transfer pursuant to this **clause 11**, the Investment Agent will issue to the Participants and AIHL a new schedule to replace **Part B of Schedule 1** setting out the revised Contributions and Percentages of the Participants.

11.4 Conditions of Assignment or Transfer

An assignment and/or transfer by a Participant will only be effective if the procedure set out in **clause 11.5** and, if required, **clause 11.7**, is complied with.

11.5 Procedure for Transfer

- (a) An assignment or transfer is effected in accordance with **sub clause (b)** below when:
 - (i) the Investment Agent receives from the Existing Participant (as defined below) and the new Participant (the "New Participant"), four copies of a duly completed transfer certificate substantially in the form set out in Schedule 2 (a "Transfer Certificate"),
 - (ii) the Investment Agent executes the Transfer Certificate and confirms the Transfer Date, and
 - (iii) the Investment Agent delivers one copy to the Existing Participant. one copy to the New Participant and one copy to AIHL.

Subject to the foregoing, the Investment Agent shall, if it consents to such assignment, execute those Transfer Certificates, confirm the Transfer Date and deliver one copy to the Existing Participant, one copy to the New Participant and one copy to AIHL.

(b) Subject to such consent and recording thereof on the Register pursuant to **clause (d)**, from and after the Transfer Date specified in the relevant Transfer Certificate, the

12-11076-shl Doc 690-6 Filed 12/04/12 Entered 12/04/12 18:01:19 Exhibit F - Investment Agency Agreement Pg 23 of 34

assignee thereunder shall be a party hereto and, to the extent of the interest assigned by such Transfer Certificate, have the rights and obligations of a Participant under this Agreement, and the assigning Participant thereunder shall, to the extent of the interest assigned by such Transfer Certificate, be released from its obligations under this Agreement (and, in the case of a Transfer Certificate covering all of the assigning Participant's rights and obligations under this Agreement, such Participant shall cease to be a party hereto but shall continue to be entitled to the benefits of expense reimbursement and indemnification provisions contained herein and in the other Finance Documents with respect to facts and circumstances occurring on or prior to the effective date of such assignment).

(c) Investment Agent, acting for this purpose as a non-fiduciary agent of AIHL, shall maintain at its offices in the city of New York a copy of each Transfer Certificate delivered to it and a register for the recordation of the names and addresses of the Participants, the Facility Commitment of each Participant and Facility Contributions owing to each Participant pursuant to the terms hereof from time to time (the "Register"). The entries in the Register shall be conclusive and AIHL, each Agent and the Participants shall treat each Person whose name is recorded in the Register pursuant to the terms hereof as a Participant hereunder for all purposes of this Agreement and the other Finance Documents, in the absence of manifest error. Upon Investment Agent's consent to a duly completed Transfer Request, Investment Agent shall record the information contained therein in the Register. Notwithstanding anything to the contrary contained in this Agreement, any assignment of any Participation or any Facility Commitment shall be effective only upon appropriate entries with respect thereto being made in the Register. The Register shall be available for inspection by AIHL, any Agent and any Participation (solely with respect to its Participations), at any reasonable time and from time to time upon reasonable prior notice.

11.6 Limitation of Responsibility of Existing Participants

- (a) Unless expressly agreed to the contrary, an Existing Participant makes no representation or warranty and assumes no responsibility to a New Participant for:
 - (i) the legality, validity, effectiveness, adequacy or enforceability of the Finance Documents or any other documents;
 - (ii) the financial condition of any Obligor;
 - (iii) the performance and observance by any Obligor of its obligations under the Finance Documents or any other documents; or
 - (iv) the accuracy of any statements (whether written or oral) made in or in connection with any Finance Document or any other document,

and any representations or warranties implied by law are excluded.

(b) Each New Participant confirms to the Investment Agent, the Existing Participant and the Remaining Participants that it:

12-11076-shl Doc 690-6 Filed 12/04/12 Entered 12/04/12 18:01:19 Exhibit F - Investment Agency Agreement Pg 24 of 34

- (i) has made (and shall continue to make) its own independent investigation and assessment of the financial condition and affairs of each Obligor and its related entities in connection with its participation in this Agreement and has not relied exclusively on any information provided to it by the Existing Participants or the Investment Agent in connection with any Finance Document;
- (ii) will continue to make its own independent appraisal of the creditworthiness of each Obligor and its related entities whilst any amount is or may be outstanding under the Finance Documents or any Contribution is outstanding and the New Participant shall replace the Existing Participant to the extent that the Existing Participant has transferred its rights and obligations to them under this Agreement;
- (iii) has not relied on any representation by the Investment Agent as to the Shari'ah compliance of such agreements; and
- (iv) confirms that AIHL will not be liable for any costs or expenses associated solely with the transfer.
- (c) Nothing in this Agreement or in any Finance Document obliges an Existing Participant to:
 - (i) accept a re-transfer from a New Participant of any of the rights and obligations assigned or transferred under this **clause 11**; or
 - (ii) support any losses directly or indirectly incurred by the New Participant by reason of the non-performance by any Obligor of its obligations under the Finance Documents or otherwise.

11.7 Reserved

11.8 Assignment or Transfer Fee

The New Participant shall, on the date upon which an assignment or transfer takes effect, pay to the Investment Agent (for its own account) a fee of \$5,000.

11.9 Reserved

11.10 Reserved

11.11 Assignment by Obligors

No Obligor shall be entitled to assign or transfer any of its rights and obligations under this Agreement.

12. GENERAL

12.1 Payments Falling on Non-Business Days

If any payment under this Agreement is due on a day that is not a Business Day it shall be made on the next Business Day in the same calendar month (if there is one) or the preceding Business Day (if there is not).

12.2 Certifications and Determinations

Any certificate or determination issued or made by the Investment Agent as to any rate or amount payable pursuant to this Agreement or any other Finance Document shall, in the absence of manifest error, be conclusive and binding on the parties.

12.3 Communications

- (a) All correspondence in connection with this Agreement must be solely between the Investment Agent and the Participants and the Participants shall not communicate directly with AIHL in connection with the arrangements constituted by this Agreement or the existence of such arrangements.
- (b) The Investment Agent shall not be held liable for losses, costs or damage resulting from: (i) cable, telephone, facsimile or transmission errors or mutilation; (ii) equipment failure; or for (iii) delays, misrouting or losses by post offices, railroads or air carriers.

12.4 Know Your Customer

Each Participant shall promptly upon the request of the Investment Agent supply, or procure the supply of, such documentation and other evidence as is reasonably requested by the Investment Agent (for itself) in order for the Investment Agent to carry out and be satisfied it has complied with all necessary know your customer requirements or other similar checks under all applicable laws and regulations pursuant to this Agreement as far as permitted by applicable law.

12.5 No Waiver

No failure or delay on the part of any party in exercising any right hereunder shall operate as a waiver of, or impair, any such right. No single or partial exercise of any such right shall preclude any other or further exercise thereof or the exercise of any other right. No waiver of any such right shall be effective unless given in writing. No waiver of any such right shall be deemed a waiver of any other right hereunder,

12.6 Partial Invalidity

Any provision of this Agreement that is prohibited or unenforceable in any jurisdiction shall, as to that jurisdiction, be ineffective to the extent of that prohibition or unenforceability without invalidating the remaining provisions hereof or affecting the validity or enforceability of that provision in any other jurisdiction.

12.7 Counterparts

This Agreement may be executed in any number of counterparts, and all the counterparts taken together shall be deemed to constitute one and the same agreement. Delivery of an executed counterpart of a signature page of this Agreement by facsimile or electronic transmission via emailed pdf or other similar format shall be effective as delivery of a manually executed counterpart of this Agreement.

12.8 Documents to be in English

Each document to be delivered with respect to this Agreement shall be in the English language or shall be accompanied by an English translation thereof certified by the concerned party to be complete and correct.

12.9 Indemnity

Each Obligor jointly and severally agrees to indemnify each Finance Party and their Related Parties against, and hold each Finance Party and their Related Parties harmless from, any and all liabilities, obligations, losses, damages, penalties, actions, judgments, suits, costs, expenses or disbursements of any kind or nature whatsoever, including the fees, charges and disbursements of any counsel for any Finance Party or Related Party, incurred by or asserted against any Finance Party or Related Party arising out of, in connection with, or as a result of (i) the execution or delivery of the Finance Documents or any agreement or instrument contemplated thereby, the performance by the parties hereto of their respective obligations thereunder or the consummation of the or any other transactions contemplated hereby, (ii) any purchase or sale of Commodities by AIHL or the use of the proceeds therefrom, or (iii) any actual or prospective claim, litigation, investigation or proceeding relating to any of the foregoing, whether based on contract, tort or any other theory and regardless of whether any Finance Party or Related Party is a party thereto and whether or not such proceeding is instituted or brought on behalf of a third party or by any Obligor or any of its respective Affiliates; provided that such indemnity shall not, as to any Finance Party or Related Party, be available to the extent that such liabilities, obligations, losses, damages, penalties, costs, expenses or disbursements are determined, in a final, non-appealable judgment of or a court of competent jurisdiction, to have resulted solely from the gross negligence or willful misconduct of such Finance Party or Related Party.

13. NOTICES

13.1 Notices

Subject to **clause 13.2** the notice provisions set out in clause 19 of the Master Murabaha Agreement apply, *mutatis mutandis*, to this Agreement and for this purpose:

- (a) the administrative details of each Participant are those notified in writing to the Investment Agent prior to the date on which it becomes a party or, as applicable, as provided pursuant to **clause 11** or as a Participant may notify to the Investment Agent by not less than five Business Days' notice; and
- (b) the administrative details for the Security Agent shall be the same as for the Investment Agent.

13.2 Notices through the Investment Agent

All notices from a Participant to AIHL shall be sent through the Investment Agent. All payments pursuant to a notice delivered by the Investment Agent for the account of a Participant shall be received by that Participant through the Investment Agent.

12-11076-shl Doc 690-6 Filed 12/04/12 Entered 12/04/12 18:01:19 Exhibit F - Investment Agency Agreement Pg 27 of 34

13.3 Electronic Communication

- (a) Any communication to be made between the parties under or in connection with this Agreement may be made by electronic mail or other electronic means, if the parties:
 - (i) agree that, unless and until notified to the contrary, this is to be an accepted form of communication:
 - (ii) notify each other in writing of their electronic mail address and/or any other information required to enable the sending and receipt of information by that means; and
 - (iii) notify each other of any change to their address or any other such information supplied by them.
- (b) Any electronic communication made between the parties will be effective only when actually received in readable form and in the case of any electronic communication made by the any other party to the Investment Agent only if it is addressed in such a manner as the Investment Agent shall specify for this purpose.

14. CONFIDENTIALITY

14.1 Duty of Confidentiality

Subject to **clause 14.2** (*Disclosure of Information*) the Participants agree to treat this Agreement and any information supplied by the Investment Agent or any Obligor in connection herewith as being strictly confidential.

14.2 Disclosure of Information

Any Finance Party and its officers and agents may disclose information about AIHL, the Guarantors and the Finance Documents to:

- (a) the head office of that Finance Party, any of its subsidiaries or subsidiaries of its holding company, Affiliates, representative and branch offices in any jurisdiction (together with that Finance Party, the "**Permitted Parties**");
- (b) professional advisers and service providers of the Permitted Parties who are under a duty of confidentiality to the Permitted Parties and have been informed of the confidential nature of the information (defined below) and are instructed to keep such information confidential;
- (c) any actual or potential assignee, novatee, transferee, participant or sub-participant in relation to any of that Finance Party's rights and/or obligations under any Finance Document (or any agent or adviser of any of the foregoing);
- (d) any rating agency, insurer or insurance broker of, or direct or indirect provider of credit protection to any Permitted Party;
- (e) any court or tribunal or regulatory, supervisory, governmental or quasi-governmental authority with jurisdiction over the Permitted Parties;

12-11076-shl Doc 690-6 Filed 12/04/12 Entered 12/04/12 18:01:19 Exhibit F - Investment Agency Agreement Pg 28 of 34

- (f) any other person if and to the extent required by applicable law or regulation;
- (g) to any other party to this Agreement; or
- (h) in connection with the exercise of remedies hereunder or any action or proceeding relating to any Finance Document or the enforcement of rights hereunder;

(other than information that becomes publicly available other than through a breach of this Agreement and other than information that becomes available to such Finance Party from a source other than an Obligor) that an Obligor has clearly identified as confidential to such Obligor at the time of delivery thereof ("Information").

14.3 Survival of Obligation

The obligation of confidentiality in **clause 14.1** shall survive the termination of the Agreement until the later of (i) a period of 24 months after the termination of this Agreement, and (ii) the final conclusion of any dispute, suit, action, or proceeding arising from the Finance Documents and involving any Finance Party, unless any of the relevant confidential information enters the public domain through no fault of the relevant party or its agents.

15. GOVERNING LAW

This Agreement and all non-contractual obligations arising out of or in connection with this Agreement shall be governed by and construed in accordance with the law of the State of New York, United States of America.

16. DISPUTE RESOLUTION

16.1 Jurisdiction

EACH OF THE FINANCE PARTIES IRREVOCABLY AND UNCONDITIONALLY SUBMITS, FOR ITSELF AND ITS PROPERTY, TO THE EXCLUSIVE JURISDICTION OF THE BANKRUPTCY COURT AND, IF THE BANKRUPTCY COURT DOES NOT HAVE (OR ABSTAINS FROM) JURISDICTION, OF THE COURTS OF THE STATE OF NEW YORK SITTING IN NEW YORK COUNTY AND OF THE UNITED STATES DISTRICT COURT OF THE SOUTHERN DISTRICT OF NEW YORK, AND ANY APPELLATE COURT FROM ANY THEREOF, IN ANY ACTION OR PROCEEDING ARISING OUT OF OR RELATING TO THIS AGREEMENT OR ANY OTHER FINANCE DOCUMENT, OR FOR RECOGNITION OR ENFORCEMENT OF ANY JUDGMENT, AND EACH OF THE PARTIES HERETO IRREVOCABLY AND UNCONDITIONALLY AGREES THAT ALL CLAIMS IN RESPECT OF ANY SUCH ACTION OR PROCEEDING MAY BE HEARD AND DETERMINED IN SUCH NEW YORK STATE COURT OR, TO THE FULLEST EXTENT PERMITTED BY APPLICABLE LAW, IN SUCH FEDERAL COURT. EACH OF THE PARTIES HERETO AGREES THAT A FINAL JUDGMENT IN ANY SUCH ACTION OR PROCEEDING SHALL BE CONCLUSIVE AND MAY BE ENFORCED IN OTHER JURISDICTIONS BY SUIT ON THE JUDGMENT OR IN ANY OTHER MANNER PROVIDED BY LAW. NOTHING IN THIS AGREEMENT OR IN ANY OTHER FINANCE DOCUMENT SHALL AFFECT ANY RIGHT TO BRING ANY ACTION OR PROCEEDING RELATING TO THIS AGREEMENT OR ANY OTHER FINANCE DOCUMENT AGAINST AIHL OR ANY OTHER OBLIGOR OR ITS PROPERTIES IN THE COURTS OF ANY OTHER JURISDICTION, AND AIHL AND EACH GUARANTOR EXPRESSLY SUBMITS

AND CONSENTS IN ADVANCE TO SUCH JURISDICTION IN ANY ACTION OR SUIT COMMENCED IN ANY SUCH COURT, AND AIHL AND EACH GUARANTOR HEREBY WAIVES ANY OBJECTION THAT SUCH OBLIGOR MAY HAVE BASED UPON IMPROPER VENUE OR FORUM NON CONVENIENS AND HEREBY CONSENTS TO THE GRANTING OF SUCH LEGAL OR EQUITABLE RELIEF AS IS DEEMED APPROPRIATE BY SUCH COURT.

EACH PARTY HERETO HEREBY WAIVES, TO THE FULLEST EXTENT PERMITTED BY APPLICABLE LAW, ANY RIGHT IT MAY HAVE TO A TRIAL BY JURY IN ANY LEGAL PROCEEDING DIRECTLY OR INDIRECTLY ARISING OUT OF OR RELATING TO THIS AGREEMENT, ANY OTHER FINANCE DOCUMENT OR THE TRANSACTIONS CONTEMPLATED HEREBY OR THEREBY (WHETHER BASED ON CONTRACT, TORT OR ANY OTHER THEORY). EACH PARTY HERETO (A) CERTIFIES THAT NO REPRESENTATIVE, AGENT OR ATTORNEY OF ANY OTHER PARTY HAS REPRESENTED, EXPRESSLY OR OTHERWISE, THAT SUCH OTHER PARTY WOULD NOT, IN THE EVENT OF LITIGATION, SEEK TO ENFORCE THE FOREGOING WAIVER AND (B) ACKNOWLEDGES THAT IT AND THE OTHER PARTIES HERETO HAVE BEEN INDUCED TO ENTER INTO THIS AGREEMENT BY, AMONG OTHER THINGS, THE MUTUAL WAIVERS AND CERTIFICATIONS IN THIS CLAUSE.

16.2 Service of Process

Without prejudice to any other mode of service allowed under any relevant law AIHL agrees that the process by which any suit, action or proceeding is begun may be served on it by being delivered in connection with any suit, action or proceeding in New York to Corporation Service Company.

16.3 Immunity

To the extent that AIHL may in any jurisdiction claim for itself or its assets immunity from suit, execution, attachment (whether in aid of execution, before judgment or otherwise) or other legal process and to the extent that in any such jurisdiction there may be attributed to itself or its assets such immunity (whether or not claimed), AIHL hereby irrevocably agrees not to claim and hereby irrevocably waives such immunity to the full extent permitted by the laws of such jurisdiction.

THIS AGREEMENT is entered into by the parties on the date stated at the beginning of this Agreement.

SCHEDULE 1

THE PARTICIPANTS AND GUARANTORS

٨	T1.	Caramantana
Α.	1 ne	Guarantors

B. The Participants

Participant	Facility Contribution (US\$)	Relevant Percentage (%)
Total		

SCHEDULE 2

FORM OF TRANSFER CERTIFICATE

To: Fortress Credit Corp. as Investment Agent

From: [The Existing Participant] (the "Existing Participant") and [The New Participant"] (the "New Participant")

Dated:

INVESTMENT AGENCY AGREEMENT BETWEEN, AMONGST OTHERS, FORTRESS CAPITAL CORP. AS INVESTMENT AGENT AND ARCAPITA INVESTMENT HOLDINGS LIMITED DATED [] 2012 (THE "AGREEMENT")

- We refer to the Agreement. Terms defined in the Agreement have the same meaning herein unless given a different meaning herein.
- 2 This is a Transfer Certificate.
- We refer to **clause 11.5** of the Agreement:
 - (a) The Existing Participant and the New Participant agree to the Existing Participant transferring to the New Participant [all] / []% of the Existing Participant's Facility Commitment and existing Contributions and those rights and obligations referred to in the Schedule in accordance with **clause 11.5**.
 - (b) The proposed Transfer Date is [].
 - (c) The address, telefax number and attention details for notices of the New Participant for the purposes of **clause 13** of the Agreement are set out in the Schedule.
- The New Participant expressly appoints and authorizes (a) the Investment Agent to act for and on its behalf under the Finance Documents (other than the Security Documents) and (b) the Security Agent to act for and on its behalf under the Security Documents.
- The New Participant expressly acknowledges the limitations on the Existing Participant's obligations set out in **clause 11.6** of the Agreement.
- This Transfer Certificate may be executed in any number of counterparts and this has the same effect as if the signatures on the counterparts were on a single copy of this Transfer Certificate.
- 7 This Transfer Certificate is governed by the law of the State of New York, United States of America.
- This Transfer Certificate may be executed in any number of counterparts, and all the counterparts taken together shall be deemed to constitute one and the same agreement. Delivery of an executed counterpart of a signature page of this Transfer Certificate by facsimile or electronic transmission via e-mailed pdf or other similar format shall be effective as delivery of a manually executed counterpart of this Agreement.

Schedule

Contribution and rights and obligations to be transferred

Amount of Facility Commitment transferred: \$					
Amount of existing Contribution transferred: \$2					
Notice details of the New Participant:					
Address:					
Telefax Number:					
For the attention of:					
Account details of the New Participant:					
Signatories					
[Existing Participant]	[New Participant]				
By:	By:				
Title:	Title:				
The Investment Agent confirms that the Transfer Date	e is [].				
Fortress Credit Corp.					
By:					
Title:					

Details of related Purchase Contract are set forth on the Annex hereto.

12-11076-shl Doc 690-6 Filed 12/04/12 Entered 12/04/12 18:01:19 Exhibit F - Investment Agency Agreement Pg 33 of 34

ANNEX

Purchase Participant's Contribution to be transferred to (prior to transfer)

Existing Contribution to be transferred to (prior to transfer)

Contribution to be transferred to New Participant

New Participant

Remaining Contribution of Existing Payment Date

12-11076-shl Doc 690-6 Filed 12/04/12 Entered 12/04/12 18:01:19 Exhibit F - Investment Agency Agreement Pg 34 of 34

EXECUTION PAGE OF INVESTMENT AGENCY AGREEMENT

AIHL	
Signed by duly authorized for and on behalf of Arcapita Investment Holding Limited)))
INVESTMENT AGENT	
Signed by duly authorized for and on behalf of Fortress Credit Corp.)))
SECURITY AGENT	
Signed by duly authorized for and on behalf of Fortress Credit Corp.)
ARRANGER	
Signed by duly authorized for and)

on behalf of

Fortress Credit Corp.